

David Seebacher, Dipl.-Ing.

### Switched Mode Transmitter Architectures for Efficient Wireless Communication Systems

#### DISSERTATION

zur Erlangung des akademischen Grades Doktor der technischen Wissenschaften

eingereicht an der

### Technischen Universität Graz

Betreuer

Univ.-Prof. Dipl.-Ing. Dr.techn. Wolfgang Bösch

Institut für Hochfrequenztechnik

Zweitgutachter: Prof. Dr. Patrick Reynaert, KU Leuven

Graz, September 2014

### EIDESSTATTLICHE ERKLÄRUNG AFFIDAVIT

Ich erkläre an Eides statt, dass ich die vorliegende Arbeit selbstständig verfasst, andere als die angegebenen Quellen/Hilfsmittel nicht benutzt, und die den benutzten Quellen wörtlich und inhaltlich entnommenen Stellen als solche kenntlich gemacht habe. Das in TUGRAZonline hochgeladene Textdokument ist mit der vorliegenden Dissertation identisch.

*I* declare that I have authored this thesis independently, that I have not used other than the declared sources/resources, and that I have explicitly indicated all material which has been quoted either literally or by content from the sources used. The text document uploaded to TUGRAZonline is identical to the present doctoral dissertation.

Datum / Date

Unterschrift / Signature

### Abstract

With the increasing number of wireless services and users, the required bandwidth is ever increasing. To use the available spectrum efficiently modern communication standards, such as LTE, employ sophisticated modulation schemes with high crest factors. As a result, the power amplifier (PA) spends most of its time providing for less than its maximum output power, leading to low average efficiency in traditional designs. Therefore, PA concepts providing efficiency enhancement in back off are key for efficient transmitters.

Base Band PWM (BB PWM) is a promising concept to provide high efficiency over a wide output power range. Conventional BB PWM transmitters use an isolator before the band pass filter to prevent its reflections interfering with the PA operation. This causes a loss of energy in the modulation sidebands and limits the achievable efficiency. By removing the isolator and by connecting the filter directly, while accounting for the filters reflections, these losses can be mitigated. Such designs require multilevel PA structures that can deal efficiently with the resulting continuous load current. The concept of direct filter connection results in dynamic load modulation of the PA, enabling efficient operation. As the load modulation trajectories are of importance for successful designs, generalized analytical equations are presented. A design methodology based on both, the load pull contours of the PA and the load modulation trajectories, has been developed and applied to the presented designs. The band pass filter is the starting point for the design of the matching network as the phase of the reflections caused by the matching network have a significant influence on the design. The impact of the coupling to a cavity resonator and its reflections is studied. It was found that for best performance, a short connection between PA and band pass filter is vital. Thus requires a co-design of the output matching network on printed circuit board level and the cavity band pass filter. Based on the derived design requirements, an implementation study of four integrated multi level PA structures designed on a GaN MMIC process has been carried out. Two of the presented structures are based on a parallel concept, while the other two rely on series concepts. All four proposed PA structures can be combined with direct filter connection and outperform conventional BB PWM operation.

For an efficient transmitter system, the power consumption of the modulator also plays an important role. In that context the use of RF PWM modulators shows promising results, but they still lag behind conventional state of the art modulators in terms of spectral quality. As a result this work focused on the required signal processing to correct for digital, as well as analog imperfections. Aliasing effects of digital RF PWM, which are due to its discrete time nature, could be reduced by introducing cross point estimation. With a special predistortion scheme tailored to the properties of RF PWM modulation, the spectral limitations due to the driver nonlinearity could be overcome. Additionally, a concept for digitally correcting the device mismatch is presented. It can optionally be combined with delta sigma modulation to improve the inband signal quality. The proposed signal processing chain is verified by extensive simulations and supported by measurements. By using the applied corrections mismatch, driver nonlinearity and aliasing due to the discrete time nature can be compensated and high signal quality can be maintained.

## Kurzfassung

Durch die ständig steigende Anzahl an Nutzern von mobilen Breitbanddiensten wurde das verfügbare Spektrum zu einer wertvollen Ressource. Moderne Kommunikationsstandards, wie LTE, verwenden zur effizienten Nutzung des Spektrums aufwändige Kodierverfahren, wodurch die Leistung des zu sendenden Signals stark variiert. Dadurch wird der Leistungsverstärker meist mit verringerter Leistung betrieben, was bei konventionellen Verstärkern zu einer starken Reduktion des Wirkungsgrades führt. Konzepte zur Effizienzsteigerung des Leistungsverstärkers von Sendern bei niederen Ausgangsleistungen sind daher extrem wichtig.

Basis Band Pulsweitenmodulation (BB PWM) hat sich als vielversprechende Maßnahme herausgestellt. Üblicherweise wird bei BB PWM Sendern ein Isolator eingesetzt um zu verhindern, dass die Reflektionen des zur Demodulation erforderlichen Bandpassfilters (BPF) Einfluss auf die korrekte Funktion des Leistungsverstärkers haben. Dies führt jedoch zu Verlusten, da die Energie der Modulationsseitenbänder verloren geht. Durch den direkten Anschluss des BPF an den Leistungsverstärker können diese Verluste minimiert und die Effizienz des Senders gesteigert werden. Dazu werden Verstärker benötigt, die mehrere Ausgangsleistungen erzeugen und mit dem entstehenden, konstanten, Laststrom des Filters und der daraus resultierenden Lastmodulation effizient umgehen können. Um die bestmögliche Effizienz zu erreichen, ist die Berücksichtigung der Lastmodulationskennlinien erforderlich. Daher werden allgemeine analytische Gleichungen zu deren Berechnung präsentiert, welche die Grundlage der angewendeten Entwicklungsmethode unter Berücksichtigung der Effizienzkontouren des Verstärkers für verschiedene Lastimpedanzen bilden. Den Startpunkt für die Entwicklung des Anpassungsnetzwerkes bilden die Reflektionen des verwendeten BPF, in diesem Fall eines Hohlraumresonators. Da diese Reflektionen, vor allem deren Phase, stark von der Art und Stärke der Koppelung abhängen, wurde ihr Einfluss analysiert und die Ergebnisse in die Entwicklung des Anpassungsnetzwerkes mit einbezogen. Des Weiteren wurde herausgefunden, dass für den breitbandigen Betrieb eine kurze Verbindung zwischen Verstärker und Filter unabdingbar ist, was zur Folge hat, dass das Anpassungsnetzwerk auf Leiterplattenebene zusammen mit dem BPF entworfen werden muss. Anhand der aufgestellten Anforderungen wurden vier integrierte Schaltungen in einem GaN MMIC Prozess entwickelt, welche den direkten Filteranschluss unterstützen und bessere Effizienzen als konventionelle BB PWM erzielen.

Neben der Effizienz des Verstärkers spielt auch der Leistungsverbrauch des Modulators eine wichtige Rolle. In diesem Zusammenhang hat sich die Verwendung von auf RF PWM basierenden Modulatoren als eine besonders energieeffiziente Implementierungsvariante erwiesen, welche jedoch die erforderliche spektrale Qualität noch nicht bieten. Daher wurde hier der Fokus auf die erforderliche Signalverarbeitung zur Korrektur von analogen, sowie digitalen Effekten gelegt. Zur Kompensation von Aliasingeffekten, resultierend aus der zeitdiskreten Ansteuerung des RF PWM Modulators, wurde eine Methode zur Berechnung der korrekten Signalübergange vorgeschlagen. Im analogen Bereich begrenzt vor allem die Nichtlinearität der verwendeten Treiberschaltungen die Signalqualität. Dieser Effekt kann mittels einer speziell auf die Eigenschaften digitaler RF PWM zugeschnittenden Vorverzerrung bestmöglich kompensiert werden. Zudem wird ein Konzept zur Kompensation der Fertigungstoleranzen des Modulators auf der digitalen Ebene vorgestellt, welches optional mit einem  $\Delta\Sigma$  Modulator kombiniert werden kann, um die Signalqualität bei limitierter Auflösung zu erhöhen. Die vorgeschlagene Signalverarbeitungskette wurde anhand von Simulationen und Messungen verifiziert. Durch die Anwendung der entsprechenden Korrekturmaßnahmen können die vorhandenen Fehler kompensiert und beste Signalqualität erreicht werden.

### Acknowledgment

A number of people deserve thanks for their support and help with my PhD thesis. It is therefore my greatest pleasure to express my gratitude to them all in this acknowledgment. First of all, I would like to thank Peter Singerl for the industrial supervision of this thesis as well as for sharing his expertise and for his continuous support in various matters. Thanks to Christian Schuberth for sharing his rich experience in PA design with me, as well as for his patient explanations and the support during the writing of many papers. I am also grateful to Franz Dielacher for his continuous encouragement and belief in the switched mode transmitter project. The support of Martin Mataln when it came to predistortion topics and the FPGA implementation/measurements was much appreciated. I would also like to thank Marc Tibout and Koen Mertens for the review of the designs as well as for their help with various other problems regarding the UNIX system. Important to mention are also Bittolo Alberto, who investigated different matching structures during his master thesis, and Stefano Lenuzza, who offered me support with various measurement topics. Furthermore I would like to thank Wolfgang Bösch for the academic supervision and his guidance. Michael Gadringer's comments on various technical problems, as well as on many of the published papers, shall also be acknowledged. I am furthermore grateful to Patrick Revnaert for spending the effort to be my second examiner. I would also like to thank Matthias Thorsell and Christian Fager for their support and discussion regarding the GaN MMIC process. The discussions with Shailesh Kulkarni and Wim Dehane on the digital implementation of the RF PWM modulator provided valuable input for the development of the compensation algorithms. Also Yannis Papananos and Nikolaos Alexiou for providing me insights in the analysis and design of the analog RF PWM modulator shall not remain unmentioned. Last but not least I want to mention family and friends, who accepted my occasional absence and supported me in various ways during these years.

# Contents

	Introduc	ction	1
2	Amplifie	r Fundamentals	4
2	2.1 Line	ar Amplifiers	4
2	2.2 Clas	sJ	6
2	2.3 Clas	sF	6
2	2.4 Clas	s D	8
2	2.5 Clas	s DE	9
2	2.6 Clas	s E	10
2	2.7 Cont	trolled Rectifiers	11
	2.7.1	Class B Rectifier	11
	2.7.2	2 Class D Rectifier	11
	2.7.3	Class DE Rectifier	12
	2.7.4	Class E Rectifier	13
	2.7.5	Class F Rectifier	13
2	2.8 Com	parison	14
2	2.9 Loss	Mechanisms RF Transistors	15
2	2.10 Aver	age Efficiency	18
3 I	Efficienc	y Enhancement Concepts	20
3	3.1 Doh	erty	20
3	3.2 Line	ar Amplification Using Nonlinear Components	21
3	3.3 Chir	eix	21
3	3.4 Enve	elope Tracking / Envelope Elimination and Restoration	23
3	3.5 Load	l Modulation	24
3	3.6 Base	Band PWM	24
	3.6.1	Isolated Broadband Load	24
	3.6.2	2 Direct Filter Connection	26
		3.6.2.1 Parallel Resonator	26
		3.6.2.2 Series Resonator $\ldots$	26
		3.6.2.3 Series Besonator with Switch	27
			41
	3.6.3	Multilevel	21
	$3.6.3 \\ 3.6.4$	Multilevel       Multiphase	21 28 30
5	3.6.3 3.6.4 3.7 Ener	Multilevel       Multiphase         gy Recovery       Summer	28 30 32
Ş	3.6.3 3.6.4 3.7 Ener 3.7.1	Multilevel	28 30 32 32
3	3.6.3 3.6.4 3.7 Ener 3.7.1 3.7.2	Multilevel       Multiphase         gy Recovery       DC Energy Recovery LINC         DC Energy Recovery BB PWM	28 30 32 32 33
	3.6.3 3.6.4 3.7 Ener 3.7.1 3.7.2 3.7.3	Multilevel       Multiphase         gy Recovery       DC Energy Recovery BB PWM         RF Energy Recovery       RF Energy Recovery	27 28 30 32 32 33 33
5	3.6.3 3.6.4 3.7 Ener 3.7.1 3.7.2 3.7.3	Multilevel       Multiphase         gy Recovery       DC Energy Recovery LINC         DC Energy Recovery BB PWM         RF Energy Recovery         3.7.3.1	28 30 32 32 33 33 34
ę	3.6.3 3.6.4 3.7 Ener 3.7.1 3.7.2 3.7.3	Multilevel       Multiphase         gy Recovery       DC Energy Recovery LINC         DC Energy Recovery BB PWM       RF Energy Recovery         3.7.3.1       Reuse During OFF Period         3.7.3.2       Reuse During ON Period	28 30 32 32 33 33 33 34 36
	3.6.3 3.6.4 3.7 Ener 3.7.1 3.7.2 3.7.3	Multilevel       Multiphase         gy Recovery       DC Energy Recovery LINC         DC Energy Recovery BB PWM         RF Energy Recovery         3.7.3.1         Reuse During OFF Period         3.7.3.2         Reuse During ON Period         3.7.3.3         Full Reuse	27 28 30 32 32 33 33 33 34 36 37

	3.8 2.0	RF PWM	40
	ა.ყ		40
4	BB	PWM	42
	4.1	Modulator	42
	4.2	Direct Filter Connection	44
		4.2.1 Filter Load Modulation	44
		4.2.2 Filter and Matching Network Requirements	45
		4.2.3 Coupling Effects in Cavity Resonators	47
		4.2.3.1 Capacitive Coupling	48
		4.2.3.2 Inductive Coupling	50
		4.2.3.3 Broadband Modeling	52
		4.2.3.4 Summary	53
		4.2.4 Direct Filter Signal Conditioning	54
	4.3	BB PWM PA Circuits	55
		4.3.1 Common Source Parallel PA	55
		4.3.2 Common Drain Parallel PA	59
		4.3.3 Switchable Cascode PA	64
		4.3.4 Source Modulated Amplifier	71
		4.3.5 BPSK DC Energy Recovery	74
		4.3.6 Switched Doherty	77
	4.4	Spectral Considerations	80
	4.5	Summary BB PWM	83
<b>5</b>	$\mathbf{RF}$	PWM	85
	5.1	RF PWM PA Characteristics	86
		5.1.1 Class B $\ldots$	86
		5.1.2 Class $\mathbf{F}$	88
		5.1.3 Class D $\ldots$	89
		5.1.4 Class E $\ldots$	91
		5.1.5 Class DE $\ldots$	93
		5.1.6 Summary RF PWM PA Characteristics	95
	5.2	Modulator	95
		5.2.1 Architecture	96
		5.2.2 Digital RF PWM	99
		5.2.3 Cross Point Estimation	101
		5.2.3.1 Single Branch Cross Point Estimation	102
		5.2.3.2 Minimum Error Signal Generation	105
		5.2.3.3 Single and Double Branch Modulator Performance	105
		5.2.3.4 Delta Sigma Cross Point Estimation Noise Shaping	106
		5.2.3.5 Measurement Results	108
		5.2.3.6 Summary CPE	109
		5.2.4 Driver Properties	109
		5.2.5 Predistortion	112
		5.2.6 Predistortion Measurement Results	116
		5.2.7 Digital Mismatch Compensation	118
	5.3	Summary RF PWM	119
6	Con	aclusion	121
7	Apr	pendix	123
-	<b>PP</b> 7.1	Generic Linear PA	123

7.2	Class I	)	125
	7.2.1	Current Mode Class D RF PWM	125
	7.2.2	Voltage Mode Class D	128
	7.2.3	Inverter Based Voltage Mode Class D	129
7.3	Class I	DE	131
	7.3.1	Class DE With Parallel Compensation Inductance	131
	7.3.2	Class DE Amplifier With Series Inductance	134
7.4	BB PV	<i>WM</i>	135
7.5	Energy	y Recovery	136
	7.5.1	DC Energy Recovery BB PWM	136
	7.5.2	DC Energy Recovery Outphasing	137
	7.5.3	RF Energy Recovery	138
		7.5.3.1 Energy Recovery OFF Period	139
		7.5.3.2 Energy Recovery ON Period	140
		7.5.3.3 Full Energy Recovery	141
	7.5.4	DC BPSK Energy Recovery	142
7.6	Chirei	x1	144
	7.6.1	Quarterwave Chireix Outphasing	144
	7.6.2	Asymmetric Chireix Outphasing	146
	7.6.3	Error Analysis	148
7.7	Equive	alent Load Impedance	149
	7.7.1	Full PWM Modulation	149
-	7.7.2	Partial PWM Modulation	151
7.8	Power	Loss Band Pass Filters	152
7.9	Transi CDE I	stor Parameters	153
7.10	CPE I		194
List of	Public	cations 1	125         at Mode Class D RF PWM       125         ge Mode Class D       128         er Based Voltage Mode Class D       129
List of	Refere	ences 1	.58
List of	Figure	es 1	.78
List of	Tables	3 1	.84
List of	Abbre	eviations 1	.85

### 1. Introduction

Mobile communications have become an indispensable commodity of today's society. The rapid growth of this market sector can be seen best when observing the number of mobile subscriptions over the recent years. From 2005 to 2013 world wide mobile cellular subscriptions have more than tripled (see Fig. 1.1(a)). With the advances in technology and the resulting commercialization of broadband data services, this user group increased even more rapidly. Fig. 1.1(b) shows that the number of mobile-broadband subscriptions from 2007 to 2013 have even more than seven folded. This growth imposes a challenge for the infrastructure providers in terms of bandwidth, but also in terms of energy consumption.



Figure 1.1: Number of mobile subscriptions using broadband services and total number of subscriptions, data retrieved from [1].

To account for this development new standards are continuously developed leading from  $UMTS^1$  [2,3] towards  $LTE^2$  [4,5] and LTE Advanced [6]. To increase the spectral efficiency in order to provide high data throughput these standards employ sophisticated modulation schemes. As a result the Peak to Average Power Ratio (PAPR) of these signals becomes very large. This means that the average signal is far below its maximum value. For the design of the Power Amplifier (PA) this imposes a great challenge, as it has to be designed to provide the maximum output power, although being operated at far lower output powers for most of the time. In traditional PA designs the efficiency is significantly reduced when operating the PA at lower output powers, which leads to a rather low overall efficiency. As a consequence the PA is a major contributor to the energy consumption of a radio base station, as can be seen in Fig. 1.2. Therefore, its efficiency, especially at low output powers, is an important parameter for the total efficiency of a transmitter [7,8].

Besides the use of spectrally efficient modulation schemes the trend goes towards smaller (pico cells) and smaller cell sizes (femto cells) to increase the network capacity [9]. This results in a larger number of base stations with their related energy consumption. Due to the rather inefficient PA and the large number of transmitters, there is a large potential for energy savings by improving the PA efficiency.

Therefore, this thesis is dedicated to the development of efficient transmitter architectures. A main focus is set on switched mode architectures, which are known for their high peak

 $<sup>^13^{\</sup>rm rd}$  generation mobile cellular standard

 $<sup>^24^{\</sup>rm th}$  generation mobile cellular standard



Figure 1.2: Power consumption radio base station, data retrieved from [8].

efficiency [10], as they reduce the overlap between current and voltage waveforms and thus their losses. In that context two different methods, namely Base Band Pulse Width Modulation (BB PWM) and Radio Frequency Pulse Width Modulation (RF PWM), are investigated. Their properties and prospects in terms of energy efficiency are presented. In the context of BB PWM operation the concept of direct filter connection is a very promising solution for highly efficient operation. Different integrated circuit implementations capable of supporting direct filter connection are investigated.

Additionally, aspects like flexibility, reconfigurability, multi band and multi standard operation are of importance. The use of RF PWM goes hand in hand with the trend towards purely digital solutions moving the digital domain closer towards the antenna enabling Software Defined Radio (SDR). The required signal processing for compensating digital and analog effects to achieve best signal quality in RF PWM transmitters is analyzed and corresponding solutions are proposed.

The second chapter provides an introduction on the different PA classes and discusses their properties under rectifier operation. In addition, the requirements on the static efficiency curve for good average efficiency are derived and the loss mechanisms of RF power transistors with a focus on load modulation aspects are discussed.

The third chapter introduces well established as well as recently developed methods for efficiency enhancement in the low power region. A main focus is set on BB PWM operation of the PA, which means that the PA is either operated at maximum power or switched off, being highly efficient under both conditions. The theoretical concept of BB PWM is introduced and further efficiency enhancement by direct filter connection is discussed. Additionally, different energy recovery methods are presented. Also the concept of RF PWM operation, which means that the PWM encoding is applied on the carrier pulse length itself and not on the signal envelope, is covered.

Chapter 4 is dedicated to the development of integrated PA circuits suitable for BB PWM operation in combination with direct filter connection. First, modulation aspects regarding the signal quality are covered. In a next step the load modulation properties of direct filter connection under BB PWM are derived. Based on them, the requirements for the matching network design are determined. In order to provide an optimum design the orientation of the resonance locus is a very important parameter. A design methodology for the direct filter connection based on the load modulation trajectories considering the orientation of the resonance locus is developed. The starting point for the matching network design is the input impedance, respectively resonance locus, of the Band Pass Filter (BPF). Therefore, the impact of the coupling on the reflections of a cavity filter is investigated and considered in the final designs. Four different integrated PA designs are presented. Two of them rely on a parallel concept, while the other two are based on series concepts in order to provide an efficient multilevel PA structure capable of supporting

direct filter connection. In addition, an implementation study of a biphase energy recovery transmitter, as well as a Doherty PA designed for isolated BB PWM operation is presented. Finally, different spectral considerations, such as predistortion and achievable bandwidth, are considered.

The focus of chapter 5 is on RF PWM operation. First the properties of the different PA classes under RF PWM excitation are discussed. Besides the PA efficiency, flexibility and reconfigurability of the modulator are very important parameters. Therefore, the properties of the RF PWM modulator itself are investigated in detail. In order to provide high spectral quality non ideal effects of the modulator have to be compensated. Thus, a main focus is set on the required signal processing. The impact of the discrete time nature of the achievable signal quality due to aliasing effects is covered and a method to mitigate this limitation is presented. Additionally, the properties of real driver circuits, which show nonlinear behavior, are studied. In order to compensate the driver nonlinearity under broadband signal excitation a special predistortion scheme is proposed. Finally, the aspects of digital mismatch compensation and the use of  $\Delta\Sigma$  modulators for resolution enhancement are presented.

### 2. Amplifier Fundamentals

This chapter is dedicated to provide the basics of RF Power Amplifiers. The different amplifier classes and their properties are discussed in detail. Additionally, RF rectifiers are covered, as they are of importance for efficiency enhancement methods using energy recovery. In addition, the loss mechanisms of RF power transistors and their impact on the load pull contours are presented. Finally, the impact of the amplitude distribution on the average efficiency and the resulting requirements for static efficiency behavior is covered.

#### 2.1 Linear Amplifiers

This section provides an introduction to the basic PA classes as A, B and C respectively. The schematic for this three PA classes is the same and provided in Fig. 2.1(a). The transistor  $S_1$  is biased by means of an RF Choke (RFC), which provides the DC bias current and an open circuit at the carrier frequency. To avoid any DC current flowing to the load R, leading to unnecessary power dissipation, a series decoupling capacitor  $C_{DC}$  is required. The higher order harmonics of the drain voltage are short circuited by the parallel shunt resonator, which is high ohmic for the carrier frequency.



Figure 2.1: Class A schematic and efficiency.

The difference between the amplifier classes is in the biasing condition for the drain current of the transistor, which can be written as

$$I_D(\theta) = \max(I_{D,MAX} ((1-k) \ m \sin(\theta) + k); 0),$$
(2.1)

where  $I_{D,MAX}$  denotes the maximum drain current of the transistor, k is the normalized bias current. The parameter m denotes the normalized drain current amplitude, respectively the normalized gate voltage amplitude, and  $\theta$  is the normalized (angular) time variable. The normalized drain current factor k basically defines the class of operation and n controls the output power by varying the drive level. For class A operation the drain bias current is set to half of its maximum value (k = 0.5). The resulting drain current and voltage time domain waveforms for class A operation and maximum output power are depicted in Fig. 2.2(a). It can be seen that the drain current varies in a sinusoidal way around the bias value of 0.5. This class of operation has a significant overlap of voltage and current and thus power dissipation in the transistor, which leads to rather poor efficiency. The DC current consumption remains constant independently of the drive level, as the current never becomes zero. Therefore the DC power consumption remains constant regardless of the output power and the efficiency is directly related to the output power as  $\eta_A = V^2/2$  [10], as depicted in Fig. 2.1(b). The general derivation of the efficiency is provided in section 7.1.



Figure 2.2: Drain voltage and current time domain waveforms for class A, B and C.

A way to improve the efficiency is to reduce the drain current biasing to the threshold of the transistor (k = 0), which is defined as class B [10]. The sinusoidal gate drive is now translated into a half sinusoidal drain current, which leads to a reduced overlap of current and voltage, as can be seen in Fig. 2.2(b). As the DC current consumption linearly depends on the drain current for this operational mode, the efficiency is improved to  $\eta_B = V\pi/4$ , where V denotes the normalized output voltage. The factor  $\pi/4$  is a result of the residual overlap of drain voltage and current.

The fundamental drain current for class B operation is the same as for class A operation. This can be seen in Fig. 2.3(a), which depicts, the DC, fundamental and higher order harmonics amplitudes versus the conduction angle. From Fig. 2.3 it can also be seen that the DC current  $I_{DC}$  is significantly higher for class A biasing than for class B. This leads to the improvement in efficiency as depicted in trace  $\eta$  in Fig. 2.3(b). It can also be seen that the output power is highest for class AB operation, meaning that the biasing is somewhere inbetween class A and class B.



Figure 2.3: General harmonic contribution and power, gain and efficiency versus conduction angle.

When moving to even smaller conduction angles than class B operation the efficiency is further improved as the overlap current and voltage decreases, as can be seen in Fig. 2.2(c) for a conduction angle of  $\pi/2$ . This operational region is referred to as class C. Unfortunately, the output power  $P_L$  rapidly decreases for short conduction angles as the fundamental component of the drain currents gets very low. On the other hand the required gate drive swing increases, which leads to higher input power, causing a degradation in gain. This effect in general leads to low Power Added Efficiency (PAE) for class C operation making this mode of operation less attractive for single PAs.

Finally it shall be noted that only the ideal class A and class B circuit show linear behavior over the whole operational range. The gain under class AB bias and especially class C biasing experiences gain expansion before compressing at maximum output powers, which leads to nonlinear behavior.

#### 2.2 Class J

In practice a perfect short circuit of the higher order harmonics can hardly be achieved or may be even undesired. Usually the parasitic drain source capacitance  $C_{DS}$  is resonated out at carrier frequency and is supposed to provide the desired short circuit at the carrier harmonics. At the second harmonic this short circuit is normally not very pronounced and one can use this capacitive termination to contribute to the drain voltage waveform. This has led to the class J amplifier [11–17], which essentially provides an inductive reactance at the fundamental and a capacitive termination for the second harmonic. The basic schematic of a class J amplifier can be found in Fig. 2.4(a).



Figure 2.4: Class J schematic and time domain waveforms.

According to [11] the general drain voltage for maximum voltage swing (N=1) can be defined by

$$V_D = (1 - \beta \cos(\theta)) (1 - \alpha \sin(\theta)), \qquad (2.2)$$

while  $\beta$  denotes the maximum voltage swing and the parameter  $\alpha$  defines the reactive termination contribution. Fig. 2.4(b) depicts the time domain waveforms for  $\beta = 1$  and  $\alpha = 1$  as an example. The maximum drain voltage is higher than for the class B operation, where it never exceeds two times the supply voltage. The resulting efficiencies are equivalent to the one of a class B amplifier [11].

#### 2.3 Class F

To achieve higher efficiency than the class B PA the class F circuits use proper higher order terminations to shape the rectangular drain voltage. To achieve this behavior the even order harmonics require a short circuit termination, while the odd order harmonic terminations have to be left open [18–26]. Ideally this is done by means of quarter wave line which can also be used for biasing purposes, as depicted in Fig. 2.5(a). To have an open circuit for the odd order harmonics the load has to be connected by a series resonator. The

resulting drain current is half sinusoidal, but the drain voltage is rectangular, leading to basically no voltage and current overlap and highly efficient operation. The corresponding time domain waveforms are provided in Fig. 2.5(b).



Figure 2.5: Class F schematic and time domain waveforms.

When exchanging the harmonic terminations and providing an open circuit for all even order harmonics and a short circuit for all odd order harmonics the shape of drain current and voltage are exchanged as well. This leads to the inverse class F amplifier [27–32], whose schematic is depicted in Fig. 2.6(a). The position of the quarter wave line is different compared to the class F PA. The series resonator provides an open circuit at the input of the quarter wave line for all carrier harmonics. This open circuit is transformed to a short circuit for all odd order and remains an open circuit for all even order harmonics. This way the desired harmonic terminations can be achieved. To not interfere with these terminations the biasing is done using an RFC, which provides high impedance for all frequencies except DC. The resulting drain current and voltage waveforms are depicted in Fig. 2.6(b).



Figure 2.6: Inverse class F schematic and time domain waveforms.

Basically the shapes of drain current and voltage are exchanged compared to the class F PA. The drain current is now rectangular and the drain voltage is half sinusoidal. As there is no overlap of current and voltage also this circuit shows 100 % efficiency in theory. In practice the parasitic capacitance  $C_{\rm DS}$  is present and has to be properly compensated for the higher order harmonics to provide the desired terminations and highly efficient operation.

#### 2.4 Class D

Two different versions, namely the current mode and the voltage mode of the class D amplifier exist. The class D amplifier family [33,34] is in general an extension of the class F family to the push-pull configuration. The current mode class D amplifier, whose schematic is depicted in Fig. 2.7(a), basically relates to the inverse class F amplifier. Instead of using a quarter wave line to transform the series resonator of the inverse class F into a parallel resonator, the current mode class D amplifier [35–38] directly uses a parallel resonator located between the two transistors. In practice the load will be connected by a Balanced Unbalanced (BALUN) circuit to provide a single ended output. The even order harmonics of the drain voltage waveform will be canceled out due to the 180 ° out of phase push-pull operation. Therefore no special care about their termination has to be taken.



Figure 2.7: Current mode class D schematic and time domain waveforms.

The resulting drain currents are rectangular and the drain voltage waveforms show a half sinusoidal shape, as can be seen in Fig. 2.7(b). The waveforms of a single transistor are essentially the same as for the inverse class F in Fig. 2.6(b) and thus also the current mode class D has a theoretical efficiency of 100 % as ideally there is no overlap of current and voltage on the switch. The general mathematical definition of time domain waveforms can be found in section 7.2. In practice the parasitic drain source capacitance  $C_{DS}$ , that would cause switching losses, can be absorbed into the parallel resonator maintaining Zero Voltage Switching (ZVS).

The voltage mode class D amplifier [39–42] uses a series resonator connected via a BALUN instead of the parallel resonator, as depicted in Fig. 2.8(a). The resulting rectangular drain voltage waveform due to the series resonator and the BALUN in push-pull drive can be seen in Fig. 2.8(a). The drain current has a half sinusoidal shape and essentially looks the same as for the class F amplifier.

Instead of using a push-pull architecture an inverter based architecture as given in Fig. 2.9(a) can be used [43–50]. The drain voltage is either pulled to ground or to the supply voltage by the transistors forcing a rectangular signal. To not dissipate any energy in the higher order harmonics and to allow for the rectangular drain voltage, the load is decoupled by a series resonator. The resulting drain currents are given in Fig. 2.9(b) and are half sinusoidal and the same as for the conventional voltage mode class D.

The inverter based voltage mode class D has the advantage of directly providing a single ended output signal mitigating the demand for a BALUN. But on the other hand it requires a P-channel high side switch, which is normally avoided due to its worse RF properties compared to N-channel devices. As for RF power transistors usually only the N-channel is available the resulting gate control circuitry for the high side switch will



Figure 2.8: Voltage mode class D schematic and time domain waveforms.



Figure 2.9: Inverter based voltage mode class D schematic and time domain waveforms.

become complex and power consuming. Also the  $C_{DS}$  switching losses will degrade the amplifier efficiency in practice.

#### 2.5 Class DE

Instead of charging and discharging the parasitic capacitance  $C_{DS}$  actively, which leads to large power dissipation, it can be made an integral part of the operational concept. By introducing a dead time between the two periods and adding a series reactance, as depicted in Fig. 2.10(a), to the load, ZVS can be established. This operational mode is known as the class DE amplifier [51–64]. The parasitic capacitance is now charged and discharged by the load current and not the transistors. Fig 2.10(b) provides the drain currents and voltage waveforms. It can be seen that the drain voltage at the beginning of the ON period already has reached the correct value and ZVS occurs. Also the derivative of the drain voltage is zero at this point in time and thus also Zero Current Switching (ZCS) is possible. Thus no switching losses occur and the circuit allows perfect DC to RF conversion. In section 7.3.2 the derivation of the matching network parameters is provided.

A drawback of the conventional class DE amplifier using a series compensation inductance is its limited load modulation [51] as a certain load current has to flow to charge/discharge the capacitances enabling ZVS. Instead of using a series compensation inductance a parallel inductance as depicted in Fig. 2.11(a) can be used. The corresponding calculation of the parameter values is provided in section 7.3.1. Using the parallel compensation



Figure 2.10: Conventional class DE schematic and time domain waveforms.

inductance offers the possibility of efficient load modulation over a wider dynamic range compared to the conventional class DE amplifier [S1].



Figure 2.11: Voltage mode class DE using parallel compensation inductance.

The big disadvantage of the circuits in Fig. 2.10(a) and Fig. 2.11(a) is the inverter like structure that requires either complementary transistors or complex and power consuming gate signal generation for the high-side switch. This problem can be solved by using a push-pull variant as depicted in Fig. 2.11(b). It has the advantage of using two low-side switches, which are easier to implement for high frequencies. The disadvantage is that the transformer's parasitic parallel inductance  $L_P$  makes the conventional circuit design based on the series compensation inductance more difficult. But this parallel inductance  $L_P$  can also be used instead of the series inductance  $L_S$  to achieve class DE operation. Therefore the parasitics of the transformer can be absorbed into the matching network and the load modulation properties can be preserved.

#### 2.6 Class E

Since the introduction of the class E amplifier in 1964 [65] its properties and aspects have been studied in numerous papers [66–79]. The basic circuit diagram of the class E PA is given in Fig. 2.12(a). It consists of a biasing inductance L and a shunt capacitance  $C_{DS}$ , which may only be the parasitic capacitance of the transistor for high frequency operation. The load R is connected by a series resonator and a series reactance  $X_S$ . The values of the elements depend on many design parameters. One of the most important ones is the q-value, which basically defines the ratio of the resonance frequency of the parallel circuit (L and  $C_{DS}$ ) to the operational frequency [80–82].



Figure 2.12: Voltage mode class E schematic and time domain waveforms.

The time domain waveforms of the class E amplifier for q = 1.3 [81] are depicted in Fig. 2.12(b). When observing the time domain waveforms it can be seen that ZVS and also ZCS occurs, leading to ideally 100% efficiency. A drawback of the class E amplifier is its relative high peak drain voltage with respect to the DC supply voltage. The maximum drain voltage that occurs is three times larger, or even higher depending on the configuration, than the supply voltage, as can be seen from Fig. 2.12(b).

#### 2.7 Controlled Rectifiers

This section is dedicated to cover the field of circuits for RF rectification [83–89]. Some efficiency enhancement concepts relying on RF energy recovery require efficient rectifiers and therefore it is important to discuss their properties. Several circuit variants will be discussed starting from the basic rectifier circuits and leading to highly efficient switched mode rectifiers.

#### 2.7.1 Class B Rectifier

The schematic of the class B rectifier is given in Fig. 2.13. The higher order harmonics are short circuited and the gate is actively controlled in the same manner as for a class B amplifier. The power comes from an RF source showing an input impedance  $Z_{IN}$ , which is matched to the transistor properties.

As a result of the active control there is a large overlap of the drain current, especially positive current, which leads to a big power dissipation and low efficiency. Therefore the class B rectifier using active gate control is not a good candidate for efficient RF rectification.

But when using a self aligned diode rectifier, as depicted in Fig. 2.14, the drain current is similar to the one of a class C PA with similar (high) efficiency behavior [88]. This circuit essentially behaves like a common one way rectifier.

#### 2.7.2 Class D Rectifier

A switched mode PA structure has major advantages in terms of efficiency not only for PAs, but also for rectifier operation. As the drain voltage during the conduction period is



Figure 2.13: Class B rectifier.



Figure 2.14: Class C rectifier (self aligned diode).

ideally zero, it makes no difference if the drain current is positive or negative. The inverter based class D amplifier is a textbook example of a switched mode PA. Fig. 2.15(a) depicts a class D circuit that can also be used for active or self aligned rectification. From the time domain waveforms in Fig. 2.15(b) it can be seen that the drain currents are negative and thus rectification occurs.

The circuit basically behaves as an ideal full wave rectifier with rectangular drain voltage, which is possible due to the series resonator. As there is no overlap between voltage and current the rectifier is ideally 100 % efficient. In practice the same problems with  $C_{DS}$  losses as for the class D PA will occur leading to decreased efficiency. It shall be noted that the current and voltage class D rectifiers are not discussed as they behave similarly to their single ended class F counterparts that will be covered in section 2.7.5.

#### 2.7.3 Class DE Rectifier

To overcome the problems with the  $C_{DS}$  losses, which impose a major limitation for efficiency, the class DE circuit is a good candidate [90–92]. Its schematic is depicted in Fig. 2.16(a) and its operational principle is already discussed in section 2.5. The main concept is to leave a dead time between the two ON states of the transistor and let the load current charge and discharge the parasitic capacitances. This leads to zero voltage switching and thus no capacitive losses as can be seen in Fig. 2.16(b).



Figure 2.15: Class D rectifier.



Figure 2.16: Class DE rectifier.

The time domain waveforms are basically inverted, both in amplitude and time, compared to PA operation (Fig. 2.10(b)) [93, 94]. Only the ON resistance of the switches causes energy dissipation and reduces the theoretical 100 % efficiency.

#### 2.7.4 Class E Rectifier

Instead of using two transistors either in push-pull or complementary arrangement it is also possible to achieve perfect RF to DC conversion with a single ended class E circuit [91,92,95-98] as depicted in Fig. 2.17(a).

Similar to the class DE rectifier the time domain waveforms are basically inverted both in time and amplitude [93,94]. Nevertheless there is no overlap between drain current and voltage leading to highly efficient operation.

#### 2.7.5 Class F Rectifier

Another single ended circuit besides the class E is the class F circuit family. The schematic of the standard class F rectifier can be found in Fig. 2.18(a). Similar to the class F PA case the time domain waveforms of current and voltage of the class F rectifier are presented in Fig. 2.18(b). These waveforms are similar to the voltage mode class D rectifier.

Beside the class F circuit the inverse class F, as depicted in Fig. 2.19(a), can also be used



(b) Time domain waveforms

Figure 2.17: Class E rectifier.



Figure 2.18: Class F rectifier.

[89]. Its harmonic terminations are set to achieve a half sinusoidal drain voltage and a rectangular drain current. This leads to no overlap between drain voltage and current, as depicted in Fig. 2.19(b) and ideally 100% efficiency.

As the matching network at the fundamental carrier frequency basically behaves like a parallel resonator, which is not desired for some concepts such as BB PWM, it would be beneficial to have a structure that behaves like a series resonator. For inverse class F PAs the matching network can be modified to do so, while maintaining the same drain current waveforms. The resulting PA circuit in Fig. 2.6(a) can also be used in the same way for rectification.

#### $\mathbf{2.8}$ Comparison

In this section the main circuit parameters of the different PA classes are summarized. They are compared in terms of their maximum efficiency  $\eta_{MAX}$ , maximum drain voltage  $V_{D,MAX}$ , and their Power Utilization Factor (PUF). The PUF defines the ratio of the output power with respect to the maximum drain current and voltage compared to the class A output power of the same device [10]. In addition the mode of operation (linear or switched) is indicated by the field Switched Mode Power Amplifier (SMPA). The circuits abilities to operate also as rectifier are evaluated as well. The results are summarized in Tab. 2.1 below.



(a) Schematic

(b) Time domain waveforms

class	$\eta_{MAX}$	SMPA	V <sub>D,MAX</sub>	PUF	rect
А	50%	no	$2V_{DC}$	1	no
В	78.5%	no	$2V_{DC}$	1	~
С	< 100 %	no	$2V_{DC}$	«1	yes
J	78.5%	no	$> 2V_{DC}$	1	~
CM D	100%	yes	$\pi V_{DC}$	$4/\pi$	yes
VM D	100%	yes	$2V_{DC}$	$4/\pi$	yes
DE	100%	yes	$2V_{DC}$	$<4/\pi$	yes
Е	100%	yes	$3-4V_{DC}$	<1	yes
F	100%	yes	$2V_{DC}$	$4/\pi$	yes
$F^{-1}$	100%	yes	$\pi V_{DC}$	$4/\pi$	yes

Figure 2.19: Inverse class F rectifier.

Table 2.1: Overview amplifier classes.

#### 2.9 Loss Mechanisms RF Transistors

In this section the main loss mechanisms and their impact on the load pull contours will be discussed. A particular focus is set on the influence on the load modulation capabilities, as this is an important property for several efficiency enhancement techniques such as Doherty or Chireix PAs. Based on a simplified transistor model the influence of the  $R_{on}$ , the biasing current, parasitics Q-factor and parallel losses as well as the gain are presented. When performing a load pull, amplifier parameters like efficiency output power and gain are evaluated over a range of different load impedances. Therefore, the load impedance is modified as depicted in Fig. 2.20(a). The resulting contours are usually plotted in a smith chart to easily identify the spots of maximum power or efficiency.

For the PA a simple model, as depicted in Fig. 2.20(b) is considered [S2]. It consists of the gate input with its parasitic capacitance  $C'_{GS}$  and series resistance  $R_G$ . The gate voltage  $V_G$  controls the drain current  $I_D$  source which forms together with its  $R_{on}$  the intrinsic transistor. Additionally, the drain source capacitance  $C'_{DS}$  and the drain resistor  $R_D$  are considered.

The higher order harmonics are assumed to be short circuited such that class B operation occurs. For the calculation of the intrinsic transistor losses  $P_T$  the equations derived in Appendix 7.1 are considered. The corresponding efficiency of the intrinsic transistor can



Figure 2.20: Load pull setup and transistor model.

be calculated by

$$\eta_D = \frac{P_L}{P_L + P_T}.\tag{2.3}$$

The effect of  $R_{on}$ , respectively the knee voltage can be considered by

$$\eta_{knee} = \frac{R_L}{R_L + R_{on}(k, n)}.$$
(2.4)

The effective  $R_{on}$  depends on the biasing (k) and the normalized drive level (n). Therefore, it is easier to reduce the normalized drain voltage swing N and current n in (7.6) accordingly to account for the drain efficiency degradation.

The drain resistance  $R_D$  defines the Q-factor of the parasitic drain source capacitance  $C_{DS}$  and will for this analysis be parameterized by it. In order to ease the calculation of the influence of  $C_{DS}$  and the series resistor  $R_D$  on the efficiency, they can be transformed into a parallel circuit as depicted in Fig. 2.20(c). Although mathematically not 100 % correct, it was found to be a good approximation [99]. The resulting parallel shunt resistance is DC wise decoupled to avoid static losses and may also be modified to account for substrate losses [100]. The efficiency degradation due to the equivalent parallel resistor  $R_{D,P}$  of  $C_{DS}$  can be calculated by

$$\eta_P = \frac{R_{D,P}}{R_{D,P} + R_{L,P}},$$
(2.5)

where  $R_{L,P}$  denotes the effective parallel load resistance of  $Z_L$ .

The influence of the gain on the PAE can be accounted for by

$$\eta_G = 1 - \frac{1}{G}.$$
 (2.6)

To discuss the impact of the mentioned parasitics a frequency of  $2.65 \,\text{GHz}$ , a normalized capacitance  $C_{\text{DS}}$  per maximum current of  $1.65 \,\text{pF}/\text{A}$  and a Q-factor of 40 for  $C_{\text{DS}}$  were selected.

The knee voltage was assumed to have 10% of the nominal supply voltage (28 V). The biasing inductance  $L_P$  in Fig.2.20(c) is set to resonate  $C_{DS}$  out such that the resulting load pull contours are symmetrical around the real axis.

The influence of the different effects on the load pull contours is depicted in Fig. 2.21. In Fig. 2.21(a) the impact of the ON resistance of the transistor can be seen. Ideally any load resistance large enough to get maximum voltage swing would result in the theoretical class B efficiency of 78.5% (Fig. 2.21(a)). But the ON resistance of the transistor reduces the efficiency, especially for smaller load resistor values (higher power). Thus the ON

resistance of the transistor moves the points of maximum power and efficiency apart. This can be seen in Fig. 2.21(b), which depicts the drain efficiency for class B biasing considering the ON resistance, respectively knee voltage. It can be seen that for very large resistance values towards the open circuit maximum efficiency is achieved.

Considering a realistic biasing with 1% quiescent current (or leakage) reduces the efficient for very high resistance values. Fig. 2.21(c) illustrates this effect.

Beside the biasing, the influence of the parallel resistor  $R_{D,P}$ , which accounts for the losses due the limited Q-factor of  $C_{DS}$ , significantly reduces the range of impedances for high efficiency operation. In Fig. 2.21(d) the influence of this effect is shown. It can be seen that the region of highest efficiency has shrunk and its maximum is reduced. The Q-factor of  $C_{DS}$  might in practice be further reduced by any series resistance (eg. bondwires). Also the biasing inductor will have a limited Q in practice. It shall be noted that the Q-factor of 40 was chosen as an example and may vary according to transistor technology, generation and operational frequency.



Figure 2.21: Theoretical load pull contours and impact of loss mechanisms.

When considering the gain, which has an influence on the PAE, the efficiency contours are shifted again towards higher impedance values, as can be seen in Fig. 2.21(e).

The drain biasing with  $L_P$  is in practice ascribed to the matching network and usually considered as a part of the load impedance. Fig. 2.21(f) shows its influence on the position of the load pull contours.

It can be summarized that the ON resistance of the transistor and the gain basically move the point of maximum efficiency towards higher resistance values. This behavior is beneficial for systems that rely on load modulation such as the Doherty amplifier. These systems require high efficiency over a wide range of load impedances to work properly. Unfortunately the biasing current and especially the Q-factor of the parasitic drain capacitance  $C_{DS}$  limit this range significantly. Thus it can be said that the Q-factor of the parasitic capacitance plays an important role for the load modulation capabilities of a transistor and it imposes a limit for the high efficiency region.

Beside the behavior at a single frequency the evolution of the PAE versus frequency is interesting to observe. Therefore, Fig. 2.22(a) depicts the PAE as a function of the real valued axis of the load reflection coefficient and the frequency. It can be seen that for very low frequencies the transistor operates efficiently over a wide range of impedances. In this region the transistor may be oversized to reduce the influence of the knee voltage and to achieve better efficiency. But when moving to higher frequencies the maximum efficiency region is significantly reduced, moving it close to the maximum output power. This trend highlights that using a larger transistor to reduce the impact of  $R_{ON}$  becomes ineffective at higher frequencies. The fact that the region of highest efficiency is reduced for higher frequencies reduces the useful load modulation range and thus limits the achievable efficiency enhancement.



Figure 2.22: Theoretical efficiency versus sweep parameters.

Reducing the conduction angle from class B to class C is a way to improve the drain efficiency. Fig. 2.22(b) depicts the influence of the biasing on the PAE. It presents the dependency of the efficiency on the real axis of the load reflection coefficient for a carrier frequency of 2.6 GHz versus different biasing conditions. It can be seen that the efficiency maximum occurs for slight class C biasing. For class B operation (k=0) the efficiency is only slightly smaller, while it is significantly reduced for class A (k=0.5) operation. From Fig. 2.22(b) it can be concluded that deep class C biasing is not useful, as the C<sub>DS</sub> related losses (R<sub>D,P</sub>) and the reduced gain lower the PAE.

#### 2.10 Average Efficiency

In this section the calculation of the average drain efficiency under excitation with modulated signals will be discussed. Additionally the design requirements to achieve good average efficiency will be conducted.

The average efficiency can be calculated according to (2.7) [101].

$$\eta_{\text{AVG}} = \frac{\int\limits_{0}^{V_{max}} V^2 \text{pdf}(V) dV}{\int\limits_{0}^{V_{max}} \frac{V^2}{\eta(V)} \text{pdf}(V) dV} = \frac{\int\limits_{0}^{V_{max}} \text{w}(V) dV}{\int\limits_{0}^{V_{max}} \frac{\text{w}(V)}{\eta(V)} dV},$$
(2.7)

where  $V_{max}$  denotes the maximum voltage and pdf(V) the amplitude Probability Density Function (PDF) of the modulation signal. The numerator of the equation normalizes the result, while the denominator depends on the PDF and the weighting function

$$\mathbf{w}(V) = V^2 \mathrm{pdf}(V). \tag{2.8}$$

A large weighting function indicates the regions which contribute the most to the average efficiency. Therefore high efficiency within this regions is important for the average efficiency.

Fig. 2.23(a) provides PDFs with different PAPRs. The corresponding weighting functions for are plotted in Fig. 2.23(b).



Figure 2.23: Probability density and resulting weighting functions.

From Fig. 2.23(b) it can be seen that the region of intermediate output power is most important for the average efficiency. To show these three different efficiency curves are considered in Fig. 2.24(a). A class B efficiency curve  $\eta_{\rm B}$ , a constant efficiency  $\eta_1$  and efficiency curve  $\eta_2$  adapted to the weighting function are considered.



Figure 2.24: Example efficiency functions and resulting average efficiency.

The resulting average efficiencies versus different PAPRs are depicted in Fig. 2.24(b). It can be seen that the average efficiency for the class B amplifier heavily depends on the PAPR, while the constant efficiency curve  $\eta_1$  is not at all affected by the PAPR of the signal. But also the efficiency curve  $\eta_2$  optimized for the weighting function barely varies with the PAPR and is only slightly smaller than the maximum efficiency  $\eta_1$ . Thus neither the efficiency at peak power nor for very low output powers have a big impact on the average efficiency. It can be concluded that the intermediate region for a normalized signal magnitude between 0.2 and 0.8 is most important for the average efficiency and should be optimized in amplifier designs.

# 3. Efficiency Enhancement Concepts

In this chapter several efficiency enhancement concepts will be discussed, briefly touching well established concepts such as Doherty and Chireix but also covering topics like envelope tracking. Emphasis is put on the theory of Base Band Pulse Width Modulation (BB PWM), which is often also referred to as burst mode. In that context a particular focus is set on the effects and perspectives of isolated load and direct filter connection. The properties of multilevel and multiphase structures will also be covered. Additionally, several concepts of energy recovery methods are evaluated. Finally, the theoretical background of RF PWM operation is presented. It shall be noted that for a fair comparison a class B maximum efficiency is assumed for all concepts unless otherwise noted.

#### 3.1 Doherty

The concept of the Doherty amplifier exists for many decades and was introduced by W. H. Doherty in 1936 [102]. Recently the Doherty PA is experiencing a revival [103–107]. Several different variants exist reaching from integrated Doherty PAs [108–113] to highly efficient circuits using class E [114, 115] or class F [116–118] amplifier concepts at high output power levels. Further improvement is made by using separate controlled inputs [119,120] or even going to 3 way [121–123] or 4 way architectures [124]. The basic circuitry is depicted in Fig. 3.1(a) and essentially consists of two amplifiers connected by a quarter wave line.



Figure 3.1: Doherty block diagram and efficiency.

The upper main amplifier is constantly active and the peak amplifier is only activated for higher powers and performs load modulation of the main PA enabling efficient operation. For low power only the main PA is active and the peak PA ideally provides an open circuit not influencing the operation. The impedance of the quarter wave line is set such that the main PA sees twice its optimum load impedance for this operational region. When increasing the drive of the main PA its drain current I<sub>main</sub> continuously increases as depicted in Fig. 3.2(a). The drain voltage V<sub>main</sub> raises accordingly until the main PA saturates. In this region the PA behaves like a regular class AB amplifier, as its load resistance remains constant (Fig. 3.2(c)).



Figure 3.2: Doherty control, output power and partial efficiency.

Once the main PA is saturated, the peak PA starts operating and provides additional output power. Its current contribution  $I_{peak}$  at the summation point increases the total load current and hence the load voltage, respectively the peak amplifier's drain voltage. The load current contribution of the main PA remains constant. Due to the quarter wave line the drain voltage of the main PA stays constant and the saturated operation condition is kept. On the other hand the drain current of the main PA increases with higher load voltage swing, leading to linearly increased power contribution of the main PA, as depicted in Fig. 3.2(b). The efficiency of the main PA remains at its maximum in this region, as the main PA experiences maximum drain voltage swing. When the peak PA starts to operate its efficiency is low, as the voltage swing is limited, but the efficiency linearly increases with the load voltage. In the back off range the power contribution of the peak PA compared to the main PA is small and its lower efficiency has only a limited impact on the overall efficiency curve depicted in Fig. 3.1(b). This way highly efficient operation over a wide power range is possible.

#### 3.2 Linear Amplification Using Nonlinear Components

The concept of Linear Amplification Using Nonlinear Components (LINC) is based on two amplifiers, whose powers are combined using an isolated combiner [125]. The PAs are operated in saturation and the output power is controlled by modifying the phase  $\varphi$ of the two PAs with respect to each other [126]. Fig. 3.3 depicts the block diagram of a LINC transmitter using a Wilkinson combiner to provide isolation. Due to the isolated combining scheme the load impedance for each PA remains constant regardless of the driving conditions. This leads in combination with the saturated drive of the PAs to a constant output power with high drain efficiency. While the in-phase power is transfered to the load, the out of phase components are dissipated in the resistor R<sub>OUT</sub>. A result of the constant input power is the rather poor efficiency curve that depends in a square relation to the output voltage, as depicted in Fig. 3.3(b).

To improve this efficiency curve several different architectures based on multiple levels for the outphasing operation are known. One way is to reduce the input drive [127], which results in a class B efficiency trend. Using multiple supply levels on the other hand results in high efficiency for the different levels [128]. By also using asymmetric supply combinations [129–132] the number of levels and thus the efficiency can be enhanced.

#### 3.3 Chireix

The Chireix combiner was proposed by H. Chireix in 1935 [133]. Instead of using an isolated combiner structure as for LINC the Chireix combiner uses the interaction of both PAs, which results in load modulation for the single branches. The basic arrangement



Figure 3.3: LINC.

[134–140] is shown in Fig. 3.4(a) and consists of two PAs connected to the common load by two quarter wave lines. The PAs are operated in saturation by keeping the drain voltage constant. The output power is controlled by modifying the relative phase  $\varphi$  of the control signals.

When using only quarter wave lines the resulting impedances for each PA branch shows a significant reactive component causing overlap of drain current and voltage and leading to efficiency degradation. In order to compensate for this reactive components the shunt reactances  $X_P$ , as depicted in Fig. 3.4(a), are employed.



Figure 3.4: Chireix block diagram.

The resulting impedance trajectories for the single PA branches exemplified for an efficiency peak at 6 dB back off are provided in Fig. 3.5(a). It can be seen that due to the shunt compensation reactances the load modulation trajectories are close to the real axis, enabling efficient operation [141]. The component values for the efficiency peak at distinct back off values have been determined using the equations derived in section 7.6.1. The corresponding efficiency curve for 6 dB as well as for 9 dB back off are plotted in Fig. 3.5(b).

Instead of using shunt elements  $X_P$ , which can be lossy in practice, the Chireix combiner can also be built by using series transmission lines only [142], leading to an architecture as depicted in Fig. 3.4(b). The load modulation behavior for proper dimensioning (see section 7.6.2) is exactly the same as for the conventional Chireix architecture shown in Fig. 3.4(a). A drawback of the Chireix, as well as the LINC, is the fact that the generation of low output signals is based on cancellation and that they are therefore very sensitive to parameter mismatch (supply, TL, phase). This leads to a limited dynamic range (see section 7.6.3). Using the outphasing approach until the maximum efficiency in back off is reached and then lowering the drive level to achieve low output powers has advantages both in terms of dynamic range and efficiency [143].


Figure 3.5: Chireix load modulation trajectories and efficiency.

Practical applications of the Chireix combiner reach from low power CMOS circuits [40, 144] over high power systems using class E PAs [77, 145, 146] to 4 way combiners [147–149].

# 3.4 Envelope Tracking / Envelope Elimination and Restoration

Another way to improve amplifier efficiency is to modify the supply voltage according to the output power in order to reduce current and voltage overlap. In general two concepts, namely Envelope Tracking (ET) [150–155] and Envelope Elimination and Restoration (EER) [156–158], exist. EER was suggested by R. Kahn in 1952 [159,160] and is therefore often referred to as Kahn transmitter. Both concepts ET, as depicted in Fig.3.6(a), and EER, as shown in Fig. 3.6(b), essentially rely on the use of an envelope PA, which modifies the supply voltage of the actual PA according to the output of the Envelope Detector (ED). This way the PA is always operated with maximum voltage swing and thus maximum efficiency. The main difference between ET and EER lies in the input signal generation. For envelope tracking the amplitude information is still contained in the PAs driving signal. This leads to a combined control of the output power by the envelope PA and the driving signal. For EER the amplitude information is only controlled by the envelope PA and the driving signal only contains the phase information  $\varphi$  extracted by the limiter.



Figure 3.6: Block diagrams of envelope tracking and envelope elimination and restoration.

In practical implementations the required bandwidth of the envelope PA due to bandwidth expansion of the envelope signal imposes a challenge. Also the fact that the system efficiency is the combined efficiency of envelope and regular PA tightens the requirements.

# 3.5 Load Modulation

Instead of changing the supply voltage it is also possible to modify the load impedance to control the output power. The general block diagram for load modulation is provided in Fig. 3.7. The impedance  $Z_L$  provided to the PA is varied by tuning the Output Matching Network (OMN) [17,161–171]. This way the power delivered to the load can be controlled. In practice the variable matching network is mainly implemented using varactors, which require control signals with very high voltage swings and complex control circuitry. Also the synthesis of broadband load modulation networks imposes a big challenge.



Figure 3.7: Load modulation block diagram.

# 3.6 Base Band PWM

In this section the concept of Base Band Pulse Width Modulation (BB PWM) will be introduced. First the properties of BB PWM using an isolated combiner and single level PA will be discussed. Furthermore, the perspectives of direct filter connection based on a series or parallel resonator will be covered. Different multilevel PA structures and their behavior in combination with BB PWM is treated. In addition the multiphase concept is introduced and compared to the multilevel approach.

## 3.6.1 Isolated Broadband Load

The efficiency of conventional amplifiers is significantly reduced for lower output powers. One way to overcome this efficiency degradation is to drive the PA only at its most efficient operating points when the transistor is either working close to saturation with maximum voltage swing and thus maximum efficiency or completely switched off, not dissipating any power at all [38, 172–174]. To drive the PA in such a way a special modulator arrangement, as depicted in Fig. 3.8(a) is required. The input amplitude a(t) is encoded in the MODulator (MOD) by means of Pulse Width Modulation (PWM) and upconverted using the Local Oscillator (LO) signal c(t). The phase information is introduced by (phase) modulating the carrier signal c(t).

Unfortunately the PWM introduces modulation sidebands close to the desired inband signal, as shown in Fig. 3.8(b) depicting the Power Spectral Density (PSD) versus the normalized modulation frequency. To only transmit the desired information a BPF is used to filter out the inband signal. In order to avoid any feedback of the filter to the PA and to provide a broadband load, a circulator configured as isolator, as depicted in Fig. 3.8(a), can be used. This way a perfect match not only for the inband signal, but also for the modulation sidebands is provided.

Fig. 3.9(a) depicts the drain voltage waveforms for a class B PA with broad band load under PWM excitation with 50% duty cyle. During the first period (ON) the PA is operated with maximum drain voltage swing and thus maximum power and efficiency. During the second period (OFF) the PA is switched off and no power is dissipated at all, as no drain current is flowing. The information is now encoded in the bursts at the drain of the transistor, therefore this type of operation is also referred to as burst mode operation.



Figure 3.8: Block diagram of broadband load operation.



Figure 3.9: Drain voltage and current for PWM and  $\Delta\Sigma$  operation for an isolated load and efficiency.

By using this efficient coding scheme high drain efficiency of the PA can be achieved. The drain efficiency  $\eta_D$  is defined as

$$\eta_{\rm D} = \frac{P_{\rm D}}{P_{\rm DC}},\tag{3.1}$$

where  $P_D$  denotes the total power delivered from the drain to the (broadband) load and  $P_{DC}$  is the total DC power consumption. It shall be noted that the total power delivered from the drain does not only include the inband power delivered to the load, but also the energy of the modulation sidebands dissipated in the termination resistor.

Although the drain efficiency is high, the total efficiency curve of the amplifier  $\eta_A$ , as depicted in Fig. 3.9(b), degrades to the one of a class B amplifier. This is caused by the fact that also the modulation sidebands see the optimum load impedance and thus their power (P<sub>SB</sub>) is dissipated in the sideband termination resistor. The coding efficiency defined by (3.2) describes this effect by taking the ratio of the power delivered to the load P<sub>L</sub> versus the power delivered from the PA P<sub>D</sub> into account.

$$\eta_{\rm C} = \frac{{\rm P}_{\rm L}}{{\rm P}_{\rm D}} \tag{3.2}$$

As a significant amount of energy is dissipated in the modulation sidebands, the calculation of the coding efficiency, as provided in section 7.4, results in a linear dependence on the generated load voltage. Therefore the overall amplifier efficiency, which is defined as

$$\eta_{\rm A} = \eta_{\rm D} \eta_{\rm C},\tag{3.3}$$

is significantly reduced. Although the PA itself is operated highly efficient, the class B efficiency trend is not exceeded.

It shall be noted that the above considerations are also valid for  $\Delta\Sigma$  noise shaping, if all spectral components see the optimum load impedance. Exemplary time domain waveforms for 50 % relative output voltage and  $\Delta\Sigma$  operation are plotted in Fig. 3.9(c).

## 3.6.2 Direct Filter Connection

To mitigate the losses in the modulation sidebands, the isolator can be removed connecting the filter directly [175–179]. The BPF can either be implemented as a series or parallel resonator, each of them showing different effects. Their properties will be discussed in the following two subsections. In addition, the possible efficiency enhancement by combining the series resonator with a switch is also covered.

#### 3.6.2.1 Parallel Resonator

The use of a parallel resonator results in the block diagram depicted in Fig. 3.10.



Figure 3.10: Block diagram of direct filter operation.

The parallel resonator provides a short circuit for the modulation sidebands and causes the drain voltage to be demodulated. This can be seen in Fig. 3.11(a), which depicts the drain voltage and current for a 50% duty cycle signal. The drain current pulses are demodulated by the parallel resonator resulting in a constant drain voltage. This constant drain voltage results in an overlap of voltage and current for lower output powers and degrades the drain efficiency. Fig. 3.11(b) shows the corresponding efficiency curves. The coding efficiency is high, as no energy is dissipated in the modulation sidebands, but the drain efficiency is degraded leading to a class B like overall performance.



Figure 3.11: Drain voltage and current for PWM and  $\Delta\Sigma$  operation for a parallel filter and resulting efficiency.

Using  $\Delta\Sigma$  envelope coding, as depicted in Fig. 3.11(c), shows essentially equal behavior, leading to the same efficiency performance in Fig. 3.11(b).

#### 3.6.2.2 Series Resonator

Instead of the parallel resonator its series counterpart, as depicted in Fig. 3.12, can be employed.



Figure 3.12: Block diagram of direct filter operation.

The series resonator causes the current to be constant within one period, while the drain voltage can be shaped to follow the desired modulation. This can be seen in Fig. 3.13(a), which shows the drain voltage and current for direct filter connection of a series resonator. During the first period the voltage swing is at its maximum enabling highly efficient operation. Due to the series resonator the current also has to flow during the second period. This results in a large overlap of current and voltage and leads to high power dissipation. Although the coding efficiency  $\eta_{\rm C}$  is theoretically 100 % the overall amplifier efficiency is degraded to the one of a class B amplifier, as can be seen in Fig. 3.13(b).



Figure 3.13: Drain voltage and current for PWM and  $\Delta\Sigma$  operation for the series filter and efficiency.

It shall be noted that the presented efficiency considerations are also valid for  $\Delta\Sigma$  operation, as depicted in Fig. 3.13(c).

#### 3.6.2.3 Series Resonator with Switch

When observing the drain voltage and current waveforms of the series filter connection in Fig. 3.13(a) it can be seen that during the second period no power is generated as there is no voltage swing. On the other hand the series resonator requires a continuous current. The resulting overlap causes only power dissipation and does not deliver RF power. As the drain voltage swing is basically zero during this period a DC decoupled switch can be placed in parallel to provide current conduction instead of using the PA itself. Fig. 3.14(a) depicts the corresponding schematic, which includes the DC decoupled switch  $S_1$  in parallel to the transistor. The DC decoupled switch is controlled by the modulator and closed during the second period, while during this time no gate signal is generated.

In Fig. 3.15(a) the resulting drain voltage and current for the DC decoupled switch circuit in Fig. 3.14(a) for PWM operation with 50 % duty cycle are given. During the ON period the PA is active and operates with maximum voltage swing and hence maximum efficiency. During the OFF period the PA is inactive and the switch  $S_1$  is closed. Its voltage and current waveforms are depicted in Fig. 3.15(c). A drawback of this arrangement is that the switch has to withstand high positive and negative voltages during the ON period and it has to conduct positive and negative current during the OFF period.

Nevertheless there is no additional power dissipation in the PA during the OFF period. When assuming an ideal switch theoretically constant efficiency can be achieved.



Figure 3.14: Direct filter connection of class B PA with switch.



Figure 3.15: Drain voltage and current for PWM and series filter with switch and efficiency.

Fig. 3.15(b) depicts the corresponding efficiency curve. It can be seen that the coding efficiency is high over the whole power range as there are only small residual losses in the modulation sidebands. Therefore, the drain and the system efficiency are almost the same. For very low output powers the efficiency degrades due to the influence of finite transition times between the two states leading to additional losses. In reality the switch will not be ideal and its ON resistance will significantly reduce the efficiency in the back off region.

An alternative to the DC decoupled switch is to place the switch in the DC supply, as depicted in Fig. 3.14(b), keeping it closed during the ON period supplying the DC current and opening it during the OFF period. By using a quarter wave line the open circuit is transformed into a short circuit at carrier frequency allowing the continuous current from the series resonator to flow. While with this arrangement the current through the switch is mainly positive, high voltage swings with an amplitude up to twice the supply voltage can occur. In both cases the requirements for the switch are very tough and can not be fulfilled with state of the art device technologies.

## 3.6.3 Multilevel

The system efficiency for single level PAs, even under highly efficient BB PWM operation, shows significant efficiency degradation in back off. A way to overcome this problem is the use of multilevel PA structures. Many different architectures for generating a multilevel signal exist and the most prominent ones are provided in Fig. 3.16.

The quarter wave (Fig. 3.16(a)) [171,180–182] and the transformer based combiner (Fig. 3.16(b)) [183–188] are circuits that offer multiple output power levels with high efficiency depending on their input control. At carrier frequency both combiners have the same properties and will therefore be treated as equivalent. A drawback of them is, that they require a (DC decoupled) switch to provide low impedance when the PA is off.

Depending on the ratio of power contribution of the single PAs different efficiency curves

can be achieved. When considering equal power for both PAs, it makes no difference which of the two PAs is active in the low power region. For the highest power both PAs are active. The resulting efficiency curve is depicted in trace '2 level' in Fig. 3.17(a) and is similar to a sawtooth function. But when using uneven power contribution it makes a difference



Figure 3.16: Block diagrams for multilevel combining.

which of the PAs is active for lower output powers. For very low output powers only the smallest PA contributes. When the output power exceeds its capabilities the PA with the higher output power is activated and for highest power both are active. This results in an additional efficiency peak, as depicted in trace '3 level' in Fig. 3.17(a). This means that by uneven dimensioning of the PAs an additional efficiency peak can be achieved.

Another possibility to generate multiple efficiency peaks is to switch the DC drain supply [10], as depicted in Fig. 3.16(c), in order to provide (discrete) supply modulation to follow the envelope function. This results in essentially the same sawtooth efficiency behavior. Fig. 3.16(d) provides the schematic of two PAs whose drains are connected in parallel RF wise, but each of them having a different DC supply [189, 190]. Using the smaller PA for lower and the larger one for higher powers shows the same efficiency trend.



Figure 3.17: Multi level efficiency for static, BB PWM and direct filter operation.

By applying the BB PWM coding using an isolated load, as covered in section 3.6.1, the sawtooth efficiency curves depicted in Fig. 3.17(a) can be improved in the high power region. The supply levels may be optimized in order to provide best average efficiency [191, 192]. In Fig. 3.18 example time domain waveforms for the quarter wave, respectively the transformer based combiner, for BB PWM operation for 75 % normalized load voltage, are depicted. The second PA is continuously switched on, as can be seen from its drain voltage

and current waveforms in Fig. 3.18(b). This results in a constant (CW) contribution to the load voltage depicted in Fig. 3.18(c). The bursts are provided by the first PA, which is switched on and off accordingly. For this output power range the relative contribution of the PWM is reduced compared to single level PWM coding, as a constant part (CW) is present in the load signal. This leads to reduced modulation sidebands for multilevel coding, as can be seen in Fig. 3.20(a). Exactly this reduced sideband contribution in the higher power regions leads to an improvement by BB PWM operation from the (static) sawtooth efficiency curves in Fig. 3.17(a) to the efficiency trends in Fig. 3.17(b). The theoretical derivation of this efficiency trend is provided in section 7.4.



Figure 3.18: Drain voltage and current for 2 level BB PWM encoding with 75% output voltage.

An exception to this is the Doherty amplifier (Fig. 3.16(e)), which shows already under static excitation the good efficiency depicted in Fig. 3.17(b). By combining the multilevel PAs with the concept of direct filter connection the efficiency can even be further increased, connecting the efficiency peaks of the Doherty PA directly [176,177,193]. This concept is not limited to the Doherty PA, but can be applied to any multilevel structure presented in Fig. 3.16. In the high power regions highly efficient operation, as depicted in Fig. 3.17(c), is possible, as at any time at least one PA is active, providing the required continuous load current. For the low power region, when only one PA is active, the efficiency is degraded to a class B trend due to overlap of voltage and current as described in section 3.6.2.2. Nevertheless highly efficient operation in the intermediate and high power region can be achieved.

#### 3.6.4 Multiphase

Instead of making use of multiple levels, the signal can be split into multiple phase components. This leads to the concept of multiphase BB PWM transmitters [194, 195]. The corresponding block diagram of a multiphase transmitter based on the quarter wave combiner is given in Fig. 3.19(a). Instead of using offset reference functions for the PWM generation of multilevel signals, the multiphase concept relies on (multi) phase shifted reference functions [195]. In Fig. 3.19(b) and Fig. 3.19(c) the encoding process for a 2 level transmitter with 25%, respectively 75%, output signal is depicted. It can be seen, that the bursts for the single branches  $V_1$  and  $V_2$  are generated at different phases. This is achieved by using the phase shifted reference functions  $REF_1$  and  $REF_2$  which are used to encode the amplitude information A. The (normalized) combined signal is composed of both pulses and shows an increased effective number of bursts. The idea is to reduce the actual modulation frequency for the single branches by making use of this increased effective modulation frequency by the multiphase coding. This can be seen in Fig. 3.20(b), which depicts the spectra of different multiphase encodings with the same modulation frequency. By increasing the number of encoding levels the first modulation sideband is moved further away from the carrier and reduced in magnitude. The magnitude reduction



Figure 3.19: Multiphase block diagram and signal generation for 2 levels.

is essentially the same as for the multilevel encoding in Fig. 3.20(a), therefore the resulting efficiency enhancement is also the same as in Fig. 3.17(b).



Figure 3.20: Multilevel and Multiphase spectra.

In Fig. 3.21 the corresponding time domain waveforms for the encoding periods of the example of Fig. 3.19(c) are depicted. It can be seen that the drain voltage for the single PA branches contains only one burst per encoding period, while the combined signal at the load contains two bursts. This is due to the fact that the single branch signals overlaps and adds up twice per modulation period.



Figure 3.21: Drain voltage and current for 2 phase BB PWM encoding with 75% output voltage.

This burst at the load, when both PAs are active at the same time, directly defines the drain currents of the single branches. Due to the quarter wave line the drain currents for the single transistors are related to load voltage and are therefore increased during the periods when both PAs are active. Although the effective modulation frequency seems to be increased by the multiphase encoding, the gate signal frequency components to generate the required drain current are increased as well. This can be seen in Fig. 3.21(c), which shows the spectra of the load and the gate signal for 2 phase encoding. It can be recognized that the drain voltage shows the desired doubling of the modulation frequency,

but that also the main modulation components of the gate signal are increased in the same way. The frequency multiplication effect also relies on cancellation of the corresponding frequency components by the single branch PAs, as the signal is multiplexed between them. To provide this cancellation the single branch PAs have to perfectly match in terms of amplitude and phase contribution.

Due to the required increased gate control frequency and the stringent matching requirements the multiphase concept has no significant advantages compared to the multilevel approach. It is also possible to provide an additional efficiency peak by uneven dimensioning of the multilevel encoding, which is not possible using multiple phases, which require exactly equal levels.

## 3.7 Energy Recovery

In this section various efficiency enhancement concepts involving energy recovery or energy reuse will be introduced. Starting with the DC energy recovery for LINC transmitters, also the aspects of DC energy recovery in the context of BB PWM operation will be covered. Instead of recovering the energy on DC level for the BB PWM operation, a concept capable of directly reusing the energy on RF level is presented. Finally, the idea of BPSK PWM coding and related DC energy recovery operation is discussed.

## 3.7.1 DC Energy Recovery LINC

The LINC concept presented in section 3.2 uses an isolated combiner to avoid any load modulation effects for the PAs. A result of this isolated combiner is that a significant amount of energy is dissipated in the isolation resistor. Instead of using a Wilkinson combiner, also a 180 ° hybrid, as depicted in Fig 3.22(a), can be used. The output power is controlled by modifying the relative phase  $\varphi$  of the single PAs. The inphase component is transfered to the load  $R_L$ , while the out of phase power is dissipated in the termination resistor  $R_R$ .



Figure 3.22: DC energy recovery outphasing.

A way to improve the resulting poor efficiency  $(\eta_{OUT})$  as seen in Fig. 3.22(b) is to rectify the power of the termination resistor  $R_R$  and feed it back to the DC supply [87, 196, 197]. Assuming the same rectifier as PA efficiency leads to the improved efficiency curve  $\eta_{OUT, DC}$  in Fig. 3.22(b). The theoretical derivation of the efficiency is provided in the Appendix 7.5.2. A drawback of this concept is the limited possible efficiency enhancement in back off, as the recovered energy is already generated with drain efficiency, leading to unavoidable losses. The signal at the rectifier also varies with the load voltage, requiring the rectifier now to be efficient over a wide power range. Thus, the design problem is rather shifted to the rectifier than solved.

## 3.7.2 DC Energy Recovery BB PWM

The concept of BB PWM operation with an isolated broadband load, as introduced in section 3.6.1, shows reduced efficiency due to the energy dissipated in the isolator. This energy can be reused by replacing the sideband termination resistor with a rectifier, as depicted in Fig. 3.23(a). Instead of dissipating the energy in the sideband termination resistor the signal is now rectified and fed back into the DC power supply [198, 199].



Figure 3.23: DC energy recovery BB PWM.

The derivation of the efficiency enhancement is provided in the Appendix 7.5.1 and the results are represented by the trace  $\eta_{\rm DC}$  in Fig. 3.23(b). Even when assuming a rectifier efficiency of 100 % the resulting efficiency enhancement is limited due to the fact that the recovered energy is generated with the drain efficiency and thus not all of the initial DC power used to generate the sideband components can be recovered. As for the LINC energy recovery concept the signal at the rectifier significantly varies with the output power, but also significantly within one BB PWM period. Thus, the rectifier has to be efficient for varying input power signals. Again, the design problem is rather shifted to the rectifier than solved.

#### 3.7.3 RF Energy Recovery

Instead of rectifying the signal at the termination resistor it can also be directly reused at RF level. Fig. 3.24(a) depicts the block diagram of a conventional BB PWM transmitter [38, 198, 200]. The power dissipated in the modulation sideband termination resistor  $R_{SB}$  leads to a significant reduction in efficiency. This power can be reused directly at RF level in order to provide constructive interference at the drain of the PA [S3]. In Fig. 3.24(b) the general block diagram of such a system is provided. To reflect the power in the desired phase the sideband termination resistor is replaced with a variable reflection  $S_R$ .

In order to derive the requirements of the variable reflector it is useful to investigate the time domain waveforms at the transistor and the sideband termination resistor. For the simulations presented in this section a Class B PA with a parallel resonator, placed in parallel to the DC decoupled transistor drain, is used [10]. It provides an open circuit for



Figure 3.24: Block diagrams of BB PWM transmitters with/without energy reuse.

the carrier and the modulation sidebands and short circuits all the carrier harmonics. The output filter is implemented as a high-Q parallel resonator.

In Fig. 3.25(a) the drain current and drain voltage of a Class B PA in burst mode operation with a duty cycle of 40 %, a supply voltage of 20 V and a carrier frequency of 1 GHz are plotted. It can be seen that the drain voltage is at maximum swing during the first period  $t_{ON}$  and zero during the second period  $t_{OFF}$ . The drain current is modulated accordingly and hence the amplifier is always efficient. Fig. 3.25(b) depicts the corresponding voltage across  $R_{SB}$ . It can be seen that the voltage at the sidebands termination resistor is the difference between the (DC decoupled) drain voltage and the average output voltage. In this case the average output voltage is 8 V due to the 20 V supply voltage and the 40 % duty cycle. This means that the voltage amplitude is 12 V during  $t_{ON}$  and -8V during  $t_{OFF}$ . This leads to a significant energy dissipation, resulting in poor overall efficiency for lower output powers which is identical to a Class B PA.



Figure 3.25: Duty cycle of 40% and conventional burst mode PA.

To reuse the energy it is required to reflect the signal from the sidebands termination resistor  $R_{SB}$  in the correct phase to the PA in order to have constructive interference [S3]. The required reflections will be derived in the following subsections based on partial energy recovery during  $t_{OFF}$  and  $t_{ON}$ , leading to full energy recovery in both periods. It shall be noted that for easier reading the mathematical derivation for all methods is provided in the Appendix 7.5.3.

#### 3.7.3.1 Reuse During OFF Period

One possibility to improve efficiency is to recover the energy during  $t_{OFF}$ . During the ON period the circulator port is terminated with 50  $\Omega$ , but during  $t_{OFF}$  the signal is reflected.

The correct phase to achieve constructive interference of the drain voltage depends on the used type of band pass filter. A parallel resonator, as considered in this simulations, provides a short circuit for the modulation sidebands and therefore an open circuit is required to reflect the signal with the correct phase for constructive interference. The series resonator on the other hand provides an open circuit for the modulation sidebands and thus a short circuit at the sideband resistor would be required.

The required reflections for the parallel resonator can be generated by simply using a switch in series to the sideband termination resistor, as depicted in Fig. 3.26(a). During the ON period the switch is closed, resulting in a perfect match, but during the OFF period the switch is open, providing the desired reflection. The resulting drain current and voltage for a duty cycle of 40 % are plotted in Fig. 3.26(b). It can be seen that also during  $t_{OFF}$  a drain voltage is present. The amplitude of the drain voltage corresponds to the output voltage, as the signal is fully reflected from the sideband termination and no energy is lost during this time. The drain voltage and current during  $t_{ON}$  are the same as for the conventional operation presented in Fig. 3.25(a), meaning the PA effectively sees the load resistor during this time.



Figure 3.26: RF energy recovery block diagram OFF period for a duty cycle of 40%.



(a) Efficiency versus normalized load voltage (b) Normalized load voltage versus duty cycle

Figure 3.27: Simulation results energy recovery during OFF period.

The presence of the drain voltage also during the OFF period causes a nonlinear dependence of the output voltage on the duty cycle  $d = t_{ON}/(t_{ON} + t_{OFF})$  and results in

$$V_{L^{OFF}} = \frac{2d}{1+d}.$$
(3.4)

The simulated output voltage dependence curve OFF is plotted in Fig. 3.27(b). It corresponds well to the theoretically calculated output voltage defined in (3.4). As can bee seen in Fig. 3.27(b), the resulting output voltage  $V_{L^{OFF}}$  is larger for low duty cycles compared to conventional operation  $V_{L^B}$ . Also the efficiency is higher than for conventional operation and can be calculated by

$$\eta_{OFF} = V_L \left(2 - V_L\right) \eta_D,\tag{3.5}$$

where  $V_L$  denotes the normalized load voltage. The efficiency enhancement is caused by the fact that no energy is dissipated in the sideband termination resistor  $R_{SB}$  during the OFF period anymore. In Fig. 3.27(a) the corresponding simulated efficiency curve  $\eta_{OFF}$ is plotted. For low output voltages the efficiency enhancement is limited, but for higher output voltages the efficiency can be significantly improved. It shall be noted that all simulations for efficiency and load voltage are performed with a modulation frequency of 5 MHz in order to reduce the influence of band limitation and aliasing effects [201], whereas for time domain plots a modulation frequency of 50 MHz is used for illustrative purposes.

#### 3.7.3.2 Reuse During ON Period

To recover the energy during  $t_{ON}$ , the energy from the sidebands termination resistor is reflected back to the PA during the first period, and terminated during  $t_{OFF}$ . As for the energy recovery during  $t_{OFF}$  the required reflection depends on the used type of band pass filter. For a parallel filter a short circuit is required during  $t_{ON}$ , whereas a series filter requires an open circuit. By using a switch in parallel to the sideband termination resistor, as depicted in Fig. 3.28(a), it is possible to generate a short circuit during the ON period and by leaving it open a perfect match is provided during the OFF period. The reflected signal causes constructive interference for the drain voltage and destructive interference for the drain current, as can be seen in Fig. 3.28(b). This means that the equivalent load resistance for the PA is increased, resulting in a lower drain current in comparison to conventional operation (Fig. 3.25(a)) during  $t_{ON}$ .



Figure 3.28: RF energy recovery block diagram ON period for a duty cycle of 40 %.

Additionally to the increased equivalent load resistance no energy is dissipated in the sideband's resistor during  $t_{ON}$ , which leads to a significant efficiency improvement. The resulting efficiency in dependence of the normalized load voltage  $V_L$  is given by

$$\eta_{ON} = \frac{1 + V_L}{2} \eta_D. \tag{3.6}$$

The simulated efficiency curve  $\eta_{ON}$  is depicted in Fig. 3.29(a). A significant efficiency improvement is possible, especially for low output voltages. The efficiency deviates from



Figure 3.29: Simulation results energy recovery during ON period.

the theoretical limit for very low output voltages due to the finite rise and fall times of the pulses caused by bandwidth constraints.

As well as for the energy recovery during  $t_{OFF}$  the output, respectively load voltage, does not depend linearly on the duty cycle d anymore and is given by

$$V_{L^{ON}} = \frac{d}{2-d}.$$
(3.7)

The corresponding control characteristic resulting from simulation can be seen in trace  $V_{L^{ON}}$  in Fig. 3.29(b). For low duty cycles the output voltage is smaller than for conventional operation and increases rapidly to its maximum value towards the maximum duty cycle.

#### 3.7.3.3 Full Reuse

Up to now only energy reuse either during the ON or OFF period was considered. Both methods have the drawback of nonlinear output voltage control characteristic and remaining loss in the sidebands termination resistor, which lowers efficiency. By combining both methods the energy in the sidebands can be fully reused and efficiency can be enhanced. This means that recovery takes place during the whole cycle. It is only required to reflect the signal from the sideband's resistor  $R_{SB}$  in the correct phase according to the period. By using a single switch, as shown in Fig. 3.30(a) the signal envelope at the sidebands termination resistor (Fig. 3.25(b)) can be rectified and fed back to the drain of the PA. Constructive interference for the drain voltage during  $t_{ON}$  and  $t_{OFF}$  occurs. The resulting drain voltage and current waveforms for a duty cycle of 40 % can be seen in Fig. 3.30(b).

The drain current during  $t_{ON}$  is lower than for conventional operation, meaning that the equivalent load resistance for the PA also increases for this type of operation. During  $t_{OFF}$  the load voltage is present at the drain of the PA. As no energy is dissipated at all in the sidebands termination resistor anymore, the efficiency  $\eta_{FULL}$  is simply the drain efficiency  $\eta_D$ . The corresponding simulated efficiency curve  $\eta_{FULL}$  is given in Fig. 3.31(a). For lower output voltages the efficiency starts to slightly increase, as the influence of the knee voltage is reduced for higher load resistances, and decreases for very low output voltages. This decrease is also present for the energy recovery during  $t_{ON}$ , due to the influence of the signal transitions on the drain efficiency.

In contrast to the nonlinear dependence of the output voltage on the duty cycle for partial energy recovery, the output voltage linearly depends on the duty cycle for full energy recovery. This can be seen in the simulation results in Fig. 3.31(b) in trace V<sub>L</sub>FULL. The slight



Figure 3.30: RF energy recovery block diagram full recovery for a duty cycle of 40 %.



(a) Efficiency versus normalized load voltage (b) Normalized load voltage versus duty cycle

Figure 3.31: Comparison of partial and full energy recovery.

increase in comparison to conventional operation is due to the increase of the equivalent load resistance presented to the PA during  $t_{ON}$ , resulting in reduced influence of the knee voltage, respectively ON resistance of the transistor, for the same drive level.

For a practical system implementation a very compact solution, minimizing all delays, is desired. A delay in the system will result in a different phase of the reflected signal and impacts constructive interference, which can be overcome by providing the correct reflection coefficients for the sideband termination. In case of a delay, there will also be a short overlap of both recovery phases. This can be solved either by introducing a short dead time between both recovery phases using the general reflection circuitry suggested in Fig. 3.32 or by performing partial energy recovery and selecting the most efficient mode according to the duty cycle.



Figure 3.32: General RF energy recovery block diagram.

## 3.7.4 DC BPSK Energy Recovery

The basic idea of BPSK operation is to drive the PA with a constant envelope signal to achieve high drain efficiency. During the first period the PA is driven in phase and during the second period 180° out of phase. The resulting driving signal is encoded as a Binary Phase Shift Keying (BPSK) signal. By changing the duty cycle from 0% to 50% (or 50% to 100%) the output voltage respectively power can be modified. The corresponding block diagram for isolated operation is given in Fig. 3.33(a). The resulting efficiency trend  $\eta_D = \eta_{max}V^2$  is similar to the one of a class A PA with higher peak efficiency, as the input power is constant and the output power increases with the duty cycle. The reflected (out of band) power from the BPF is dissipated in the isolation resistor resulting in a low efficient operation.



Figure 3.33: BPSK PWM with isolator.

The concept of DC BPSK energy recovery is, instead of dissipating this energy in the isolation resistor, to rectify and reuse it [202], [S4]. This can be done by removing the isolator as depicted in Fig. 3.34(a). To recover the energy a PA being capable of operating as PA and rectifier is required. Possible architectures are already discussed in section 2.7.



Figure 3.34: BPSK PWM with DC energy recovery.

The resulting efficiencies for different maximum drain efficiencies (under the assumption of equal and constant PA and rectifier efficiency) are plotted in Fig. 3.34(b). The theoretical derivation is provided in Appendix 7.5.4.

## 3.8 RF PWM

The concept of RF PWM is to operate the PA in a saturated mode and to modify the conduction angle to control the output power [171,203–207]. Some concepts combine the load modulation and the RF PWM approach to get best performance [169,170]. In this section the properties of an inverter based voltage mode class D amplifier will be discussed to explain the basic principles. The behavior of other amplifier classes under RF PWM excitation is covered in section 5.1. In Fig. 3.35(a) the circuit diagram of the inverter based voltage mode class D amplifier is given. It consists of an inverter pair that forces a rectangular drain voltage. The load is connected by a series resonator to only extract the fundamental frequency component and to provide an open circuit for the higher order harmonics.



Figure 3.35: Inverter based voltage mode class D.

The resulting spectral contributions versus different duty cycles of the RF PWM signal are depicted in Fig. 3.35(b). Due to the series resonator only the fundamental  $(1^{st})$  component is transferred to the load. Theoretically also a higher order harmonic could be selected as output signal, with the price of reduced maximum output power and higher sensibility to duty cycle variations.

The use of the series resonator enables in theory 100% efficiency over the whole power range. Unfortunately, the maximum efficiency in practice is limited by constant losses due to the parasitic drain source capacitance or drive signal generation for the gates. These losses are independent of the duty cycle and result in the following efficiency

$$\eta(V) = \frac{V^2 \eta_{MAX}}{\eta_{MAX}(V^2 - 1) + 1}$$
(3.8)

with respect the normalized load voltage V. The definition of the maximum efficiency  $\eta_{MAX}$  and the derivation of (3.8) are provided in Appendix 7.2.3.

In Fig. 3.35(c) efficiency plots for different maximum efficiencies are depicted. It can be seen that only for high maximum efficiency the back off efficiency is improved compared to class B performance.

## 3.9 Summary

In this chapter several efficiency enhancement methods have been introduced, covering well established methods like Doherty, Chireix, LINC or ET/EER. In the context of energy recovery based efficiency enhancement a new method for direct energy reuse on RF level of BB PWM operated transmitters has been presented.

Among the different options of BB PWM the concept of direct filter connection shows high potential for achieving good efficiency in back off. Therefore, a main chapter of this thesis is dedicated to the analysis and development of circuits for the use of BB PWM and direct filter connection.

But also the concept of RF PWM is of interest due to its purely digital control signal. The use of RF PWM control enables shifting the digital to analog boundary closer to the PA towards fully digital and reconfigurable transmitters. These aspects are presented and discussed in detail in chapter 5.

# 4. BB PWM

In this chapter the properties of BB PWM, with a special focus on direct filter connection, will be covered. This provides the foundation for the proposed circuits for direct filter connection. First the differences and advantages of digital PWM and band limited PWM modulation will be discussed. In the subsequent sections the properties of direct filter connection in terms of load modulation, filter requirements and modeling are covered. Additionally a study about the impact of the coupling on the input impedance of the cavity filter included in the output matching network of the PA circuits is presented. The main part of this chapter covers the different circuits developed for direct filter connection. Finally spectral and linearity aspects will be discussed and the properties of the circuits will be compared.

## 4.1 Modulator

In this section the aspects and properties of two different modulator implementations for the generation of BB PWM signals will be discussed. The PWM modulator can be either implemented in a purely digital way, where the PWM output signal is limited to 0 and 1, or by using a limited number of harmonics, leading to an analog PWM signal with intermediate output levels. Fig. 4.1(a) depicts the block diagram of the digital PWM modulator. The amplitude information a(t) is used to encode the PWM signal, which itself modulates the envelope of the carrier signal c(t). The phase information is introduced by modulating the carrier signal c(t) accordingly. In trace 'D PWM' in Fig. 4.1(b) a digital PWM signal at base band level with 40 % duty cycle is depicted. It can be seen that it is strictly bounded to 0 and 1.



(a) Digital BB PWM (b) Time domain 40 % PWM signal (c) Band limited BB PWM

Figure 4.1: Block diagram of BB PWM modulators.

In Fig. 4.2(a) the base band spectrum of the digital PWM modulator for a modulated signal is depicted. Due to the digital nature of the PWM signal an infinite number of higher order modulation harmonics exist. This can be well seen in the decaying spectral density towards higher frequencies. Upconverting this signal to RF level leads to the spectrum shown in Fig. 4.2(b). It can be seen that the tails of the PWM spectrum continue across DC and still have a significant contribution at the inband signal range around the carrier frequency. The aliasing around DC of the higher order harmonics in the digital PWM signal severely limits the achievable dynamic range of the RF signal [201].

Instead of using all modulation harmonics one can generate the BB PWM signal only with a limited number of harmonics to avoid the aliasing around DC [208,209]. The corresponding block diagram of such a band limited BB PWM modulator is given in Fig. 4.1(c). The



Figure 4.2: Aliasing problem digital BB PWM.

resulting output signal of the Band Limited (BL) modulator core for 10 PWM harmonics and a 40% duty cycle is plotted in trace 'BL PWM' in Fig. 4.1(b). It can be seen that due to the band limitation small ripples are present. Thus the output signal is not purely digital anymore and requires analog signals between 0 and 1 to be generated. Therefore, a D/A converter, as depicted in Fig. 4.1(c), is required after the PWM modulator core. This D/A converter is additional hardware that needs to be implemented compared to the purely digital approach. But on the other hand the band limited PWM signal can be generated with conventional IQ upconversion chains, meaning that simple integration is possible.

The main advantage of the BL PWM signal is its aliasing free output signal. The baseband signal, as depicted in Fig. 4.3(a), now shows only a limited number of harmonics, with no signal components above a certain maximum frequency. During the upconversion of this signal no aliasing occurs, due the band limited nature. The resulting aliasing free spectrum of the BL PWM modulator at RF level is provided in Fig. 4.3(b).



Figure 4.3: Aliasing free band limited BB PWM.

It can be seen that the spectrum exhibits a band pass behavior. This matches with the frequency characteristics of the input matching of PAs, which usually also exhibit band pass behavior. This means that for the digital modulator, the signal at the transistor's gate will exhibit a band pass nature, with the difference that the aliasing products due to the upconversion are still present in this case.

Thus by using the band limited modulator approach high signal quality and easy integration into existing modulator setups can be achieved. Therefore, for all following simulations it will be assumed, unless otherwise noted, that a band limited modulator is used for the generation of the BB PWM signals.

## 4.2 Direct Filter Connection

In this section the load modulation properties of a Band Pass Filter (BPF) will be discussed and a design methodology based on the equivalent load impedances will be presented. The requirements for the filter and the matching network for optimum performance are covered. Additionally the influence of the coupling on the reflections of cavity filters is analyzed, as they are of great importance to the matching network design. Finally the signal conditioning to achieve best performance of direct filter connection will be presented.

## 4.2.1 Filter Load Modulation

The use of direct filter connection results in load modulation during the BB PWM periods. As for well established concepts such as Doherty or Chireix, the load modulation trajectories of the direct filter connection are vital for the design of the PA. Therefore, the corresponding theory and the resulting design method are discussed in this chapter.

In Fig. 4.4(b) the circuit used for the derivation of theoretical load modulation is depicted. The series resonator is excited by a BB PWM modulated voltage source, which is used to model the saturated PAs. The existing theory [175] based on a series resonator is expanded by a Transmission Line (TL) to cover arbitrary orientation of the resonance locus. Fig. 4.4(a) depicts exemplary time domain traces for input voltage and current for full PWM operation. As the input voltage is different for the ON and the OFF period, while the current depends on both periods, two equivalent impedances  $Z_{ON}$  and  $Z_{OFF}$  for each period can be defined [S4].



Figure 4.4: Equivalent load impedances for full/partial load modulation.

In order to provide a general solution the PWM signal is allowed to be pulse width modulated between two arbitrary voltages  $V_1$  and  $V_2$ , leading to partial PWM modulation, as depicted in Fig. 4.4(c). Similarly to the full modulation, equivalent impedances for both periods ( $Z_1$  and  $Z_2$ ) can be defined.

The full derivation of the resulting load impedance trajectories for full and partial PWM modulation are provided in Appendix 7.7. Parameterizing the equations with the impedance at carrier frequency  $Z_{\rm C}$  and the symmetry impedance of the resonance locus  $Z_{\rm S}$  makes the calculation independent of the specific filter structure and generally applicable. The definition of  $Z_{\rm C}$  and  $Z_{\rm S}$  can be found in Fig. 4.5.

The resulting load impedances  $Z_{ON}$  and  $Z_{OFF}$  for full modulation depth and a delay  $\theta$  of 160° are plotted in Fig. 4.5(a). The delay  $\theta$  of 160° results in a detuning from the ideal series resonance locus by  $\Phi = 40^{\circ}$ . The impedance during the ON period  $Z_{ON}$  is very

high (close to the symmetry impedance  $Z_S$ ) for low duty cycles d and moves towards the carrier impedance  $Z_C$  for high duty cycles. As the voltage swing during the OFF period is zero for full PWM modulation the resulting equivalent impedance during that period  $Z_{OFF}$  is a short circuit, as can be seen in Fig. 4.5(a).



Figure 4.5: Example resonance locus with impedance definitions of  $Z_{\rm C}$  and  $Z_{\rm S}$  and resulting load impedances for full and partial modulation.

Exciting the same circuit with a partial PWM signal, varying between 0.5 and 1, results in different load modulation trajectories, as depicted in Fig. 4.5(b). The equivalent impedance  $Z_1$  varies along part of the modulation trajectory  $Z_{ON}$  for full modulation and shows the same trend with respect to the duty cycle d. As the voltage during the OFF period is nonzero for partial PWM modulation, the equivalent impedance  $Z_2$  during this period is modulated, as can be seen in Fig. 4.5(b).

For the theoretical derivation provided in Appendix 7.7 it is assumed that the band pass filter has infinite Q and that the impedance for all modulation sidebands is exactly the symmetry impedance  $Z_S$ . It was found that the theoretical equations are still a good approximation for finite Q resonators [S4] even if the impedances for the modulation sidebands  $Z_{SB}$ , as depicted in Fig. 4.5, deviate from the symmetry impedance  $Z_S$ . The only requirement is that the most important modulation sidebands (approx. 10) still see a reasonably high reactive impedance.

With the knowledge of the modulation trajectories the matching network can be designed in such a way that the load modulation trajectories follow efficient regions of the PA. It is shown that the orientation of the resonance locus, respectively the symmetry impedance  $Z_S$ , has a high impact on the modulation trajectories and is therefore an important design parameter. Beside for the design of the matching network the equations can be used to optimize signal generation for high efficient operation when operating the circuit not at the design frequency.

## 4.2.2 Filter and Matching Network Requirements

For the use of direct filter connection it is desirable to provide a high impedance over the widest range possible for the modulation sidebands above and below the carrier frequency. Also the Q-factor of the band pass filter and the related losses are an important parameter.

Based on the circuits presented in Fig. 4.6 the challenges of designing the matching network will be discussed. The ideal circuit for direct filter connection would be a series resonator with the correct load impedance directly connected to the transistor's drain, as depicted in Fig. 4.6(a). In order to provide the DC biasing of the drain a quarter wave length line is added.

The corresponding resonance locus is provided in Fig. 4.7(a). It can be seen that the resonance locus is ideally centered around the open circuit due to the series resonator. It



Figure 4.6: Different matching networks including high-Q BPF for direct filter connection.

can be observed that the impedances at the modulation sidebands  $Z_{SB}$ , which are marked with 'x', deviate from the ideal open circuit. This is caused by the quarter wave length line used for the DC drain biasing. The phase behavior of the reflection coefficient can be seen in trace ' $\angle S_{11}$  (a)' in Fig. 4.7(b). The impedance, respectively reactance, remains above the carrier impedance for angles between -90° and 90°, which corresponds to 0.5 f<sub>c</sub> and 1.5 f<sub>c</sub> in this case.

Considering a parallel resonator, as depicted in Fig. 4.6(b), an additional quarter wave is required in order to transform the resonance locus into the trajectory of a series resonator, as depicted in Fig. 4.7(a). The quarter wave line causes a delay between the transistor and the load, reducing the frequency range with high impedance, as can be seen from the phase of trace  $2 S_{11}$  (b) in Fig. 4.7(b).



Figure 4.7: Impedance behavior of the circuits in Fig. 4.6.

Increasing the delay, as in the circuit in Fig. 4.6(c), leads to a further reduction of the high impedance region. This can be observed in the corresponding trace  $2 \leq S_{11}$  (c) in Fig. 4.7(b).

The delay on the one hand reduces the region of high impedance around the carrier and thus should be as short as possible. But on the other hand the delay can be used in order to provide fundamental impedance matching. The delay is helpful in providing the correct termination for the higher order harmonics. In practice always a certain shift of the resonance locus due to the coupling itself and a delay due to the connection to the coupling will be present, such that the ideal case in Fig. 4.6(a) is practically not realizable. Also the implementation of the circuit in Fig. 4.6(b) can hardly be implemented with reasonable coupling and therefore the circuit in Fig. 4.6(c), which shows high impedance between  $0.8 \, f_c$  and  $1.2 \, f_c$ , will be used for the definition of the bandwidth requirements.

The maximum PWM frequency  $f_{PWM}$  has to be set in such a way that the most important

sideband harmonics (the first 5-10) fall into the region of high impedance. Fig. 4.8(a) depicts such an example. In order to provide sufficient spectral spacing between the inband signal and the first modulation sidebands the ratio of the PWM frequency  $f_{PWM}$  to the signal bandwidth BW has to be at least a factor of 5; This is required to allow for the filter's transition from pass to stop band and to provide some margin for Digital Predistortion (DPD) purposes.

Considering these two requirements results in a factor of 50 between the maximum signal bandwidth and the maximum bandwidth of high impedance. Taking some margin into account, as the matching network with the band pass filter cannot be used efficiently up to its 3 dB points, leads to a factor of 100. The circuit of Fig. 4.6(c) with its 20% relative bandwidth of high impedance leads to a maximum signal bandwidth of 0.2% with respect to the carrier frequency. This signal bandwidth, taking the margin to the 3 dB points of the filter into account, leads to the requirement of a loaded Q-factor  $Q_L$  in the range of 250-500 in order to fulfill the bandwidth requirements.



Figure 4.8: Unloaded/loaded Q-factor requirements of band pass filter.

A real band pass filter will always exhibit some losses, usually characterized by the unloaded  $Q_0$ , as depicted in Fig. 4.8(b). The insertion loss of the band pass filter with respect to the loaded/unloaded Q-factor is given by

$$\eta = 1 - \frac{Q_L}{Q_0}.\tag{4.1}$$

Its derivation is provided in Appendix 7.8. Fig. 4.8(c) depicts the corresponding efficiency curve of (4.1). It can be seen that for low insertion loss it is required to have a ratio of the unloaded to the loaded Q-factor  $Q_0/Q_L$  of around 10 or even higher. This means that for the design of direct filter connection circuits  $Q_0$  values in the mid 1000 range are required in order to fulfill spectral requirements and to obtain good efficiency.

Some filter implementations, such as lumped element or microstrip, have much lower unloaded  $Q_0$ -factors and can therefore not be used. Other filter structures such as dielectric resonators or cavity filters can provide such high  $Q_0$ -factors. Cavity filters have the advantage of easy to manufacturing and tuning and where therefore selected for the design of the matching networks for the direct filter connection presented in this chapter. Their properties will be analyzed and discussed in the next section.

## 4.2.3 Coupling Effects in Cavity Resonators

The location of the resonance locus of the filter and the delay due to the coupling can have a significant impact on the frequency performance of the matching network, as discussed in section 4.2.2. Based on the circular cavity resonator used for the matching network design the influence of the coupling on the filter's reflections will be discussed. The measurement and the influence of the Q-factor in the presence of coupling is very well described in [210–214], but very little attention is paid to the phase of the reflections of the filters. Therefore, a special focus on the relation of the inductive and capacitive coupling to the phase, respectively location of the resonance locus, will be paid in this section [S5].

In Fig. 4.9(a) the mechanical dimensions of the cavity resonator with a diameter of 85.6 mm and a height of 41 mm, are depicted. Capacitive coupling at the top and the bottom and inductive loop coupling at the cavity's walls was considered. As feedthrough into the cavity an extension of an SMA connector, which yields a 50  $\Omega$  coaxial transmission line, was used.



Figure 4.9: Mechanical dimensions of the cavity and coupling equivalent circuit elements.

The measured and simulated data was deembedded to the internal cavity walls, as shown in Fig. 4.9(b) and Fig. 4.9(c) respectively. The results presented in section 4.2.3.1 and section 4.2.3.2 are from finite element simulations performed with HFSS. Additionally the results from simulation have been confirmed by measurements on a coarse grid.

## 4.2.3.1 Capacitive Coupling

The influence of the coupling on the resonator's properties was studied using the equivalent circuit model in Fig. 4.10 [210]. In addition to the standard model, an ideal transformer is added [S5] to account for the different coupling polarities that can occur, depending on the orientation and location of the coupling, as well as the mode to which it is coupled to.



Figure 4.10: Equivalent circuit for capacitive coupling.

For analysis and representation purposes the circuit parameters of the parallel shunt circuit  $L_P$  and  $C_P$  are expressed in terms of resonance frequency  $f_0$ , internal Q-factor  $Q_0$  and

parallel resistance R<sub>P</sub>.

$$L_{P} = \frac{R_{P}}{2\pi f_{0} Q_{0}}; \ C_{P} = \frac{Q_{0}}{2\pi f_{0} R_{P}}$$
 (4.2)

For the capacitive coupling the center conductor of an SMA connector was extended to reach into the cavity, as depicted in Fig. 4.9(b). The coupling length l was swept from 0 to 12 mm during simulation. Additionally 4 different conductivities ( $\sigma = \{1.1, 7, 15, 58\}$ MS/m) resulting in internal Q-factors ranging from  $Q_0=2223$  to  $Q_0=16118$  were considered. The corresponding results for the transmission coefficient  $S_{21}$  are depicted in Fig. 4.11. For illustrative purposes only  $S_{21}$  versus frequency of selected values is plotted for the trace with the highest insertion loss ( $Q_0 = 2223$ ) and for all the others only the maximum value  $\max(S_{21})$  for each length l and  $Q_0$  are plotted. For stronger capacitive coupling (longer l) the insertion loss and the external resonance frequency  $f_{EXT}$ , which defines the maximum of the transmission coefficient  $\max(S_{21})$ , decrease. At the same time the coupling bandwidth increases. For higher internal  $Q_0$  the insertion loss for the same coupling length is reduced, while the frequency shift and the coupling bandwidth remain the same.



Figure 4.11: Insertion loss  $(S_{21})$  and minimum insertion loss  $(max(S_{21}))$  for inductive and capacitive coupling for coupling length, respectively diameter and different Q-factors versus frequency.

The reflections' coefficients of the filter versus frequency and coupling length l are plotted in Fig. 4.12(a). The inner cavity walls as shown in Fig. 4.9 have been selected as reference plane. Only the full data of selected values for the lowest Q-factor are plotted and for higher Q-factors the minimum reflections  $\min(S_{11}(f;l))$  for different coupling lengths are depicted. In the smith chart it can be seen that the resonance locus is slightly shifted towards the capacitive region. This is caused by the parasitic capacitance of the coupling element. For higher  $Q_0$  the losses are reduced and thus the reflections for the same coupling length are lower. This means that for the same mismatch the length of the coupling and also the parasitic capacitance for higher  $Q_0$  is lower and the shift towards the capacitive region is smaller. For strong coupling the traces converge towards the same point (perfect match).

To better understand the impact of the coupling, the parameters of the equivalent circuit model of Fig. 4.10 were fitted to the simulation data for each sweep value (length and conductivity). This was done by fitting the S-Parameters using the MATLABs' fminsearch algorithm. During the fitting process the initial parameters for  $f_0$ ,  $Q_0$ ,  $R_P$  and  $C_S$  were estimated and their accuracy was improved using the optimizer. The transformer's ratio  $N_1$ was set to -1 to account for the coupling polarity. The optimizer output was crosschecked by means of a circuit simulator. As the S-Parameters of the equivalent circuit model are in



Figure 4.12: Reflection  $S_{11}$  versus frequency and minimum reflection  $\min(S_{11})$  for different coupling lengths, respectively diameter and Q-factors for  $Z_0 = 50\Omega$ .

good agreement with the data presented in Fig. 4.11 and Fig. 4.12(a) they are considered to be equivalent. Hence only the resulting values of the cavity fit will be discussed.

The values for the parasitic capacitance of the coupling element versus length l are plotted in trace  $C_S$  in Fig. 4.14(a). It can be seen that the capacitance increases roughly linearly with some offset for zero coupling length. In addition the fringe capacitance  $C_{S^{FR}}$  of the coupling element above an infinite ground plane is evaluated to show that it is equal to the series capacitance  $C_S$  of the full model fit. As both values are in good agreement, the values for the series coupling element may be determined by the fringe field only to reduce the degrees of freedom and thus complexity for the fitting of the equivalent circuit. The coupling capacitance is independent of the  $Q_0$  of the cavity, as it is only determined by the geometric dimensions and not by the conductivity of the cavity.

The dependence of the internal  $f_{0^{CAP}}$  and external resonance frequency  $f_{EXT^{CAP}}$  for capacitive coupling is given in Fig. 4.14(b). It can be seen that also the internal resonance frequency  $f_{0^{CAP}}$  of the fitted model decreases with coupling length l, which causes a similar effect as inserting a tuning screw. The external resonance frequency  $f_{EXT^{CAP}}$  is even lower than  $f_{0^{CAP}}$ , as for capacitive coupling the parallel shunt resonator has to be inductive, and thus below the resonance frequency.

The Q-factor is independent of the coupling and also remains constant over different coupling lengths for the fitted model. The parallel resistance  $R_P$  increases with stronger coupling due to the resulting lower losses.

## 4.2.3.2 Inductive Coupling

Round coupling loops located at the cavity side walls, as depicted in Fig. 4.9, were considered for the inductive coupling. For the coupling loop a flat conductor with a thickness t of 0.3 mm and a width of 2 mm was considered. The diameter of the coupling was swept from 1 mm to 12 mm and the used conductivities, and therefore internal Q-factors, were the same as for the capacitive coupling in section 4.2.3.1. It shall be noted that also other shapes have been considered, but the general trend is approximately the same. Similar as for the capacitive coupling an equivalent circuit model, as given in Fig. 4.13, was used to analyze the impact of the inductive coupling.

In Fig. 4.11 the transmission coefficient  $S_{21}$  is plotted. As for the capacitive coupling only selected traces for the highest insertion loss ( $Q_0 = 2223$ ) are plotted and for lower insertion loss the maximum max( $S_{21}(f;d)$ ) for each parameter sweep is depicted. It can be seen that with increasing coupling the external resonance frequency, which is at the maximum transmission max( $S_{21}$ ), and also the coupling bandwidth increase. But the



Figure 4.13: Equivalent circuit for inductive coupling.

frequency shift versus coupling strength is not as high as for capacitive coupling and thus a lower insertion loss can be achieved with a smaller frequency shift.

The impact of inductive coupling on the reflection is much more severe compared to capacitive coupling. This can be seen in Fig. 4.12(b), which shows the reflection coefficient  $S_{11}$  of the inductive coupling in the same manner as for the capacitive coupling. For small diameter d the series inductance and the coupling are small. But with increasing coupling the resonance locus moves from the inductive region towards the capacitive region of the smith chart. This is due to the fact that the coupling element itself imposes a small delay and that for large diameters d the frequency of interest is already in the vicinity of the self resonance frequency of the inductor. For strong inductive coupling  $S_{11}$  is in the same region as for capacitive coupling and can even be shifted further towards the short circuit.

For the coupling element of the equivalent circuit model in Fig. 4.13, an inductor in series with a short transmission line was used. The transmission line was required to model the extended feed of the center conductor and to take into account that the 2 mm wide coupling loop acts in the vicinity of the cavity walls like a short transmission line. For simplicity the impedance of the transmission line was assumed to be 50  $\Omega$ . The transformers ratio N was set to 1 and the delay  $\theta$  of the transmission line was determined to be in the range of 10 ps to 30 ps (strong coupling) by evaluating the parameters of the coupling element only above an infinite ground plane. Beside the delay only the resulting inductance of the coupling element L<sub>SFR</sub>, as depicted in Fig. 4.14(a), was determined. Also for the inductive coupling the parameters for the coupling with (L<sub>S</sub>) and without (L<sub>SFR</sub>) resonator are in good agreement, showing that the parameters of the coupling are independent of the cavity. The series inductance L<sub>S</sub> increases with the diameter d. For very large diameters the inductance decreases again, as the frequency region of interest is already in the vicinity, respectively above, the self resonance frequency of the coupling element.



Figure 4.14: Fitted values for inductive and capacitive coupling versus coupling length, respectively diameter.

The impact of the inductive coupling on the internal resonance frequency  $f_{0^{IND}}$  and external resonance frequency  $f_{EXT}^{IND}$  can be seen in Fig. 4.14(b). The internal resonance frequency  $f_{0^{IND}}$  of the model increases only slightly with the coupling. In contrast to capacitive coupling, the external resonance frequency  $f_{EXT}^{IND}$  for inductive coupling is higher than the internal resonance frequency  $f_{0^{IND}}$  due to the series inductance. The internal Q-factor and the parallel resistance  $R_P$  behave the same as for capacitive coupling.

Other important aspects are the (external) coupling bandwidth and the insertion loss of the resonator. The relation between them is given in Fig. 4.15. It was observed that the insertion loss for the same coupling bandwidth is the same for capacitive and inductive coupling and therefore only one curve is shown. It can be seen that for higher  $Q_0$  the insertion loss for the same coupling bandwidth is significantly reduced.



Figure 4.15: Insertion loss  $(S_{21})$  and coupling length and diameter versus coupling bandwidth.

Additionally the required coupling length, respectively diameter, for a certain bandwidth is depicted. As the self resonance of the inductive coupling element causes the coupling to become ineffective for very large diameters, it is not possible to couple as strong as with capacitive coupling.

#### 4.2.3.3 Broadband Modeling

For the design of power amplifiers, not only the impedance at the fundamental, but also at higher order harmonics is important. Therefore, the equivalent circuit model used for inductive/capacitive coupling is expanded to the circuit in Fig. 4.16 to take the higher order resonances, respectively modes, into account. The higher order modes of the cavity resonator can be considered by an extension of the shunt RLC circuit to N-levels [210]. To consider the different polarity of the coupling, due to the location/orientation and the used mode, a transformer for each resonator is added [S5]. The coupling elements are considered in form of a series impedance  $Z_S$  with an optional transmission line TL.

To verify the simulation results the cavities shown in Fig. 4.17 have been built and measured. The aluminum cavity shown in Fig. 4.17(a) showed a rather low Q-factor and therefore the copper cavity depicted Fig. 4.17(b) was used for further analysis and the design of the matching network.

To validate the proposed broadband model of Fig. 4.16 it is fitted to the measured data for capacitive coupling to the manufactured cavity depicted in Fig. 4.17(b). The first four resonances of the TE mode with significant coupling are taken into account and the influence of the remaining higher order resonances was considered by using a fifth resonator. To achieve good convergence of the optimizer, the parameters of each peak were determined separately and acted as initial parameters for the full model fit.



Figure 4.16: Equivalent circuit model for resonators considering the coupling and higher order modes with different coupling polarity.



(a) Aluminum

(b) Copper

Figure 4.17: Manufactured aluminum and copper cavity.

In Tab. 4.1 the resulting parameters are given. For the coupling element a series capacitor  $C_S$  with a resulting value of 0.214 pF was used. The low Q-factor for the second resonance is attributed to the fact that multiple modes are excited. The parallel resistance  $R_P$  determines the shunt reactances according to (4.2) and thus defines the 'characteristic impedance' of the resonance. In Fig. 4.18 it can be seen that both, magnitude and phase of  $S_{21}$  are in good agreement. The magnitude of  $S_{11}$  also fits well, but the phase slightly deviates as the real coupling element has a frequency dependence of  $C_S$  due to its physical dimensions.

## 4.2.3.4 Summary

For strong coupling the reflections of capacitive and inductive coupling are in the same region caused by the high inductance of the coupling loop and the operation in the vicinity of the self resonance. Due to that the phase of the reflection for capacitive coupling is less sensitive to the coupling strength than for inductive coupling, as can be seen well in Fig. 4.12. Therefore, capacitive coupling is better suited for the design of the matching networks for direct filter connection. The capacitive coupling has the advantage that it implicitly provides DC decoupling of the load mitigating the demand for a series capacitor

f <sub>0</sub>	$2.6799\mathrm{GHz}$	$4.5817\mathrm{GHz}$	$6.166\mathrm{GHz}$	$7.7736\mathrm{GHz}$	$9.2285\mathrm{GHz}$
$Q_0$	4728.8	120.45	1408.6	933.4	31.6
$R_{\rm P}$	$14148\Omega$	$161.4\Omega$	$5319.6\Omega$	$17191\Omega$	$747.9\Omega$
Ν	-1	1	-1	1	-1

Table 4.1:	Equivalent	$\operatorname{circuit}$	parameters.
------------	------------	--------------------------	-------------



Figure 4.18: Measurement and model fit data (markers) of reflection coefficient  $S_{11}$  and insertion loss  $S_{21}$  for the first 4 TE-mode resonances.

serving this purpose.

The PA designs involving direct filter connection presented in the section 4.3 will be based on the measured data of the cavity filter shown in Fig. 4.17(b). With its external -3 dB bandwidth of around 11 MHz and its insertion loss of -0.53 dB in an 50  $\Omega$  environment it fulfills the requirements derived in section 4.2.2.

#### 4.2.4 Direct Filter Signal Conditioning

For real circuit implementations the filter is not perfect in terms of filter bandwidth and resonance locus orientation. This results in impedances for the modulation sidebands that deviate from the ideal open circuit and eventually lead to some residual modulation of the filter current. In order to cope with that effect and to operate the PA at its optimum operational points, also within one BB PWM period, it is required to modify the applied gate signal. Fig. 4.19(a) shows a circuit for direct filter connection including the gate signal conditioning by the band pass filter. It shall be noted that the position of the band pass filter can be moved into the digital (complex) base band, as indicated in Fig. 4.19(b)



Figure 4.19: Gate signal conditioning direct filter connection.

In order to generate the required gate current in terms of phase and amplitude the filter's transfer function  $S_{21}$  can be calculated by

$$S_{21}(\omega) = \frac{Z_L(\omega_C)}{Z_L(\omega)},\tag{4.3}$$

where  $Z_L(\omega_C)$  denotes the input impedance at carrier frequency and  $Z_L(\omega)$  is the input impedance with respect to frequency. In case of an open circuit for the modulation sidebands the resulting signal is constant with the amplitude corresponding to the input amplitude a(t). But if the input impedance at the modulation sidebands is finite there is a small residual remainder from the PWM modulation, which is required to provide the optimum load current throughout one BB PWM period.

Fig. 4.20(a) depicts an example for the input impedance of the BPF, where 'x' denotes the impedances at the modulation sideband frequencies.



Figure 4.20: Example impedance of direct filter connection and associated signal generation.

The corresponding transfer function  $S_{21}$  of the filter, according to (4.3), is provided in Fig. 4.20(b). While for an ideal series resonator the signal at the modulation sidebands would be completely attenuated it can be seen that in this example the PWM signal is still present to some extend and corrected in terms of amplitude and phase in order to generate the desired drain voltage.

Fig. 4.20(c) shows the drain current and voltage time domain waveforms for the circuit in Fig. 4.19(a) considering the predistortion shown in Fig. 4.20(b). It can be seen that the drain current shows some residual amplitude and phase modulation. By correcting the gate signal accordingly the PA is always operated under optimum compression, leading to best efficiency performance.

If the impedance at the modulation sidebands is sufficiently high, meaning that only very small residual modulation of the drain current is present, the signal conditioning might not be required and the gate can be excited by a CW signal.

# 4.3 BB PWM PA Circuits

This section is dedicated to the development of PA circuits for direct filter connection. To enable the direct filter connection the PA circuits have to provide a multilevel signal and to be capable of dealing with the constant current of the filter. Four different integrated PA circuits designed for direct filter connection under multilevel excitation are presented. Two of them are based on parallel concepts and the other two rely on series concepts in order to provide multilevel PA structures. The properties of a DC BPSK energy recovery circuit, requiring a single transistor only, are evaluated. Additionally the performance of a Doherty amplifier under BB PWM operation with an isolated broadband load is investigated.

## 4.3.1 Common Source Parallel PA

The main requirement for efficient operation of the direct filter connection is the use of a multilevel excitation. In addition the PA has to be capable of dealing with the resulting

constant drain current efficiently for all output levels. One structure that is capable of doing so is the parallel PA, which is based on two PAs in parallel, each supplied with a different supply voltage and only one operated at a time. The basic operational principle was introduced in section 3.6.3.

Fig. 4.21 depicts such a design of a parallel PA on a common source basis [S6, S7]. The design is split into 3 main parts. The Input Matching Network (IMN) and the Output Matching Network (OMN) are implemented on Printed Circuit Board (PCB) level using a Rogers 4350B substrate to enable the use of high Q passives and tuning after manufacturing. To provide best performance the connection of the two transistors' drains has to be very close together. The low power transistor requires a series diode for protection and therefore the core of the circuit is designed in an integrated GaN process [215]. The used transistors have a gate periphery of  $4\times100\,\mu$ m with a maximum current of around 400 mA, their DC IV curves can be found in Appendix 7.9. It shall be noted that the same transistor size is used for all Monolithic Microwave Integrated Circuit (MMIC) designs presented in this chapter. For the direct filter connection a codesign of the OMN and the cavity BPF was done in order to provide the desired resonance locus and to achieve best performance.

The basic principle of the circuit relies on biasing the two transistors  $T_1$  and  $T_2$  with different DC supply voltages, where  $V_{DC1} > V_{DC2}$ . To provide the DC decoupling the drain of the two transistors is connected using the series capacitor  $C_{DC}$ , whose parasitic self resonance frequency is designed to be at the carrier frequency in order to provide the best connection. For efficient operation only one of the two transistors is active at a time, while each exhibits its maximum voltage swing. This results in two different (non zero) envelope voltage levels, for which high drain efficiency can be achieved. The input signal is encoded and predistorted in order to shape the drain voltage to these two voltage levels, and then multiplexed between the two transistors according to the duty cycle of the PWM signal.

Fig. 4.22 depicts exemplary time domain waveforms of the drain voltages and currents for both transistors and a duty cycle of 50 %. For illustrative purposes a high modulation frequency of 132.5 MHz was selected, while the real circuit is operated at 20.7 MHz ( $f_c/128$ ). Transistor T<sub>1</sub> is biased with 30 V and transistor T<sub>2</sub> with 15 V. In combination with the 50 % duty cycle the resulting load voltage is 22.5 V, which corresponds to a normalized load voltage of 0.75. The drain voltage swing toggles between the two DC supply voltages and the drain current is multiplexed between both transistors according to the duty cycle. It can be seen that the drain voltage has a maximum swing for each transistor during current conduction. Therefore, the drain efficiency is at its maximum. In combination with the high coding efficiency, due to the open circuit for the PWM sidebands, high system efficiency can be achieved.

When the high power transistor  $T_1$  is active  $(t_{ON})$  the drain voltage of the lower power transistor  $T_2$  drops below zero. This would lead to conduction of the parasitic gate-drain diode of the GaN transistor. To prevent negative voltages at transistor  $T_2$ 's drain a series diode  $D_1$  is added. In order to ensure a positive voltage at the transistor's drain also during signal transitions a small anti parallel diode  $D_2$  is required as well.

For the design of the matching networks of a PA the Load Pull (LP) contours are of importance. Fig. 4.23(a) and Fig. 4.23(b) depict the load pull contours at the reference plane  $Z_L$  for low and high power operation respectively. It can be seen that the efficiency maxima are different in terms of resistance due to the two different supply voltages. Also in terms of reactance the efficiency contours deviate. This is due to the fact that for the high power operation the series combination of the parasitic capacitances of  $D_1$  and  $T_2$  load the circuit, while for low power operation the parasitic capacitance of  $D_1$  is effectively short



Figure 4.21: Schematic of the common source PA with direct filter connection.

circuited increasing the capacitive loading as only T<sub>2</sub>'s parasitic capacitance is present. For the design the focus was set on the efficiency of the low power operation, as the efficiency at low powers is more important for the average efficiency than the maximum efficiency of the circuit. Another important parameter for direct filter connection, which was also considered during the matching network design is the orientation of the resonance locus. By properly designing the matching network in Fig. 4.21(a), including the orientation of the resonance locus, the load modulation trajectories  $Z_{ON}$  and  $Z_{OFF}$  shown in Fig. 4.23(c) follow efficient regions in the smith chart. Additionally the impedances at the modulation sidebands  $Z_{SIDE}$  for a modulation frequency of 20.7 MHz are indicated.

In Fig. 4.24 the drain voltage waveforms of the circuit in Fig. 4.21 for low power operation are provided. It can be seen that the low power PA is operated with maximum voltage swing leading to maximum efficiency. The drain voltage  $V_{DD2}$  before and  $V_{D2}$  after the diode are almost identical as the diode is conductive. The voltage at drain of the high power transistor in Fig. 4.24(b) is simply a DC shifted replica of T<sub>2</sub>'s drain voltage. The drain current of the high power transistor is essentially zero due to the class C gate biasing, leading to almost no power consumption of T<sub>1</sub>. As the drain of T<sub>1</sub> is actively excited there is some feedback to the gate due to the parasitic gate-drain capacitance. This feedback can lead to conduction of the transistor resulting in power consumption and even instability. Therefore, the class C biasing was required not only for the low DC drain current, but also to provide some margin such that the feedback effect doesn't cause any drain current conduction in the OFF state. To ensure stability a resistance network was added to the transistor's gates, which also improves the feedback behavior. These resistance networks



Figure 4.22: Drain voltages and currents for the Class-B parallel PA circuit for a PWM frequency of 132.5 MHz, a carrier frequency of 2.65 GHz and a duty cycle of 50 %.



Figure 4.23: Load pull efficiency contours, matching network and load modulation trajectories for a carrier frequency of 2.6515 GHz.

consist of a combination of series and shunt elements and are located between the probe pads and the transistor, as can be seen in the layout in Fig. 4.21(b).

During high power operation the transistor  $T_1$  is excited, resulting in maximum voltage swing, as depicted in Fig. 4.25(b). The low power transistor is inactive during this time. It can be seen that the series diode prevents negative voltage on the transistor's drain such that  $V_{D2}$ , given in Fig. 4.25(a), always remains positive. Nevertheless the drain voltage  $V_{D2}$  shows still some voltage swing, as the parasitic capacitances of the diode and the transistor form a capacitive voltage divider.

The frequency dependence of the gain, output power and efficiency is plotted in Fig. 4.26. Although the drain efficiency for low power operation, as shown in Fig. 4.26(a) is quite high, the PAE is severely reduced by the low gain of less than 10 dB. The low gain is a result of the class C biasing and the reduced supply voltage for the low power operation. The diode also increases the series resistance of the transistor and thus contributes to the efficiency degradation. Due to the band pass filter required for direct filter connection output power and efficiency degrade quite quickly when deviating from the design frequency of 2.6515 GHz.

For high power operation in Fig. 4.26(b) it can be seen that higher values in terms of gain and efficiency are achieved due to the higher supply voltage. When comparing the efficiency of both operational conditions, as given in Fig. 4.26(c), it can be seen that their maxima are different. This is caused by the different capacitive loading for both operational conditions.


Figure 4.24: Drain current of both PAs for low power mode.



Figure 4.25: Drain current of both PAs for high power mode.

Fig. 4.27 depicts the efficiency curves under static and BB PWM excitation for the design frequency of 2.6515 GHz. From the static efficiency curve in Fig. 4.27(a) it can be seen that the desired efficient multilevel operation is achieved. Two efficiency peaks at 0.65 W (8 V load voltage amplitude) and 2.5 W (15.8 V load voltage amplitude) output power lead to a sawtooth like efficiency curve for CW excitation. By using BB PWM in combination with direct filter connection it is possible to connect the two maximum drain efficiency points leading to the efficiency curve provided in Fig. 4.27(b). It can be seen that for the high power range above the output power of the low power transistor highly efficient operation can be achieved. Due to the rather low gain the corresponding PAE is reduced. Accounting for the insertion loss of 0.54 dB of the band pass filter, the drain efficiency and PAE are further reduced, but still show the highly efficient trend of direct filter connection.

It can be concluded that by combining the common source parallel PA presented in this section with direct filter connection and BB PWM highly efficient operation in the high power region can be achieved.

## 4.3.2 Common Drain Parallel PA

The operational principle of the common drain parallel PA is essentially the same as for the common source parallel PA presented in the previous section. The main difference is that for the common drain parallel PA the drain of the two transistors is connected



Figure 4.26: Efficiency, gain and output power versus frequency.



Figure 4.27: Efficiencies versus load voltage for a carrier frequency of 2.65 GHz and 20.7 MHz modulation frequency.

directly, as can be seen in the schematic provided in Fig. 4.28(a). Its core is designed in a GaN MMIC process [215] to minimize the physical and electrical length of their connection. The corresponding layout is given in Fig. 4.28(b). In order to provide two different supply voltages the DC source potential of  $T_2$  is modified instead of the DC drain supply voltage. Transistor  $T_1$  is supplied with 30 V, while  $T_2$  is biased with 15 V. The biasing for transistor  $T_2$  is done with respect to the supply voltage of transistor  $T_1$ , meaning that the source potential of  $T_2$  is shifted to 15 V in this case. To provide a stable source voltage an on chip shunt capacitor  $C_{DC}$  is included, which is designed to have its parasitic self resonance frequency close to the carrier frequency. Compared to the common source PA this circuit has the advantage of requiring only one DC bias on the RF domain leading to a reduction of the losses and a compact matching network design [S7, S8].

In combination with the direct filter connection the common drain amplifier can be operated highly efficient between the region of the low power  $(T_2)$  and the high power mode  $(T_1)$ . Fig. 4.29 depicts the drain currents and voltage for the common drain PA and a duty cycle of 50 %. For illustrative purposes the high modulation frequency of 132.5 MHz was chosen. During the ON period the high power transistor  $T_1$  is active, being operated with its maximum voltage swing and thus maximum efficiency.

During the OFF period the low power transistor  $T_2$  is active, being also operated at its maximum voltage swing. Due to the special biasing concept the source potential of  $T_2$  is increased to 15 V resulting in a minimum overlap of the drain-source voltage and the drain current of  $T_2$ . This way also  $T_2$  is operated highly efficient during the OFF period. As the source potential of  $T_2$  is higher than the ground potential, transistor  $T_2$  would see a negative drain source voltage, when  $T_1$  is active. To prevent this, the series diode  $D_1$  has been added.



Figure 4.28: Schematic of the common drain PA.

To achieve best performance and enable tuning after manufacturing the input matching network and the output matching network are implemented on PCB level. To achieve the required matching network a codesign of the cavity band pass filter and the PCB output matching was done. For the BPF measured data of the cavity filter presented in section 4.2.3.3 was used, while for the PCB a Rogers 4350B substrate was considered. The resulting arrangement is depicted in Fig. 4.28(c). The transition from the band pass filter to the PCB is essentially the same as the interface of SMA connectors to a PCB and has no major influence.

The LP contours for both operational modes, provided in Fig. 4.30, were used as a starting point for the design of the output matching network. The efficiency maxima  $\eta_{T1}$  and  $\eta_{T2}$  are different in resistance due to different supply voltages and slightly different in reactance. The difference in reactance is due to the fact that the parasitic capacitance of T<sub>2</sub> is different for both operational modes similar to the common source amplifier. Compared to the common source amplifier the efficiency contours of the common drain PA are slightly shifted towards higher inductive regions. This is caused by the fact that the fringe capacitance to ground of C<sub>DC</sub> does not load the RF part of the common drain circuit. In addition the achievable maximum efficiencies are higher, as only one DC biasing at RF level is required, leading to fewer losses and leaving more degrees of freedom for the higher order harmonics' termination.

Also in the design of the common drain PA the focus was set to the efficiency of the low power PA. During the matching network design the orientation of the resonance locus, the fundamental  $Z_{\rm C}$  and the impedances of the higher order harmonics, were considered. In Fig. 4.30(c) it can be seen that resulting load modulation trajectories  $Z_{\rm ON}$  and  $Z_{\rm OFF}$  with respect to the duty cycle d run along efficient regions of the LP contours. Additionally the impedances  $Z_{\rm SIDE}$  for the first 10 modulation sidebands for a modulation frequency of



Figure 4.29: Exemplary drain voltages and currents for the common drain PA for a PWM frequency of 132.5 MHz, a carrier frequency of 2.65 GHz and a duty cycle of 50 %.

20.7 MHz are plotted.



Figure 4.30: Load pull efficiency contours, matching network and load modulation trajectories for a carrier frequency of 2.6495 GHz.

Fig. 4.31 depicts the drain currents and voltages of the final design for low power operation. Only the transistor  $T_2$  is active during that time, being operated with its maximum voltage swing. Its source potential is shifted to 15 V by the biasing concept. It can be seen that the overlap of its drain current and voltage has been minimized, leading to high efficiency. The drain voltage of  $T_1$  is the same as for  $T_2$ . Its DC drain current is basically zero, thus  $T_1$  does not consume power in that operational state. As for the common source PA the gates of both transistors are equipped with a resistive network for stability and biased in class C in order to provide some margin for the drain-gate feedback.

For high power operation, as plotted in Fig. 4.32, the high power transistor  $T_1$  is active. In Fig. 4.32(b) it can be seen that  $T_1$  exhibits its maximum voltage swing and thus is operated highly efficient. The higher order harmonic terminations have been optimized towards class F operation to improve the efficiency. Due to the direct connection of both transistors drains their voltage swing is the same. The series diode  $D_1$  prevents negative drain source voltage for  $T_2$  (with respect to its 15 V source potential). The combination of the parasitic capacitances of  $D_1$  and  $T_2$  acts as a capacitive voltage divider, keeping the drain voltage  $V_{D2}$  in Fig. 4.32(a) above 15 V.

Fig. 4.33 provides the frequency trend of gain, output power and efficiency. It can be seen that due to the narrow BPF used in the output matching network the efficiency quickly degrades when deviating from the design frequency of 2.6495 GHz. Due to the low gain the PAE significantly degenerates compared to the Drain Efficiency (DE). The difference



Figure 4.31: Drain current of both transistors for low power mode.



Figure 4.32: Drain current of both transistors for high power mode.

between the output powers of both operational conditions is roughly 6 dB. In the efficiency traces of both modes in Fig. 4.33(c) it can be seen that their efficiency maxima occur at different frequencies due to the different capacitive loadings of the modes caused by the series diode.



Figure 4.33: Efficiency, gain and output power versus frequency.

The efficiency curves for the design frequency of 2.6495 GHz under static excitation are provided in Fig. 4.34(a). It can be seen that for the CW case a sawtooth like efficiency behavior can be achieved. The maximum efficiency is slightly increased compared to the common source PA, as only a single DC biasing at RF level is required allowing for an optimized matching network.

The resulting BB PWM efficiency curves versus the load voltage are given in Fig. 4.34.

Trace  $\eta_{\text{TH}}$  depicts the theoretical drain efficiency curve (derived in Appendix 7.4) of the circuit in Fig. 4.28, while trace  $\eta_{\text{D}}$  shows the simulation results for a modulation frequency of 20.7 MHz. The difference between the two traces is attributed to the residual losses in the modulation sidebands and due to signal transitions. The gain for the low power region with 8.6 dB is however limited leading to a reduced power added efficiency PAE<sub>D</sub>. For the high power operation with 3 W output power, the gain is around 12 dB leading to an improvement in PAE<sub>D</sub>. The negative PAE for very low output voltages is caused by the class C gate biasing. When considering the insertion loss of 0.65 dB of the matching network and the filter, the drain efficiency  $\eta_{\text{BPF}}$  and the power added efficiency PAE<sub>BPF</sub> are further decreased.



Figure 4.34: Efficiencies versus load voltage for a carrier frequency of 2.65 GHz and 20.7 MHz modulation frequency.

Nevertheless efficiencies still show the highly efficient operation of the direct filter connection after the BPF. Thus the common drain PA is capable of improving the efficiency significantly compared to a standard class B efficiency implementation.

### 4.3.3 Switchable Cascode PA

Cascodes are widely used in analog CMOS circuits and power amplifiers to increase gain and voltage swing by distributing the voltage drop among the stacked transistors [216–222]. In all these designs reduced input power, respectively gate drive, reduces the drain voltage swing on the transistors and degrades the efficiency according to the class of operation [10]. This results in general in poor efficiency for low output power. A possible method to regain efficiency in back off is to perform (discrete) supply modulation [223], which is also known as envelope tracking. Traditionally envelope tracking tackles the problem by modifying the drain voltage of the PA and requires therefore a rather broadband envelope tracking PA and special baseband matching to cope with the envelope expansion of the signal. The switched cascode avoids this problem by using a second supply voltage level inherently available in cascode circuits. To do so the source potential of the upper cascode transistor  $T_1$  is modified according to the signal's amplitude. This is done by adding a switch between the intermediate supply level (V<sub>DC2</sub>) and the two transistors, as depicted in Fig. 4.35 [S9, S10].

When the switch is open the circuit acts like a conventional cascode and the drain voltage  $(V_{D1})$  drops equally across both transistors. Corresponding time domain waveforms for the drain voltages and currents under class B bias and maximum drive for a carrier frequency of 2.65 GHz are given in Fig. 4.36(b).



Figure 4.35: Switchable cascode general circuit.



Figure 4.36: Simulated drain voltages and currents for operation at 2.65 GHz with switch open/closed.

It can be seen that the drain currents of both transistors  $I_{D1}$  and  $I_{D2}$  are equal, as the switch is open and no current flows through it ( $I_S = 0$ ). The voltage is shared equally among the two transistors enabling a higher voltage swing.

By closing the switch  $S_1$  and applying no gate signal to the lower transistor  $T_2$  it is possible to disable the lower part of the cascode [S9,S10]. Due to the class B bias no drain current  $I_{D2}$  flows, although the applied drain voltage is equal to the lower supply voltage as the switch  $S_1$  is closed. This can be seen in the time domain waveforms of the drain voltages and currents in Fig. 4.36(a). As there is no drain current  $I_{D2}$  the resulting power consumption of the lower supply is zero, meaning that the lower part of the cascode is completely powered down. The power generation is done by the upper cascode transistor  $T_1$  only, which operates now as a normal PA with some DC voltage offset, which can also be observed in Fig. 4.36(a).

The drain current  $I_{D1}$  flows now through the switch  $I_S$  and back into the upper power supply  $V_{DC1}$ . As the lower supply current  $I_{DC2}$  and the input power  $P_{IN2}$  are zero, the lower part of the circuit ideally does not consume any power for this type of operation.

For the circuit implementation a GaN MMIC process [215] was chosen, to benefit from the high supply voltages and low parasitics. The high power density of GaN and the use of cascodes allows to easily achieve output powers in the watts range [219–222]. In Fig. 4.40(b) the schematic of the switchable cascode implementation is depicted. The design is split into an integrated part, which contains the cascode circuit and some prematching. The final Input Matching Network (IMN) and Output Matching Network (OMN) are done on board level. This enables the use of high-Q passives and tuning after manufacturing. In Fig. 4.40(b) the layout of the die is depicted. The main circuit consists of the two cascode transistors  $T_1$  and  $T_2$  and the switch  $S_1$ . For these transistors  $4x100\mu$ m HEMTs with

a maximum current of around 400 mA were used. To provide a stable on chip supply voltage  $V_{DC2}$  of 20 V when the switch is closed, the shunt capacitor  $C_B$  was added. Its parasitic self resonance frequency was designed to match the carrier frequency to provide low impedance. The supply voltages  $V_{DC1}$  and  $V_{DC2}$  were set to 20 V.



Figure 4.37: Switchable cascode implementation with broadband load.

In conventional cascode structures in GaN technology the gate-source voltage of the upper transistor is controlled by its own source potential. In order to not overdrive the gate, a capacitive voltage divider, consisting of the parasitic gate-source capacitance and a series capacitor, is used. The division ratio is set in such a way that equal voltage and power splitting is achieved [219–221]. This approach would require very large voltage swings (full drain voltage swing) due to the series capacitor for the signal input of the upper transistor when being in normal PA operation (switch closed). Therefore, it was decided to actively drive the upper transistor also during cascode operation. To decouple the required gate signal from the source voltage swing a transformer was used. During the design of the transformer  $T_{\rm r}$  attention was paid to minimizing the parasitic capacitance of the transformer to avoid increased capacitive feedback. This is necessary to provide the same input impedance for both operational conditions as the capacitive feedback results in a kind of load modulation for the gate impedance during cascode operation.

The gate biasing of the transistor  $T_1$  is done with the biasing inductor  $L_{BIAS}$ . In order not to short circuit the biasing, the transformer has to be decoupled by the series capacitor  $C_S$ . In addition the capacitor  $C_1$  in parallel to the gate and source of  $T_1$  was added to resonate out the transformers inductance and to further reduce the capacitive feedback ratio. When the switch is off, during normal cascode operation, it loads the circuit with its parasitic drain-source capacitance, effectively doubling the drain-source capacitance of  $T_2$ . This would result in an unequal voltage division between the two transistors due to the parasitic capacitive voltage divider at the drain of the transistors. To prevent this the drain-source capacitance of  $T_1$  is increased by adding the parallel capacitor  $C_2$  to its drain. The same effect could also be achieved by increasing the width of  $T_1$ . But with this architecture the optimum load resistance would differ too much for normal PA operation (switch closed), where  $T_1$  already sees twice optimum the load impedance and doubling its size would mean that it sees four times the optimum load impedance resulting in lower efficiency in back off. To achieve stability, resistors were added to the gate of both transistors, with the major focus on the upper transistor  $T_1$ .

For the output matching a combination of microstrip lines based on Rogers 4350B substrate and lumped components was considered. To be capable of providing the correct load impedance for both operational conditions the drain-source capacitance of each cascode transistor  $T_1$  and  $T_2$  has to be resonated out individually (using  $L_1$  and  $L_2$ ), such that the remaining required load impedance is purely resistive. This is necessary because when the switch is closed there is no voltage swing on the lower transistor  $T_2$  and the resulting parallel capacitance is only the drain-source capacitance of the upper transistor  $T_1$ . Whereas for cascode operation the parallel capacitance consists of both parasitic capacitances. If only a single compensation would be used it would not be possible to compensate the capacitances for both operating conditions leading to non optimum load impedances.

To compensate the drain-source capacitances 0201DS SMD inductors from Coilcraft are placed in parallel to the transistors  $T_1$  and  $T_2$ . Instead of being DC-wise decoupled the lower compensation inductance  $L_2$  can be connected to the intermediate supply voltage  $V_{DC2}$  as both circuit nodes have the same DC potential. For the DC decoupling of the inductor  $L_1$  and for all other capacitors the ATC600S series from American Technical Ceramics was considered. For the simulation models including not only the first but also higher order self resonances have been used [224].

The resulting drain voltage and current waveforms for the circuit given in Fig. 4.37 for operation at maximum power at 2.65 GHz are depicted in Fig 4.36(b). It can be seen that both drain currents have the same phase and amplitude implying that the two gate signals are well aligned. The voltage drop is shared equally amongst the two transistors due to the above described measures. As the output matching was optimized to provide highest efficiency, thus reduced overlap of voltage and current, the resulting drain voltages are not sinusoidal as for pure class B operation. The voltage at the switch  $V_{SW}$  has to be pulled down to 0 V in order to avoid conduction of the parasitic gate-drain diode of the switch  $S_1$ . This requires a rather large voltage swing of 20 V to switch between the two operational states.

In Fig. 4.36(a) the drain voltage and current waveforms for normal PA operation and maximum power for the closed switch are given. It can be seen that the lower part is disabled as no drain current  $I_{D2}$  flows. Only a small residual drain voltage swing for  $V_{D2}$  remains due to the finite ON resistance of the switch.

A graph providing the gain and power added efficiency versus load voltage is shown in Fig. 4.38. For cascode operation, as depicted in Fig. 4.36(b), the resulting 64.1% power added efficiency is given in trace  $PAE_{H}$ . The corresponding gain  $g_{H}$  is also shown, where trace  $g_{L}$  depicts the gain for normal PA operation. With the switch closed, the gain  $g_{L}$  is lower than for cascode operation, mainly due to the fact that the ON resistance of the switch acts as a source degeneration and lowers the gain. Both gain curves show an expansion, as the actual gate biasing used is in slight class C.



Figure 4.38: Simulated efficiency and gain versus load voltage for both operating conditions at 2.65 GHz for the broadband load circuit in Fig. 4.37.

For normal PA operation an efficiency peak  $PAE_L$  (8 V) below half the maximum load

voltage (17.1 V), at 6.5 dB back off can be achieved. Its maximum is with 58.2% slightly lower than for cascode operation. This is due to the fact that the load impedance is not split between the two transistors, as for cascode operation, resulting in twice the optimum load impedance. Thus the transistor T<sub>1</sub> operates less efficiently. The second contributor to reduced efficiency is the ON resistance of the switch. It shall be noted that the position of the efficiency peak may be changed by modifying the supply voltages.

The frequency behavior of the circuit is also important, as depicted in Fig. 4.39(c). Efficiency is largest for the design frequency of 2.65 GHz and declines moderately when deviating from the center. For higher frequencies the efficiency for cascode operation  $PAE_{\rm H}$  becomes smaller than the efficiency for normal operation  $PAE_{\rm L}$ .



Figure 4.39: Efficiency, gain and output power versus frequency for the broadband load circuit in Fig. 4.37.

This is caused by the fact that the resulting impedance for that frequencies is smaller compared to the design frequency and for cascode operation maximum voltage swing cannot be achieved anymore. The peak power for cascode operation  $P_{\rm H}$  increases a bit with frequency and reaches 2.74 W at the center frequency of 2.65 GHz. For normal PA operation the peak power at back off  $P_{\rm L}$  remains around 0.57 W over frequency. The gain  $g_{\rm H}$  and  $g_{\rm L}$  for both operational conditions varies in the same manner due to the frequency dependence of the input matching. It shall be noted that the matching networks have not been optimized for broadband operation leaving margin for improvement.

By using a CW excitation the efficiency trace can be improved from a class B efficiency trace to a sawtooth like curve, as depicted in Fig. 4.38. This efficiency curve can be further improved to a Doherty like efficiency behavior by using BB PWM. Therefore, the load of Fig. 4.37 is connected via a BPF to remove the modulation sidebands. In order to avoid any feedback to the PA an isolator is placed between the PA and BPF. The resulting efficiency enhancement is provided in Fig. 4.43(a). For the high power region the theoretical efficiency  $\eta_{\rm TH}$  derived in Appendix 7.4 corresponds well with the simulated drain efficiency  $\eta_{\rm D}$ . The gain of around 12 dB, respectively 13 dB, leads to a reduction in PAE<sub>D</sub> compared to the drain efficiency  $\eta_{\rm D}$ . Considering the 0.53 dB insertion loss of the BPF (in a 50  $\Omega$  environment) the drain efficiency  $\eta_{\rm BPF}$  and PAE<sub>BPF</sub> are further reduced. But they still show the good efficiency curve similar to a Doherty PA.

A drawback of the isolated BB PWM operation is that the power in the modulation sidebands is dissipated in the isolator leading to reduced efficiency. Direct filter connection reduces this dissipated power by providing a high impedance for the modulation sidebands and thus improving efficiency. Fig. 4.40 depicts the switchable cascode circuit with the output matching network optimized for direct filter connection. The main difference to the circuit shown in Fig. 4.37 is that the BPF is connected directly to the PA in order to provide high impedance for the modulation sidebands for a wide frequency range. The resulting arrangement of the filter-microstrip line codesign for the output matching network is given in Fig. 4.40.



Figure 4.40: Switchable cascode implementation with direct filter connection.

For the design of the output matching network the LP contours for both operational states, as provided in Fig. 4.41, are of importance. It can be seen that by individually compensating the parasitic capacitances the efficiency maxima for both operational states are close to the real axis. During the design the orientation of the resonance locus of the load impedance  $Z_L$  was set such that the load modulation trajectories for the low power period  $Z_{OFF}$  and for the high power period  $Z_{ON}$  follow efficient regions. By moving the BPF close to the PA it is possible to provide the desired high impedance  $Z_{SIDE}$  for the modulation sidebands considering 20.7 MHz BB PWM frequency.



Figure 4.41: Load pull efficiency contours, matching network and load modulation trajectories for a carrier frequency of 2.648 GHz for the direct filter connection circuit in Fig. 4.40.

The frequency performance of the direct filter connection circuit presented in Fig. 4.40(a) is dominated by the response of the band pass filter, as can be seen in Fig. 4.42. The

output power, as well as efficiency follow the band pass nature of the filter. Compared to the broad band design of Fig. 4.37 the bandwidth is significantly reduced. But this behavior is a result of the requirement for high impedance at the modulation sidebands and therefore required for the highly efficient BB PWM operation around the design frequency of 2.648 GHz. When comparing the efficiency curves in Fig. 4.42(c) for both operational modes it can be seen that their maxima are well aligned in terms of frequency. This is due to the fact that the parasitic capacitances are almost equally compensated for both operational modes. If only a single compensation inductance (instead of  $L_1$  and  $L_2$ ) would be used to resonate out the  $C_{\rm DS}$  of both transistors it is not possible to achieve a purely resistive load for both operating conditions simultaneously. Thus the efficiency curves would differ significantly in terms of frequency, making efficient operation for both conditions impossible.



Figure 4.42: Efficiency, gain and output power versus frequency for the direct filter connection circuit in Fig. 4.40.

Fig. 4.43(b) depicts the efficiency of the direct filter connection circuit versus the load voltage. It can be seen that the drain efficiency  $\eta_{\rm D}$  is improved compared to the isolated operation depicted in Fig. 4.43(a). But due to residual losses in the modulation sidebands and during the signal transitions the drain efficiency  $\eta_{\rm D}$  falls behind the theoretical possible efficiency  $\eta_{\rm TH}$  for direct filter connection. The gain of around 13 dB leads to a small reduction of PAE<sub>D</sub> compared to the drain efficiency. Accounting for the 0.682 dB insertion loss of the matching network reduces the drain efficiency  $\eta_{\rm BPF}$  and PAE<sub>BPF</sub> after the BPF accordingly. Nevertheless the efficiency behavior still shows the highly efficient connection of the two efficiency peaks due to the direct filter connection.



Figure 4.43: Simulated efficiency enhancement by BB PWM with and without direct filter connection.

It can be concluded that the switchable cascode is a PA structure capable of providing an

efficiency peak in back off. By combining it with BB PWM the efficiency can be improved to a Doherty behavior trend. Beside that the switchable cascode is perfectly suited for direct filter connection with its constant drain current. With the direct filter connection the two efficiency peaks can be connected in a highly efficient way.

#### 4.3.4 Source Modulated Amplifier

The source modulated amplifier is based on modifying the source potential to provide two operational points with maximum voltage swing and thus maximum efficiency [S11]. By modifying the source potential of the transistor the RF biasing and the DC envelope tracking networks are separated. To simplify the concept only two discrete values were considered. The schematic of the source modulated amplifier is depicted in Fig. 4.44(a) and is similar to the switchable cascode PA circuit. The main difference is that the active switch  $S_1$ , with its high control voltage, can be replaced by a diode ( $S_1$ ). Its core is designed in a GaN MMIC process [215] in order to provide lowest parasitics and compact integration of the source modulation circuitry to ensure proper operation. The corresponding layout is given in Fig. 4.44(b).



Figure 4.44: Schematic of the source modulated PA.

The actual PA circuit consists of the transistor  $T_1$  and its input matching. The source modulation is done by the switches  $S_1$  and  $S_2$ . By splitting the DC supply and using a special biasing scheme the source potential of  $T_1$  can be switched between 0 V and  $V_{DC2}$ . The switch  $S_1$  can be implemented as a diode, requiring only  $S_2$  to be controlled. To provide a stable source voltage an on chip shunt capacitor  $C_{DC}$  is included, which has its self resonance frequency close to the carrier frequency.

As the source potential of transistor  $T_1$  is modulated the biasing and the generation of the correct gate signal requires special attention. To decouple the RF signal from the source potential a transformer  $T_r$  is used. For the bias generation a series capacitor  $C_S$  has to be included. The actual biasing is done with the help of a reference diode and a constant current source  $I_{bias}$ . The diode  $D_1$  is implemented as a series connection of 4 diodes to

provide the correct reference voltage for slight class C biasing. For the reference current generation a small transistor with constant gate bias was used.

When combining the source modulated PA with the direct connection of a series resonator highly efficient operation within the region of the low power ( $V_{DC1}=15$  V) and high power mode ( $V_{DC1}+V_{DC2}=30$  V) can be achieved. Fig. 4.45 depicts exemplary drain current and voltage for a duty cycle of 50 %. For illustrative purposes the high modulation frequency of 132.5 MHz was chosen. The source potential is changed according to the duty cycle by a high power digital inverter driving S<sub>2</sub>. During the ON period the switch S<sub>2</sub> is closed and the power transistor T<sub>1</sub> is operated with its maximum voltage swing and thus maximum efficiency.



Figure 4.45: Exemplary drain voltage and current for the source modulated PA for a PWM frequency of 132.5 MHz, a carrier frequency of 2.65 GHz and a duty cycle of 50 %.

During the OFF period the switch  $S_2$  is open and  $S_1$  is conductive causing the source potential  $V_S$  to be shifted to 15 V. Also in this operational regime the power transistor  $T_1$  is operated at its maximum voltage swing and highest efficiency.

To achieve best performance and to enable tuning after manufacturing the input matching network and the output matching network are implemented on PCB level. For best performance of the direct filter connection a codesign of the cavity band pass filter and the PCB output matching was done. The measured data of the cavity filter presented in section 4.2.3.3 was used for the BPF, while for the PCB a Rogers 4350B substrate was considered. The resulting arrangement is depicted in Fig. 4.44(c).



Figure 4.46: Drain current and voltage for both operational modes.

For the design of the output matching network the load pull contours of both transistors, respectively operational modes, are important. The corresponding drain efficiency contours  $\eta_{\rm ON}$  and  $\eta_{\rm OFF}$  for both supply voltages are plotted in Fig. 4.47. The efficiency maxima are different in resistance due to different biasing. For best average performance the focus was set on the efficiency for the low power operation. Therefore, the impedance at carrier frequency Z<sub>C</sub> and the higher order harmonics' termination were optimized to obtain best efficiency in back off. Beside the impedance Z<sub>C</sub> at the carrier frequency the resulting modulation trajectories are important. They depend on the impedance at carrier frequency and the orientation of the resonance locus, as covered in section 4.2.1. Due to the fact that the resulting drain voltage PWM signal for the operation between V<sub>DC2</sub> and V<sub>DC1</sub>+V<sub>DC2</sub> has an offset, the resulting load modulation is limited to a range with good efficiency. The resonance locus was set such that the load modulation trajectories Z<sub>ON</sub> and Z<sub>OFF</sub> of both operational modes follow efficient regions. Additionally the impedances Z<sub>SIDE</sub> for the first 10 modulation sidebands for a modulation frequency of 20.7 MHz are indicated.



Figure 4.47: Load pull efficiency contours, matching network and load modulation trajectories for a carrier frequency of 2.648 GHz.

The dependence of the gain, output power and efficiency on the frequency is given in the corresponding plots in Fig. 4.48. The band pass filter dominates the frequency behavior of the circuit and allows for highest efficiency at the design frequency of 2.648 GHz. When comparing the efficiency maxima in Fig. 4.48(c) for both operational modes it can be seen that they fit very well in terms of frequency. This is due to the fact that the same transistor is active for both operational modes and no modification of the parasitic capacitance compared to the parallel PA or the switchable cascode occurs.



Figure 4.48: Efficiency, gain and output power versus frequency.

Fig. 4.49(a) depicts the efficiency of the source modulated PA under CW excitation. It can be seen that the sawtooth like efficiency behavior for a dual supply concept can be achieved. By combining this property with BB PWM and direct filter connection the efficiency can significantly be improved. The resulting efficiency curves versus the load

voltage are given in Fig. 4.49(b). Trace  $\eta_{\rm TH}$  depicts the theoretical drain efficiency curve of the circuit in Fig. 4.44 according to (7.114) derived in Appendix 7.4, while trace  $\eta_{\rm D}$  shows the simulation results for a modulation frequency of 20.7 MHz. It can be seen that both traces are in good agreement. Unfortunately the gain for the low power region with 9.3 dB is rather limited leading to a reduced power added efficiency PAE<sub>D</sub>. For the high power operation with 3 W output power, the gain is around 13.4 dB leading to an improvement in PAE<sub>D</sub>. When considering the insertion loss of 0.68 dB of the matching network and the filter the drain efficiency  $\eta_{\rm BPF}$  and the power added efficiency PAE<sub>BPF</sub> are further decreased.



Figure 4.49: Efficiencies versus load voltage for a carrier frequency of 2.648 GHz and 20.7 MHz modulation frequency.

It can be summarized that the source modulated PA is well suited to provide an additional efficiency peak in back off region due to its two source potentials. When combining the source modulated PA with the concept of direct filter connection highly efficient operation can be achieved. Another advantage of the source modulated PA is that only a single transistor is active for both operational conditions, reducing the required control signals compared to the parallel PA or the switchable cascode. A single RF signal and the digital signal for the source modulation are sufficient for proper operation. The use of a single active transistor also allows for a compact and efficient output matching network design. The fact that the RF signal has not to be multiplexed between two transistors and that the source modulation is decoupled from the drain signal leads to reduced transition losses between the two states. This results in a drain efficiency  $\eta_{\rm D}$  close to the theoretical limit.

## 4.3.5 BPSK DC Energy Recovery

The concept of DC BPSK energy recovery is introduced in [202] and is based on a modified BB PWM and direct filter connection. One of its main advantages is that it can be implemented using a single (discrete) RF transistor. While the general operational principle is considered in section 3.7.4, this section is dedicated to the analysis of its practical applicability.

The block diagram of DC BPSK energy recovery is given in Fig. 4.50. The circuit consists of a PA, being capable of operating in amplifier and rectifier mode and a band pass filter directly connected to it. The band pass filter is designed to emulate the behavior of a series resonator, providing high impedance for the modulation sidebands and requiring a constant current. Instead of switching the PA on and off, the PA is operated continuously, driving the series resonator with opposite phases (biphase) for each period. The resulting driving signal is the same as for Binary Phase Shift Keying (BPSK) but can also be represented by a PWM signal with amplitude 2 and an offset of -1, as can be seen from the drain voltage envelope  $V_D$  in Fig. 4.50.

During the first period the drain voltage  $V_{D1}$  and the drain current  $I_D$  are positive and the PA is operated as a regular amplifier driving the band pass filter. But during the second period the drain voltage  $V_{D2}$  is negative while the drain current  $I_D$  is still positive due to the series resonator. This provides a negative resistance to the PA resulting in rectifier operation recovering a part of the generated energy and leading to an improvement in efficiency.



Figure 4.50: Block diagram biphase energy recovery.

An important requirement for the design of a biphase energy recovery transmitter is that the used PA is able to efficiently operate as an amplifier as well as a rectifier. In section 2.7 different PA classes being capable of also serving as rectifier have been presented. Amongst them the class E PA was selected due to its high efficiency and switching nature, suitable for purely digital control signals. Fig. 4.51 depicts the corresponding implementation [S4]. It shows a class E PA design with a GaN transistor (Cree CGH60008D) connected to a BPF with a transmission line (TL) matching network that provides the required impedances at the carrier harmonics and the fundamental frequency. For the gate drive a high power driver [225] was considered providing a rectangular driving signal.



Figure 4.51: Schematic of GaN class E biphase energy recovery transmitter using transmission line matching and a cavity resonator.

To achieve best PA, respectively rectifier, efficiency it is important to provide the correct load impedances. This can be well seen in the LP contours in Fig. 4.52, which provide the data for the class E PA of Fig. 4.51 with a supply voltage of 18 V. Fig. 4.52(a) depicts the LP contours of the drain efficiency  $\eta_D$  in 5% steps and the power delivered to the load P<sub>L</sub> in 2 dBm steps for the amplifier operation mode. In Fig. 4.52(b) the corresponding data for rectifier operation (receiving power) is provided. Here the negative complex conjugate of the applied load impedance Z<sub>IN</sub> is plotted to achieve contours located within the smith chart.

To get best system performance it is important that the load modulation trajectories for PA and rectifier operation follow the efficient regions in Fig. 4.52. To achieve that the matching network including the band pass filter has to be designed in such a way that the orientation of the resonance locus, which defines the load modulation trajectory, allows for efficient operation. The symmetry impedance  $Z_S$  was optimized to provide optimum trajectories of the resulting load impedances  $Z_1$  and  $Z_2$  in Fig. 4.53. During the design of the fundamental matching both requirements for  $Z_C$  and  $Z_S$  were considered to get best



Figure 4.52: Load Pull contours at the intrinsic transistor plane for drain efficiency  $\eta_D$  and output power P<sub>L</sub> at a carrier frequency of 2.648 GHz.

performance. The second  $(Z_{H2})$  and third harmonic impedances  $(Z_{H3})$  were tuned to get class E operation. Measured S-Parameters from the filter presented in section 4.2.3.3 were used for simulation.



Figure 4.53: Load pull contours at the intrinsic transistor plane for drain efficiency  $\eta_D$  and load modulation trajectories at a carrier frequency of 2.648 GHz.

The resulting time domain waveforms for amplifier and rectifier operation of the class E circuit are provided in Fig. 4.54. It can be seen that the overlap of drain current and voltage for amplifier operation in Fig. 4.54(a) is minimized and that the waveforms approximate class E behavior. For the rectifier operation, as depicted in Fig. 4.54(b) the drain current is negative while the drain voltage remains positive. This allows for the recovery of energy. Thanks to the minimum overlap of voltage and current for both operational conditions high efficiency can be achieved.

In Fig. 4.55 the resulting efficiency versus the load voltage is given. Trace  $\eta_{\rm D}$  shows the drain efficiency at the intrinsic transistor plane for the desired biphase energy recovery operation. At the maximum output power of 37.1 dBm it reaches a maximum efficiency of 81.9%. But for lower output power the efficiency is lower than the theoretical efficiency curve  $\eta_{\rm LP}$  derived in Appendix 7.5.4. The difference is attributed to the fact that the losses in the modulation sidebands and the signal transitions are not considered. Considering the losses of the modulation sidebands in the matching network ( $Z_{\rm SIDE}$ ) for the theoretical calculation delivers  $\eta_{\rm LP,M}$ , which is close to the efficiency for biphase operation  $\eta_{\rm D}$ . The residual losses are attributed to the signal transitions. When taking the 0.53 dB insertion loss of the filter into account the efficiency drops as depicted in trace  $\eta_{\rm BPF}$ . It shall be noted that the impedances for the modulation sidebands were not optimized yet, leaving some margin for improvement.



Figure 4.54: Drain current and voltage for PA and rectifier operation.



Figure 4.55: Efficiencies vs. load voltage  $V_L$ .

Besides efficiency the achievable spectral purity is of importance. Fig. 4.56(a) shows a comparison between the spectrum of BB PWM and BPSK encoding. It can be seen that the modulation sidebands for BPSK operation are roughly 6 dB higher than for PWM encoding. This is caused by the biphase encoding, resulting in an effective normalized amplitude of 2, which is double the value of PWM encoding. The PSD of the first modulation sideband even exceeds the inband energy, which results in an increased sensibility of the efficiency to the residual losses in the modulation sidebands. Additionally the achievable dynamic range around the inband signal is significantly reduced for the BPSK encoding.

This is due to the fact that the first modulation sideband exhibits a very wide spectrum due to the BPSK encoding, as shown in Fig. 4.56(b). The spectral expansion of the first modulation sideband dominates the achievable inband signal quality. Compared to the PWM encoding only a very small signal bandwidth for the same modulation frequency can be used in order to achieve good signal quality.

Although good efficiency can be achieved with the DC BPSK energy recovery concept, the achievable signal quality is fundamentally limited by the concept itself. As a result the range of possible applications is severely limited.

## 4.3.6 Switched Doherty

The Doherty PA is a well known structure to provide an efficient multilevel PA and its operational principle was already discussed in section 3.1. Beside of controlling the Doherty PA in the conventional way it is well suited for combination with multilevel BB PWM for highly efficient operation.



Figure 4.56: BPSK spectrum.

The block diagram of such a transmitter employing BB PWM on a Doherty PA can be found in Fig. 4.57(a) [226]. It consists of a modulator block, which generates the required control signals, and the Doherty PA itself. The modulator core providing the band limited multi level BB PWM signal is implemented in a Field Programmable Gate Array (FPGA) design<sup>1</sup>. The digital signal of the FPGA is upconverted to the analog RF using an IQ upconversion path consisting of Digital to Analog Converters (DACs) and mixers. The Doherty PA itself is implemented using discrete LDMOS transistors<sup>1</sup>. In order to only feed the inband signal to the antenna a BPF and an isolator are used to connect the load  $R_L$ . Pictures of the corresponding FPGA modulator and Doherty PA implementation are provided in Fig. 4.57(b).



(b) Pictures

Figure 4.57: Multi level BB PWM implementation with FPGA and Doherty PA<sup>1</sup>.

In the modulator the baseband signal  $IQ_{BB}$  is converted into phase  $\varphi$  and amplitude A. The amplitude signal is used to encode the band limited multilevel PWM signal. For the encoding of the PWM<sub>p</sub> signal for the peak PA a normalized reference function between

<sup>&</sup>lt;sup>1</sup>The FPGA design was done by M. Mataln and the PA was designed and built by C. Schuberth.

0.5 and 1 is used. The signal for the main branch  $PWM_m$  is encoded using a normalized reference function between 0 and 0.5. Thus for a normalized amplitude above 0.5 the main PA is constantly turned on, as required for Doherty operation. To account for the load modulation effect of the main PA it is required to take the peaking PA signal into account. This is done by adding the  $PWM_p$  signal to the main branch control signal to achieve optimum control of the Doherty PA.

In Fig. 4.58 the output spectrum of the FPGA for a single level band limited PWM encoding using 4 PWM harmonics is depicted. The inband signal had a bandwidth of 8 MHz and a PAPR of 7 dB, while the PWM frequency  $f_{PWM}$  was set to 25 MHz in order to provide sufficient spacing. It can be seen that around the inband signal in Fig. 4.58 a high dynamic range can be achieved by the band limited PWM approach.



Figure 4.58: Single level FPGA output spectrum for a carrier frequency of 2.1 GHz and a modulation frequency  $f_{PWM}$  of 25 MHz with a signal bandwidth of 8 MHz.

The measured efficiency curves for a static sweep of the multi level BB PWM signal are depicted in Fig. 4.59. It can be seen that the total  $PAE_D$  at the drain including the output power of the modulation sidebands (before the BPF) remains high over the whole range. For the operation between the main and peak PA levels highest values for drain  $PAE_D$  can be held, but for very low duty cycles the efficiency drops as a result of the limited number of harmonics and related rise/fall time of the pulses. Considering the coding efficiency  $\eta_C$  that accounts for the losses in the modulation sidebands leads to the Doherty like efficiency curve PAE<sub>IN</sub> after the BPF at the load. By mitigating the losses in the modulation sidebands through direct filter connection, PAE<sub>D</sub> can theoretically be achieved, improving the efficiency even further.



Figure 4.59: Static total  $PAE_D$  and inband  $PAE_{IN}$  for the Doherty PA and BB PWM operation at 2.75 GHz with a modulation frequency  $f_{PWM}$  of 70 MHz.

From the static efficiency measurement  $PAE_{IN}$  of the whole transmitter chain high average efficiency also for the modulated case can be predicted. Beside the achievable efficiency the output spectrum of the PA, as provided in Fig. 4.60, is of interest. It shows the output spectrum of the Doherty PA including the three modulation sidebands used to encode the inband signal with a modulation frequency of 70 MHz. The inband signal itself has a bandwidth of 14 MHz with a PAPR of 7 dB.

In Fig. 4.60(a) it can be seen that the power in the modulation sidebands is reduced for the multi level operation compared to the single level signal provided in Fig. 4.58. The achieved Adjacent Channel Leakage Ratio (ACLR) without applying predistortion was -27.0 dBc and -26.9 dBc for the lower and upper signal band respectively, as given in Fig. 4.60(b). For the next bands the ACLR goes down to almost -40 dBc, indicating possible improvements by applying predistortion. Beyond that further improvement by predistortion could be achieved if the noise between the carrier and the first PWM modulation harmonic is caused by a deviation of the intermediate output power level, leading to a discontinuity causing spectral regrowth.



Figure 4.60: Doherty PA output spectrum for a carrier frequency of  $2.75 \,\text{GHz}$  with a modulation frequency  $f_{\text{PWM}}$  of 70 MHz and a signal bandwidth of 14 MHz.

The total wideband average output power for the signal provided in Fig. 4.60 was 38.45 dBm with a coding efficiency  $\eta_C$  of 84.89%, which results in an average inband power of 37.72 dBm. The average drain PAE including the total output power was 48.5% for the 7 dB PAPR signal. Even when taking the coding efficiency into account the inband efficiency after the isolator and BPF is still high with 41.17%.

# 4.4 Spectral Considerations

Beside the efficiency performance the resulting signal quality is a very important factor. In order to ensure proper communication and not to interfere adjacent channels or the own receive band disturbing proper communication, the spectral requirements have to be fulfilled.

One challenge resulting from the use of BB PWM includes the rather high signal components of the modulation sidebands next to the carrier, as depicted in Fig. 4.61(b). In order not to interfere with the adjacent channels and to fulfill ACLR requirements a BPF after PA, as shown in Fig. 4.61(a) has to be applied. The filter has to provide sufficient attenuation for the modulation sidebands for good ACLR at the load, respectively antenna.

A second effect related to the modulation sidebands is the possible jamming of the own receive band. As the receive power is much lower than the transmit power, the requirements for spectral purity are even far beyond the ACLR requirements for the receive band. For BB PWM this can be problematic as the receive band Rx may be in the vicinity of the transmit band Tx like shown in Fig. 4.61(b). A duplexer filter or a circulator can be used to provide isolation reducing the cross talk from the transmitter Tx to the receiver Rx. Even when using high quality components there will always be some residual cross talk to the receive band. But as the transmitted data interfering with the receive band is well known, the signal quality in the receive band can be improved by canceling the cross talk. This can be done either in the analog or in the digital domain [227–232] and enables proper operation of the receiver.



Figure 4.61: Receive band issue BB PWM.

Another very important property of PAs is their linearity, which directly relates to the ACLR. For the use of multilevel BB PWM it is very important to identify and eventually track the output levels the PA can generate in order to provide a linear response. Fig. 4.62(a) depicts an example, where the intermediate output level has not been properly identified (error), which leads to a difference compared to the ideal curve used for encoding of the low  $V_L$  and high level  $V_H$  signal.

Fig. 4.62(b) provides the spectra of the low  $V_L$  and the high level  $V_H$  individually. For proper encoding the sum of both signals will cancel out between the inband signal and the carrier resulting in the ideal spectrum shown in Fig. 4.61(b). But if there is discontinuity in the encoding the two signal components will not cancel out perfectly anymore, leading to reduced signal quality.



Figure 4.62: Multilevel BB PWM decomposition and potential error source.

To avoid this effect and correct the error, the signal has to be predistorted in order to provide the correct signal. This can either be done after the PWM encoding, as depicted in the block diagram in Fig. 4.63(a), or before the PWM encoding. Predistorting the signal after the PWM encoding has the big disadvantage of requiring to correct the wideband PWM signal with all its modulation sidebands. This severely increases the required bandwidth for predistortion compared to the pure band limited PWM encoding. Keeping the high bandwidth of the PWM signal in mind this can hardly be implemented in practice and will therefore not be considered.

The second possibility of correcting the signal is by doing the predistortion before the



Figure 4.63: Location of the predistorter.

PWM encoding, as depicted in Fig. 4.63(b). Instead of correcting the whole signal, it is sufficient to predistort the inband signal information only, before it is encoded using the PWM. This way the predistortion has to run only at a multiple of the actual signal bandwidth and not the maximum modulation sideband frequency. Beside correcting the signal for the encoder itself, also the output voltages  $V_{OUT}$  for the different PWM levels can be modified in order to improve the linearity of the PWM operation of the PA itself. To account for the actual output levels of the PA and to cope with the resulting discontinuities the reference voltages  $V_{REF}$  used for the encoding can be corrected instead of leaving the whole correction to the predistorter.

In theory the inband signal linearly depends on the duty cycle, while the harmonics show a sine relationship, as plotted in Fig. 4.64(c). When the ideal BB PWM signal passed through the PA a saturation in terms of amplitude such as in Fig. 4.64(a) occurs. In practice not only the amplitude, but also the phase will be affected. This saturation modifies the applied Band Limited (BL) PWM in a way such that the overshoots of the BL PWM will be severely compressed. In Fig. 4.64(b) a time domain waveform of an ideal BL PWM with 40 % duty cycle signal, as well as the saturated PWM signal are provided.



Figure 4.64: Ideal BL and saturated PWM signal for a duty cycle of 40%.

A disadvantage of predistorting the inband signal only is, that the amplitude relation between the inband (DC) content and the higher order harmonics eventually differs from the theoretical behavior. This could lead to spectral expansion of the modulation sidebands after predistortion of the inband signal. In Fig. 4.65(a) the spectrum of an ideal BB PWM signal decomposed into its signal components is plotted. When considering that the higher order harmonics are distorted, as for the saturated PWM signal in Fig. 4.64(b), the contribution of the higher order harmonics eventually expands into inband signal. This could limit the achievable inband signal quality, even when perfect predistortion is possible. As a consequence the bandwidth of the inband signal compared to the PWM frequency has to be reduced in order to provide sufficient spacing for the first harmonic contribution to decay.



Figure 4.65: BB PWM harmonic decomposition.

It can be concluded that predistortion of the BB PWM signal can either be done after or before the modulator. Predistorting the signal after the PWM encoder results in very large bandwidths and hence tightens the requirement. By predistorting the signal before the PWM encoder the required bandwidth after the PWM encoding remains unaffected. The only requirement is that sufficient spacing between the inband signal and the first PWM harmonic has to be assured. Beside the nonlinearity also the spectral mask and the receive band requirements have to be considered. To fulfill ACLR requirements and to provide low output power next to the band in order not to interfere with adjacent channels, a BPF with sufficient attenuation has to be employed. To ensure proper operation of the receiver digital or analog cancellation of the cross talk can be used.

# 4.5 Summary BB PWM

In this chapter the properties and prospects of BB PWM have been discussed in detail. Based on the theoretical considerations presented in chapter 3 the requirements for the circuit implementations for direct filter connection have been derived. The influence of the coupling on the input impedance of cavity filters has been studied in detail, as it is the starting point for the matching network design. Based on this, possible matching network architectures have been evaluated. In order to provide high impedance for the modulation sidebands over a wide bandwidth the connection between the filter and the transistor has to be as short as possible. Depending on the orientation of the resonance locus, which is influenced by the coupling, a certain length of transmission line is required to shift the resonance locus towards the open to emulate the behavior of an ideal series resonator. This delay on the one hand reduces the bandwidth with high impedance around the carrier, but on the other hand allows for impedance matching.

Other important aspects are the load modulation trajectories of direct filter connection, for which generalized equations have been derived. Equivalent load impedances for the two different states of the PWM signal have been defined and based on them a design methodology considering the load pull contours and the load modulation trajectories has been developed. During the design a special emphasis was put not only on the carrier impedance matching, but also on the orientation of the resonance locus, as this significantly influences the load modulation trajectories.

The required signal generation chain for optimum operation has been discussed. The difference between digital PWM generation with its aliasing problem and the implemented band limited PWM, which avoids this problem, have been highlighted. A signal conditioning approach to account for residual fluctuation of the filter current in real circuits in order to drive the transistor the optimum way also within the PWM periods has been presented. Two different approaches of correcting the nonlinear behavior of real circuit implementations have been suggested and their difference has been highlighted. In addition a possible solution to avoid the jamming of the own receive band has been discussed.

By analyzing the performance of a switched Doherty PA under BB PWM excitation with isolated load, the possible efficiency improvement by direct filter connection has been highlighted. Based on the derived requirements for direct filter connection four different integrated circuits capable of dealing efficiently with the constant filter current have been developed. Two of the circuits are based on a parallel approach on common source and common drain basis. Both circuits show the highly efficient operation of direct filter connection in the high power region. Due to the required multiplexing between the two transistors the efficiency performance falls slightly behind the theoretical limit. The other two circuits designed are based on series approaches. One of them is the switchable cascode, which makes use of the second voltage level inherently available in cascode structures by adding a switch. This way a discrete supply modulation is possible, leading to a sawtooth like efficiency curve. By combining the switched cascode with BB PWM and direct filter connection highly efficient operation can be achieved. Another promising circuit is the source modulated PA. Its basic concept is to provide (discrete) supply modulation at the source of the PA instead of the drain. This way the RF and BB envelope generation can be separated simplifying the matching network design. The efficiency traces of the source modulated PA for direct filter connection are close to the theoretical limit, as the RF signal is applied to a single transistor, mitigating the demand for switching. The source potential is modulated according to the PWM signal by a purely digital control signal. In addition to the multi level PWM circuits the performance of a biphase direct filter connection concept requiring a single discrete transistor has been evaluated. By using a constant envelope signal and modulating the phase between  $0^{\circ}$  and  $180^{\circ}$  it is possible to generate power in one period and recover energy in the other period. Based on a class E circuit design it has been shown that efficiency enhancement in back off by using a single transistor only is possible. One of the drawbacks of this circuit is that the PA and rectifier efficiency have to be very high in order to achieve significant efficiency enhancement in back off. Another disadvantage of the concept is the poor spectral quality as a result of the large spectral expansion of the first modulation sideband. This requires sufficient spacing and a high ratio of the PWM modulation frequency to the signal bandwidth. As a result, good signal quality can only be achieved for very narrowband signals.

For the multilevel PWM circuits this requirement is relaxed, but still sufficient spacing in order to allow for predistortion and to account for filter requirements is necessary. Considering the design limitation for the matching network, which impose a limit for the maximum modulation frequency, the achievable signal bandwidths are limited by the operational principle. Although excellent efficiencies can be achieved, BB PWM can only accommodate for ever rising signal bandwidth up to the limit defined by the operational principle and practical implementation.

# 5. RF PWM

In this chapter the properties of RF PWM operation with respect to PA efficiency, spectral emissions, modulator implementation and requirements as well as signal processing aspects will be discussed.

Compared to BB PWM the use of RF PWM has significant advantages in terms of spectral emissions close to the carrier. For BB PWM the modulation sidebands of the PWM signal are close to the carrier, as depicted in Fig. 5.1. In order not to interfere with adjacent channels high quality filters are required to remove the modulation sidebands. The first modulation harmonic for RF PWM are simply the second carrier harmonic, thus providing a large spacing to the signal. This drastically simplifies the filtering task compared to BB PWM.



Figure 5.1: Comparison of BB PWM and RF PWM spectrum.

Beside the spectral emission the ever increasing bandwidth is a very important aspect. Hence a higher level of flexibility and reconfigurability of the transmitters, respectively modulators, is required. Thus the trend goes towards digital solutions moving the digital domain closer towards the antenna. Digital circuits can additionally support multi band and multi standard operation leading towards Software Defined Radio (SDR). Another aspect is the low power consumption of digital modulators and their benefit from technology scaling. While for BB PWM a conventional analog IQ modulator is required to avoid the aliasing problem, the RF PWM signal is purely digital up to the PA.

In Fig. 5.2(a) the block diagram of a conventional analog IQ modulator is given. The BB IQ signal undergoes a DAC and is low pass filtered to remove aliasing components. The analog I and Q components are then upconverted using a quadrature mixer and the analog signal is fed to the PA. A drawback of this method is the rather large power consumption of the DACs and the mixer. The circuit also suffers from analog imperfections such as (frequency dependent) IQ imbalance, skew mismatch and DC offset, which lead to images and carrier feedthrough if not properly compensated.



(a) Conventional analog IQ transmitter.

(b) Digital RF PWM transmitter.

Figure 5.2: Block diagram of analog/digital transmitter.

Digital circuits on the other hand do not suffer from such problems. An example of such a digital modulator is provided in Fig. 5.2(b), which depicts the block diagram of a digital

RF PWM transmitter [233–237]. Up to the PA the signal is purely digital and also the PA might be operated purely digital as a Switched Mode Power Amplifier (SMPA) [238]. In this case the amplitude information is introduced by the pulse length, respectively duty cycle, of the signal and the phase is represented by the pulse position.

Similar architectures can be found in [204, 239] and [240]. Other purely digital solutions are for example all digital IQ modulation [241, 242] and all digital Phase Locked Loops (PLLs) [243, 244]. These developments have led to the concept of RFDACs [245, 246] with the possible resolution enhancement in combination with RF PWM [247]. Beside those solutions it is also possible to use  $\Delta\Sigma$  modulators either for direct signal generation [248] or to enhance the resolution [249–251] or efficiency [252].

The focus of this chapter is on RF PWM due to its single bit output signal eliminating the demand for any DAC, while high timing resolution in combination with low power consumption [233–237] can be achieved. In the first section the properties of different PA classes under RF PWM operation in terms of efficiency and nonlinear behavior will be discussed. In the subsequent section two different modulator architectures will be compared and their operational principle will be discussed. Additionally the combination with  $\Delta\Sigma$  noise shaping for resolution enhancement will be covered.

Due to the discrete time nature of the digital RF PWM generation the achievable spectral quality, especially for broadband signals, is limited by aliasing effects. The effect is described in section 5.2.2 and a method to mitigate this effect is introduced in section 5.2.3.

The properties of driver circuits are covered in section 5.2.4 and lead in general to nonlinear behavior and reduced signal quality. In order to compensate the driver's influence predistortion of the pulses is required. Therefore, a special predistortion scheme dedicated to RF PWM capable of recovering best performance is proposed in section 5.2.5. Finally digital mismatch compensation and its combination with  $\Delta\Sigma$  noise shaping is discussed.

# 5.1 RF PWM PA Characteristics

Very important aspects are the properties of the PA under RF PWM excitation, which lead to SMPA operation. SMPAs are known for their high efficiency due to the purely switching operation. Therefore, the performance of different PA classes under RF PWM excitation will be discussed in this section. The efficiency behavior, as well as the control characteristic of the circuits is analyzed. The analysis is based on ideal transistor models in order to investigate the influence of the RF PWM operation without experiencing any other efficiency degradation. Nevertheless the limits in terms of maximum voltage and current of real transistors are considered in the discussion.

## 5.1.1 Class B

In this section the influence of RF PWM operation on linear PAs is presented. The basic circuit of a class B PA for RF PWM operation is depicted in Fig. 5.3(a). Instead of feeding a sine signal to the transistor's gate the RF PWM signal is directly applied, switching the transistor either ON or OFF. The higher order harmonics are short circuited for class B operation and thus the load voltage depends on the fundamental drain current of the RF PWM signal only. To achieve class B operation the load resistor is designed to achieve the maximum voltage swing for a duty cycle of 50 %. The resulting control characteristic for this case follows the theoretical sine curve of the fundamental contribution of the RF PWM signal, as provided in Fig. 5.3(c). A drawback of RF PWM compared to conventional class B operation is the slightly increased overlap of voltage and current in the 50 % duty cycle region. This can be seen in Fig. 5.4(b), which depicts the drain current and voltage

for 50% duty cycle. As a result the efficiency at maximum output power is lower than the conventional class B performance, as shown in Fig. 5.3(b).



Figure 5.3: RF PWM operation class B and class C.

For very low duty cycles the efficiency is similar to the class B performance. Although the drain current amplitude is at its maximum, as given in Fig. 5.4(a), the reduced conduction angle limits the efficiency degradation. In Fig. 5.3(b) it can be seen that the efficiency for the class B design follows a linear trend, while it starts to saturate towards higher output powers. For duty cycles above 50% the efficiency rapidly drops due to the increased conduction period in combination with the reduced output power. This is caused by the large overlap of voltage and current as depicted in Fig. 5.4(c).



Figure 5.4: Time domain waveforms RF PWM operation class B.

When increasing the load resistance class C operation can be accomplished. For twice the load resistance of class B operation the output voltage starts to saturate already at a duty cycle of around 17%. The fundamental current contribution already results in maximum drain voltage swing, as can be seen in Fig. 5.5(b). Due to the reduced conduction angle for maximum output power the efficiency of the class C PA under RF PWM operation is increased compared to the class B behavior, as depicted in Fig. 5.3(b).



Figure 5.5: Time domain waveforms RF PWM operation class C.

Nevertheless no significant efficiency enhancement in back off is possible by the RF PWM

operation, as the PA is not operated in a purely switched way. This is caused by the fact the the demodulated drain voltage leads to an overlap of current and voltage and thus to a reduced efficiency for lower output powers. The class J PA essentially shows the same trend and is therefore not discussed.

### 5.1.2 Class F

Compared to the class B PA the class F PA is operated in a true switched mode and achieves ideally 100% efficiency for maximum output power, as can be seen in Fig. 5.6(b). The basic circuit of the class F amplifier for RF PWM operation is depicted in Fig. 5.6(a). Instead of driving the gate in the conventional way using a sinusoidal drive and class B biasing conditions it is controlled directly by the RF PWM signal. For maximum output



Figure 5.6: RF PWM operation class F.

power with 50 % duty cycle the drain current and voltage correspond to the regular class F operation due to its saturated operation. This can be seen in Fig. 5.7(b), which depicts the half sinusoidal drain current and the rectangular drain voltage waveform. As there is no overlap of drain current and voltage the transistor is operated highly efficient.

Reducing the duty cycle of the signal results in a residual voltage at the beginning of the ON period, preventing Zero Voltage Switching (ZVS), and resulting in capacitive losses. Only for a duty cycle of 50 % the continous current of the series resonator is fully provided by the quarter wave line. A deviation from the 50 % case causes a residual current charging the parasitic capacitance  $C_{DS}$  of the transistor resulting in capacitive losses. The drain voltage can reach very large values and can even become negative, as for the 25 % duty cylce example in Fig. 5.7(a).



Figure 5.7: Time domain waveforms RF PWM operation class F.

The capacitive losses lead to an almost linear dependence of the efficiency on the load voltage, as provided in Fig. 5.6(b). The value of the parasitic capacitance has only a minor impact on the overall efficiency trend, although it determines the maximum voltage excursions. But at the same time it influences also the highly nonlinear output voltage

characteristic in Fig. 5.6(c) such that the efficiency with respect to output voltage varies insignificantly. Therefore, only the traces for a single capacitance value are plotted. Similar to the class B PA the efficiency rapidly decreases above 50 % duty cylce due to the reduced output power, while the conduction angle and thus DC power consumption increases.

Beside the standard also the inverse class F PA exists. Its schematic is depicted in Fig. 5.8(a). Regular operation with 50% duty cylce results in a rectangular drain current and a half sinusoidal drain voltage, as plotted in Fig. 5.9(b). As there is ideally no overlap of drain voltage and current a theoretical efficiency of 100% can be achieved.



Figure 5.8: RF PWM operation inverse class F.

But when deviating from the 50% duty cyle operation only a part of the constant biasing current of the RFC flows towards the load during the OFF period. Similar to the class F PA this current charges the parasitic capacitance. This effect can be seen in Fig. 5.9(a), which depicts the drain voltage and current for 25% duty cycle operation. The voltage at the parasitic capacitance during switching ON leads to capacitive losses that reduce the efficiency.



Figure 5.9: Time domain waveforms RF PWM operation inverse class F.

Also for the inverse class F PA the value of the parasitic capacitance mainly influences the maximum voltage, but has only minor impact on the efficiency curve, due to its interdependence with the output voltage control characteristic in Fig. 5.8(c).

It can be summarized that although the class F PA is operated in highly efficient switched mode operation, the capacitive losses degrade the efficiency to a roughly linear behavior, preventing efficiency enhancement in back off.

### 5.1.3 Class D

The class D PA circuit family is basically an extension of the class F circuits to a push-pull configuration. The schematic of the current mode class D PA can be found in Fig. 5.10(a). It consists of two PA stages, with the load connected between them. For the operation

with RF PWM two different operational modes are possible. The second signal can either be the complementary (inverted) signal of the first one or a 180° delayed version of it.



Figure 5.10: RF PWM operation current mode class D.

Both operational modes have different properties, but for 50 % duty cycle the resulting time domain waveforms, as depicted in Fig. 5.11(b), are independent of the signal generation. The drain current and voltage for complementary (inverted) RF PWM control with a duty cycle of 25 % are provided in Fig. 5.11(a). It can be seen that large negative voltages occur at the drain of the first transistor and that the resulting voltage swing is larger than for 50 %. This is caused by the boundary conditions, as shown in Appendix 7.2.1, and leads to an inverse control characteristic of the output voltage as depicted in Fig. 5.10(c). In theory the circuit is 100 % efficient, when neglecting capacitive losses. But due to the inverted control characteristic, the large negative voltages and the increased current for low duty cycles, the practical applicability of this operational mode is limited.



Figure 5.11: Time domain waveforms RF PWM operation current mode class D.

The second operational mode is based on a  $180^{\circ}$  delayed control signal for the second transistor and is equal to the RF PWM operation of the inverse class F PA. This can be seen well in Fig. 5.11(c), which depicts the time domain waveforms for a duty cycle of 25 % and the delayed control scheme. As for the inverse class F, capacitive losses occur, which lead to a reduced efficiency as depicted in Fig. 5.10(b). Without parasitic capacitances the constant current of the biasing would drive into an open circuit, leading to very large voltage peaks, far beyond the limits of the device and furthermore to a very large voltage during the beginning of the transistor's conduction period. Similar to the inverse class F the efficiency curve of the current mode class D PA with delayed control depends only weakly on the parasitic capacitance.

The voltage mode class D PA, as depicted in Fig. 5.12(a) is the push-pull counterpart of the class F PA.

As for the current mode class D PA, two operational modes are possible, which are equal for 50% duty cycle. The drain current and voltage time domain waveforms for 50% duty cycle operation are provided in Fig. 5.13(b). It can be seen that the waveforms of a single



Figure 5.12: RF PWM operation voltage mode class D.

transistor correspond to the ones of a class F PA. While for the current mode class D it is possible to operate the circuit efficiently with complementary signal generation, it is shown in Appendix 7.2.2 that this is not possible for the voltage mode class D. The DC biasing determines the DC content of the drain voltage waveforms and results in a large overlap of voltage and current for duty cycles deviating from 50 %. An example for 25 % duty cycle operation is provided in Fig. 5.13(a). It can be seen that the drain current of transistor 1 is negative and the drain current and voltage of transistor 2 have a very large overlap. This overlap results in a rather poor efficiency of this operational mode (inverted), as can be seen in Fig. 5.12(b). Also the control characteristic of the output voltage, as depicted in Fig. 5.12(c), is highly nonlinear.



Figure 5.13: Time domain waveforms RF PWM operation voltage mode class D.

The properties of the operational mode based on a  $180^{\circ}$  delayed control signal are essentially the same as for the RF PWM control of the class F PA and have already been discussed.

It can be summarized that for the voltage and current mode class D PA no significant efficiency enhancement in back off is possible by RF PWM operation.

### 5.1.4 Class E

In this section the properties of the class E PA under RF PWM operation will be discussed. The class E PA is a true switched mode PA. Its design offers various degrees of freedom. One of the most important parameters is the q-value, which basically defines the ratio of the resonance frequency of the shunt elements with respect to the operational frequency [80–82]. In addition to that the class E PA can be designed for different duty cycles. Fig. 5.14(a) depicts the schematic of the class E PA for RF PWM operation. For the results presented a range of q-values and a design duty cycle of 50 % is considered. The resulting efficiency curves with respect to maximum load voltage are depicted in Fig. 5.14(b). It can be seen that the efficiency curves show a roughly linear increase until reaching their

maximum efficiency value, slightly below maximum output power. The corresponding control characteristics of the output voltages provided in Fig. 5.14(c) are highly nonlinear. For duty cycles above the design value of 50 % the output voltage, respectively power, increases only very moderately.



Figure 5.14: Efficiency and control characteristic RF PWM operation of class E designed for 50 % duty cycle.

The time domain waveforms for a class E with a q-value of 1.3, which is designed for and operated at 50% duty cycle, are provided in Fig. 5.15(b). It can be seen that there is no overlap of voltage and current at any time, enabling highly efficient operation.



Figure 5.15: Time domain waveforms RF PWM operation class E (q=1.3).

When reducing the duty cycle, deviating from the design duty cycle, the ZVS condition cannot be held anymore and capacitive losses lead to efficiency degradation. This can be seen well when observing the time domain waveforms for 25 % duty cycle operation depicted in Fig. 5.15(a). The residual voltage during switching ON causes capacitive losses that limit the efficiency. The required shunt capacitor value for class E operation is defined by the design equations. Its dependence on the supply voltage and operational frequency makes the efficiency degradation independent of the two parameters. This means that no performance improvement for different supply voltages or operational frequencies is possible.

Another important aspect of class E PAs is their high peak voltage compared to the DC supply voltage. In Fig. 5.16 the maximum drain voltage for different q-values with respect to duty cycle are plotted. For duty cycles above 50 % the maximum drain voltage quickly starts to increase, reaching very high peak values. At the same time the output voltage, respectively output power, increases only slightly in this region.

Considering the large increase of maximum drain voltage for a small benefit only in output power it is not very useful to operate the circuit above the design duty cycle. In Fig. 5.17 the efficiency and control characteristics for operation limited to 50% duty cycle are depicted.



Figure 5.16: Normalized maximum drain voltage RF PWM operation class E designed for 50% duty cycle and operation until 100% duty cycle.



Figure 5.17: Efficiency and control characteristic RF PWM operation class E designed for 50% duty cycle and operation limited to 50% duty cycle.

It can be seen that the efficiency curves in Fig. 5.17(a) show an almost linear trend and no significant efficiency enhancement is possible. This means that by simply modifying the duty cycle no efficiency enhancement can be accomplished. The analysis of class E PAs designed for different duty cycles basically delivered the same trends and is therefore not discussed further.

### 5.1.5 Class DE

In this section the properties of the class DE amplifier depicted in Fig. 5.18(a) with respect to RF PWM will be presented. The operational principle of the class DE amplifier was already covered in section 2.5 and essentially eliminates the capacitive losses of the conventional voltage mode class D by leaving a dead time between the two conduction periods. The load current charges and discharges the parasitic capacitance during this time which results in ZVS and enables highly efficient operation. The class DE amplifier can be designed to operate at different duty cycles, therefore implementations for duty cycles d from 5% to 45% in 5% steps have been considered for this analysis. Fig. 5.18(b) depicts the corresponding efficiency curves with respect to the normalized load voltage. It can be seen that the efficiency peak for some implementation variants is slightly shifted towards lower output voltages, but in general the efficiency shows a roughly linear dependence on the load voltage.

The slight shift of the efficiency peak towards back off dominantly occurs for implementations designed for lower duty cycle operation. These implementations have the drawback that the design duty cycle is already rather low and that controlling the PA with a duty cycle below the design value only results in a minor decrease in output voltage. This can be seen in Fig. 5.18(c), which depicts the relation of the output voltage to the applied duty cycle. Class DE PAs, which are designed to operate at higher duty cycles show a better control characteristic in terms of output voltage coverage.



Figure 5.18: Efficiency and control characteristic RF PWM operation of class DE for designs from 5 % to 45 % duty cycle.

The main contributor to the efficiency degradation are the capacitive losses, when deviating from the design duty cycle. In Fig. 5.19(b) the time domain waveforms of a class DE PA designed and operated at 30 % duty cycle are plotted. Due to the dead time between the two conduction periods ZVS is possible and thus no capacitive losses occur that degrade the efficiency. When controlling the PA with a higher duty cycle, such as 50 % as depicted in Fig. 5.19(c), the dead time between the two conduction periods is not long enough to charge/discharge the parasitic capacitance. This results in capacitive losses, which reduce the efficiency. Another drawback is that the drain current is negative at the beginning of the conduction period. Similar problems occur, when operating the PA with duty cycles below the design value. Fig. 5.19(a) depicts the corresponding waveforms for operation at 10 % duty cycle. Compared to the design value of 30 % the dead time between the two conduction periods is increased, which causes the drain voltage to increase again due to the current of the series resonator. As a result ZVS is not possible and the capacitive losses limit the efficiency.



Figure 5.19: Time domain waveforms RF PWM operation class DE designed for  $30\,\%$  duty cycle.

The same trend in terms of efficiency, control characteristic and time domain waveforms has also been observed for the class DE PA with parallel compensation inductance and therefore the results are not shown. Although the class DE PA can be operated highly efficient at the design duty cycle, any deviation from the design value results in capacitive losses, that prevent efficient operation over a wide range. Thus no significant efficiency enhancement in back off by RF PWM operation can be achieved. Supply voltage and operational frequency have, as for the class E PA, no influence on the efficiency behavior, as the required shunt capacitance for class DE operation scales accordingly. As a result the capacitive losses with respect to the output power are independent of supply voltage and frequency.
## 5.1.6 Summary RF PWM PA Characteristics

In this section the properties of different PA classes under RF PWM excitation have been investigated. It has been observed that no significant efficiency enhancement in back off by controlling the PA with RF PWM can be achieved. For linear PAs this is attributed to the fact that the drain voltage is demodulated, which leads to overlap of current and voltage. In the case of SMPAs capacitive losses dominate and cause an almost linear efficiency behavior. Even considering very small parasitics makes no difference, as these result in very large peak voltages which cause the capacitive losses during switching ON to remain roughly constant due to the increased voltage.

It can be summarized that the maximum efficiency can be increased by using SMPAs, but in general no efficiency enhancement in back off compared to the linear (class B) efficiency behavior is possible by simply applying RF PWM control. Only the inverter based voltage mode class D PA, as discussed in section 3.8, is capable of providing good efficiency in back off, if its capacitive losses become negligible. Thus, to enhance the efficiency, the RF PWM control has to be combined with other techniques such as load modulation [169, 170, 238] or multilevel PAs.

# 5.2 Modulator

This section is dedicated to the properties of the RF PWM modulator itself. Even though no significant improvement of the PA efficiency for single level PAs can be achieved, the use of RF PWM is still appealing due to its digital output signal and the modulator flexibility. As a result of the purely digital output signal the power consumption of the modulator can be reduced. This is an important property for low power transmitters, where the modulator starts having a significant contribution to the total DC power consumption. In Fig. 5.20(b) the impact of the modulator DC power consumption with respect to the average output power and average PA efficiency  $\eta_{PA,AVG}$  is depicted. The traces are parameterized with the parameter  $k = P_{PA, AVG}/P_{MOD}$ , which is the average PA output power with respect to the DC power consumption of the modulator.



Figure 5.20: RF PWM for direct digital to RF conversion.

For low average output powers (low k) the PA efficiency becomes less important and the DC power consumption dominates the total system efficiency  $\eta_{SYS}$ . Even for moderate output powers the modulator efficiency has a significant impact on the total system efficiency. In order to benefit from the good modulator efficiency and to introduce an intermediate step towards a fully switched mode transmitter the RF PWM modulator can be used to provide

direct digital to RF conversion. Fig. 5.20(a) depicts such a block diagram. The digital RF PWM signal from the modulator is filtered by the BPF before being fed to the (analog) PA. This operational mode will be considered in this chapter in order to investigate the achievable signal quality. Based on the presented considerations the architecture can be expanded to fully (multilevel) switched mode PAs in a subsequent step.

First the architecture of the modulator and its specific properties will be discussed. Later the aliasing problems of digital discrete time RF PWM will be described and a method based on Cross Point Estimation (CPE) to resolve the problem will be proposed. The properties of digital drivers and their impact on signal quality is covered. In order to compensate the nonlinear effects of the driver a dedicated predistortion scheme will be presented. Finally the digital compensation of device mismatch will be discussed.

#### 5.2.1 Architecture

In this section the architecture and implementation details of the RF PWM modulator will be discussed. The generation of a digital RF PWM signal is usually based on the generation of two outphasing signals with 50% duty cycle each. The final RF PWM signal is generated by a logical AND operation on the two outphasing signals. Fig. 5.21 depicts the corresponding block diagrams. In general two approaches based on either a series [237,253] or a parallel [240] architecture are possible. For both cases it is necessary to convert the IQ data to polar representation. This can be done by means of a COordinate Rotation DIgital Computer (CORDIC) [254]. The amplitude is then encoded by the inverse cosine to deliver the outphasing angle  $\Phi_A$ . The inverse cosine function is required in order to account for the sinusoidal relationship between duty cycle and fundamental signal amplitude.



Figure 5.21: RF PWM modulator block diagram.

In case of the series architecture in Fig. 5.21(a) the phase information  $\Phi$  is introduced by the Phase Modulator (PM), with its output signal sig<sub>P</sub>. In Fig. 5.22(a) an encoding example for the series architecture is provided. As the phase of the pulse is at its center, the outphasing signal  $\Phi_A$  has to be considered as well in the control signal generation for the PM. The amplitude information, respectively the PWM signal, is generated by delaying the phase modulated output signal sig<sub>P</sub> by twice the outphasing angle  $2\Phi_A$ . The result is a delayed version sig<sub>A</sub> of the phase modulated signal sig<sub>P</sub>. The final RF PWM signal is generated by the logical AND operation on sig<sub>P</sub> and sig<sub>A</sub>.

For the parallel architecture in Fig. 5.21(b) the procedure is similar, with the main difference that two full PMs are used. The outphasing signal for each branch is generated directly in a parallel fashion. In Fig. 5.22(b) an example of the signal generation for the parallel architecture is provided. The two outphasing signals sig<sub>1</sub> and sig<sub>2</sub> are shifted to each other with respect to the outphasing angle  $\Phi_A$  and the phase  $\Phi$  is considered by shifting both signals accordingly. The final RF PWM signal is generated by a logical AND operation on both outphasing signals  $sig_1$  and  $sig_2$ .



Figure 5.22: RF PWM signal generation.

An advantage of the series approach is that the Amplitude Modulator (AM) has only to cover a  $180^{\circ}$  shift in order to generate the duty cycle (0-50 %), while the PM has to cover the whole  $360^{\circ}$ . The AM can be implemented in the same way as the PM, but requires only half of the delay elements. This means that the series architecture results in a smaller implementation with lower power consumption. But the series architecture has some drawbacks compared to the parallel architecture. The fact that the output signal sig<sub>P</sub> of the PM travels through the delay elements of the AM, which takes up to  $180^{\circ}$ , can lead to inconsistent output states during that period, especially for rapidly varying signals.

In Fig. 5.23(a) a possible implementation of the phase modulator is depicted. It consists of a delay line with unit delay elements, which is fed with the rectangular LO signal. As a result the phase shifted copy of the LO signal is present at the output of the unit delay elements. The unit delay can be implemented by an inverter chain with its basic delay including further resolution enhancement by resistive interpolation. The output signal sig<sub>x</sub> is selected by a multiplexer, which is controlled by  $\Phi_x$ . This means that the output signal is a circular shifted version of the LO signal.

In order to compensate temperature drifts of the delay line a locking mechanism, as depicted in Fig. 5.23(a), can be applied [255, 256]. In this aspect the series architecture has another disadvantage in terms of locking of the AM. For the PM the input signal is the constant rectangular LO, which is used as reference for the last output tap of the delay line. But for the AM the input and thus reference signal of the delay line is the phase modulated output signal of the PM, which imposes a challenge for the locking mechanism. The series architecture has also disadvantages in terms of mismatch compensation, as the errors of the PM and AM are not independent, which requires a more sophisticated identification and compensation scheme. For the parallel architecture the impact of the mismatch of both modulators is independent and therefore individual compensation for each modulator branch, as covered in section 5.2.7, is possible. Due to this drawbacks of the series architecture, the parallel architecture will be considered in the further analysis.

Another important property is the achievable resolution of the phase modulator. In case of a digital delay line the resolution is defined by the basic inverter delay. Resolution enhancement by using resistive interpolation is limited to a small factor in practice [237, 257]. To provide even better resolution it is possible to use a cascaded approach as depicted in Fig. 5.23(b). The coarse delay is provided by the delay line and the fine delay done by an adjustable delay element. This can be done by switching the number of inverted cells driving a variable capacitive load [240]. For the separation of coarse and fine delay it



Figure 5.23: Digital phase modulator implementation.

is important to implement only a very short fine delay section in order to minimize the impact of run time effects on the signal quality.

Directly related to the phase resolution of the PMs in Fig. 5.22(b) is the combination/quantization block. A possible implementation is depicted in Fig. 5.24(a). After the decomposition for the single modulator branches the signals  $\Phi_1$  and  $\Phi_2$  are quantized and fed to the modulator. To provide a feedback signal to enable noise shaping of the quantization error the quantized IQ<sub>Q</sub> value is calculated in the reverse manner to the signal decomposition.



Figure 5.24: Combination and quantization block.

The modulator has an interesting phase relationship, as the phase of the generated IQ signal depends on the pulse length. The phase of a pulse is at exactly half of the pulse length, thus the phase for odd pulse lengths is shifted by half of the phase resolution  $\Delta \Phi/2$  compared to even pulse lengths. This can be seen in Fig. 5.24(b), which depicts the possible IQ constellation points for 4 bit resolution. The samples on amplitude circles with odd pulse lengths are shifted in phase by  $\Delta \Phi/2$  compared to the IQ samples for even pulse lengths. This effect results in better coverage of the IQ plane. By quantizing the outphasing signals, as shown in Fig. 5.24(a), the sample with the minimum IQ quantization error is selected.

The quantization noise floor is determined by the achievable phase, respectively timing resolution, of the modulator. Fig. 5.26(a) depicts examples for the quantization noise floor of a 40 MHz bandwidth signal generated by an ideal modulator operating at 2.6 GHz with 6 to 9 bit resolution. The desired reference signal as well as the quantization error (generated - reference signal) is plotted. At DC and at twice the carrier frequency the quantization error increases due to the corresponding signal components of the RF PWM. With each bit the quantization noise floor improves by roughly 6 dB. For very low resolution with

6 bit it can be seen that the quantization noise is already partially correlated to the signal, which leads to a small increase of the quantization noise around the inband signal.

In practice the achievable resolution is limited and can possibly be enhanced by using noise shaping. In Fig. 5.25 the block diagram of a RF PWM modulator including  $\Delta\Sigma$ noise shaping is depicted [258]. The digital BB signal is upsampled and aliasing effects caused by the Zero Order Hold (ZOH) nature of the modulator are compensated by the CPE block. The detailed function of the CPE block will be presented in section 5.2.3. After the CPE correction the signal is fed to the  $\Delta\Sigma$ , which works together with the RF PWM modulator block. The quantized signal from the RF PWM modulator is fed back to the  $\Delta\Sigma$ , which performs the noise shaping to increase the inband signal quality. In order to provide the desired noise shaping, the  $\Delta\Sigma$  has to be employed on the IQ data due to their orthogonal nature. Applying the  $\Delta\Sigma$  on phase and the amplitude instead of the IQ signal is not possible as due to the nonlinear relation of the RF signal to phase and amplitude the shaped noise folds back into the band.



Figure 5.25: Delta Sigma operation for resolution enhancement.

Fig. 5.26(b) depicts the quantization error for 6 to 9 bit resolution while using a second order  $\Delta\Sigma$ . By employing the noise shaping the inband signal quality around the carrier can be significantly improved and even with low resolution good results can be achieved. The  $\Delta\Sigma$  moves the quantization noise to the frequency band between DC and the carrier, respectively between the carrier and the second harmonic. The resolution of the modulator mainly determines the magnitude of the noise floor of the  $\Delta\Sigma$  operation.



Figure 5.26: Spectrum of RF PWM signal at 2.6 GHz with 40 MHz bandwidth for limited resolution without and with  $\Delta\Sigma$ .

## 5.2.2 Digital RF PWM

The focus of this section is on digital RF PWM and related aliasing problems [259]. Due to the sampled nature of the modulator, aliasing products from higher order harmonics

are generated back into the signal band [259–262]. This aliasing products impose a major limitation for signal quality.

For the analysis presented in this section the parallel modulator architecture of Fig, 5.22(b) will be considered. The IQ input signal is converted into phase  $\Phi$  and amplitude A. Subsequently the outphasing angle  $\Phi_A = \cos^{-1}(A)$  is calculated. The final control signals for the PMs consist of the phase  $\Phi$  of the signal and the outphasing angle  $\pm \Phi_A$ . These signals are fed to the PMs, which modulate the 50 % duty cycle LO signal. The actual RF PWM signal is generated by the logical AND combination of the two phase modulated signals. The modulator can only be updated once per carrier period and therefore implicitly performs a ZOH. To understand the effects of the ZOH on the signal quality it is best to start with the base band spectrum resulting from the phase modulation by the outphasing control signal  $\Phi_x$ , which is given by

$$\phi_x(f,n) = \mathcal{F}\left\{\operatorname{Re}\left\{\operatorname{e}^{jn\Phi_x(t)}\right\}\right\}$$
(5.1)

for the time continuous case. The parameter n denotes a multiplication factor for the higher order harmonics, which is 1 for the fundamental. To account for the sampled nature of the signal the sinc function has to be considered by

$$\phi_{x,ZOH}(f,n) = \operatorname{sinc}\left(f/f_s\right) \sum_{i=-\infty}^{\infty} \phi_x(f-if_s,n),$$
(5.2)

where  $f_s$  denotes the sampling frequency, which is the carrier frequency in this case.

The effects of this ZOH on the spectrum of the base band outphasing signal  $\phi_1$ , respectively  $\phi_2$  are depicted in Fig. 5.27(b). It shows the base band spectra of the phase modulator for the analog case ( $\phi_1$ ) and the sampled version ( $\phi_{1,ZOH}$ ) for a carrier frequency of 2.6 GHz. The ZOH operation results in an increase of the PSD at multiples of the signal rate (carrier frequency) due to the sinc function.



Figure 5.27: Illustration of aliasing of the higher order harmonics due to ZOH.

The signal for each outphasing branch is generated by modulating the phase of the digital LO signal (50% duty cycle) with  $\Phi_1$  or  $\Phi_2$  respectively and may be written as its inverse Fourier Series

$$sig_{x}(t) = \frac{1}{2} + \sum_{n=1}^{\infty} \operatorname{Re}\left\{\frac{2\sin\left(\frac{n\pi}{2}\right)}{n\pi} e^{jn(\omega t - \frac{\pi}{2})} e^{jn\Phi_{x}(t)}\right\}.$$
(5.3)

Its frequency domain representation is given by

$$sig_x(f) = \frac{1}{2}\phi_{x,ZOH}(f,1) + \sum^{n\neq 0} \frac{2\sin\left(\frac{\pi}{2}\right)}{n\pi} e^{-j\frac{n\pi}{2}}\phi_{x,ZOH}(f-nf_s,n)$$
(5.4)

It can be seen that not only the fundamental of the LO signal but also the higher order harmonics (3<sup>rd</sup>, 5<sup>th</sup>, 7<sup>th</sup>...) exist and are modulated. Due to the ZOH of the modulator and phase signal respectively, the contribution of the higher order harmonics folds back into the signal band lowering the signal quality. This effect is illustrated in Fig. 5.27(a), depicting the spectral contribution of the higher order harmonics for one outphasing signal sig<sub>x</sub> and the desired in band signal. The AND operation for the final RF PWM signal generation

$$pwm(t) = sig_1(t) \land sig_2(t) \tag{5.5}$$

of the digital signals is equivalent to a mixing process and results in a convolution

$$pwm(f) = sig_1(f) * sig_2(f) \tag{5.6}$$

of the spectra of the two signals  $sig_1$  and  $sig_2$ . This convolution results in a contribution of the aliasing products of the higher order harmonics to the inband aliasing effects and increases the aliasing error.

In the time domain the problem manifests itself as an error in the determination of the cross points of the signal. Fig. 5.28 shows an illustration of the error in the time domain. The topmost trace shows the analog ( $\Phi_x$ ) and the sampled version ( $\Phi_{x,ZOH}$ ) of the outphasing control signals. It can be seen that the sampled signal deviates from the desired (analog) signal at the signal transitions as the outphasing signal changes during the sample period. The resulting deviation of the outphasing signals sig<sub>x</sub> can be seen in the middle traces of Fig. 5.28. This leads to a difference of the desired (analog) and the generated PWM signal.



Figure 5.28: ZOH problem in the time domain.

The error caused by this effect is significantly limiting the signal quality of the generated signals, especially for wideband signals or signals with a carrier frequency offset (carrier aggregation).

#### 5.2.3 Cross Point Estimation

In this section the reduction of the signal error by means of cross point estimation will be described. While for conventional CPE [263–265] a linear interpolation and a constant sawtooth reference function are sufficient, the CPE for RF PWM requires a triangular reference function, as depicted in Fig. 5.29. To account for the phase  $\Phi$  the reference function has to be phase modulated in order to generate the desired modulation signal. The root cause for the error due to the ZOH when using a triangular reference function  $V_{REF}$  is visualized in Fig. 5.29(a). It can be seen that the analog amplitude signal  $A_{ANA}$  changes during one symbol and crosses the reference function at different times than the constant ZOH amplitude signal  $A_{ZOH}$ . This results in a different signal for the ZOH case and subsequently in reduced signal quality. The same applies for the phase signal  $\Phi$ , which is modulating the reference function.

The signal quality can be improved by linearly interpolating the amplitude outphasing signal  $\Phi_{A,CPE}$  between the samples and using it for the signal generation as depicted in Fig. 5.29(b) [263–265]. The resulting signal is very close to the ideal analog signal and sufficiently accurate to improve the signal quality.



Figure 5.29: CPE using (phase modulated) triangular reference function.

This interpolation can be simplified by performing the cross point estimation for the single branch outphasing signals individually, as depicted in Fig. 5.30, which shows the block diagram of the proposed architecture [S12, S13].

In a first step the signal is separated to get the modulator control signals  $\Phi_1$  and  $\Phi_2$ . The procedure is equal to the control signal generation for the parallel modulator architecture, with the exception that no quantization is involved. Then the signal transitions are detected using CPE, which requires knowledge about the following sample and therefore optional signal prediction. The output of the CPE is the correct normalized edge timing of the outphasing signals. Based on the normalized edge timing the RF PWM signal is calculated. The final signal generation depends on the calculated RF PWM signal and the used mode of operation. The detailed function of each block will be described in the following subsections.

#### 5.2.3.1 Single Branch Cross Point Estimation

For the linear interpolation of the signals, as depicted in Fig. 5.29(b), knowledge of the next sample is required. To get the information about the next sample, signal prediction techniques [266–270] can be used. But when one sample delay can be tolerated ideal signal prediction can be achieved by using the delayed sample for the calculation and considering the current sample as the predicted one. This simplifies the signal prediction block in Fig. 5.30 to a simple one tap delay.

Due to the addition of the outphasing angle  $\Phi_A$ ,  $\Phi_1$  determines the falling edges, while  $\Phi_2$  is responsible for the rising edges of the RF PWM signal after the AND gate. By using this property the computational effort can be reduced as only information about the rising edges of  $\Phi_1$  and the falling edges of  $\Phi_2$  is required.

For the calculation of the crossing points the time domain waveform of the modulator signals  $sig_1$  and  $sig_2$  are of importance. Their definition in (5.3) may be rewritten to

$$sig_x(t) = \mathrm{H}\left(\sin\left(\omega t + \Phi_x(t)\right)\right) = \mathrm{sign}\left(\sin\left(\Phi_{\mathrm{Cx}}\right)\right),\tag{5.7}$$



Figure 5.30: CPE block diagram.

with  $\Phi_{Cx}$  being the argument of the sin function. It is given by

$$\Phi_{\rm Cx} = \omega t + \Phi_x(t) \tag{5.8}$$

and takes  $\omega t$  and the interpolated phase modulation signal  $\Phi_x(t)$  into account. The function H(x) generates the digital signal and is defined as

$$H(\mathbf{x}) = \begin{cases} 0, & x < 0\\ 1, & x \ge 0 \end{cases} .$$
(5.9)

Considering (5.7) it can be concluded that signal transitions occur when  $\Phi_{Cx}$  is equal to  $\pi$  and multiples of it. Fig. 5.31 depicts an example for the outphasing signal and the resulting signal calculation. Based on the input signal  $\Phi_x$ , which is plotted for the interpolated  $\Phi_{x,CPE}$  and the ZOH  $\Phi_{x,ZOH}$  case, the argument  $\Phi_{Cx} = \omega t + \Phi_x$  of the sine is calculated. It can be seen that the negative edge of the resulting signal sig<sub>x</sub> occurs each time the argument  $\Phi_{Cx}$  crosses  $\pi$  or  $3\pi$ . While the positive edge occurs when  $\Phi_{Cx}$  crosses 0 or any cyclic repetition of it  $(2\pi, 4\pi...)$ .



Figure 5.31: CPE calculation for a single branch signal.

The contribution of the carrier  $\omega t$  is linearly increasing from 0 to  $2\pi$  within one period. Also the contribution of the interpolated input signal  $\Phi_{x,CPE}$  is linear. Using this property the cross points of the signal can be calculated by interpolation using the state of  $\Phi_{Cx}$  at the beginning and at the end of the sample, which can be calculated by

$$\Phi_{\mathrm{Cx},\mathrm{N}} = \Phi_{\mathrm{x},\mathrm{N}},\tag{5.10}$$

$$\Phi_{\rm Cx,N+1} = 2\pi + \Phi_{\rm x,N+1}.$$
(5.11)

During the calculation of  $\Phi_{Cx,N+1}$  the unwrapped signal of  $\Phi_{x,N+1}$  has to be considered to avoid any error due to the circular nature of the outphasing signals. Based on this

information the cross points (normalized edge timing) of the signals can be calculated using

$$t_{edge} = \frac{\Phi_{REF} - \Phi_{Cx,N}}{\Phi_{Cx,N+1} - \Phi_{Cx,N}},$$
(5.12)

where  $\Phi_{Cx,N}$  denotes the current and  $\Phi_{Cx,N+1}$  the upcoming (predicted) sample. The edge timing  $t_{edge}$  is normalized to one carrier period, such that the calculation is independent of the carrier frequency.

For the calculation of the rising edges the reference vector  $\Phi_{\text{REF}} = \{0, 2\pi, 4\pi\}$  is used. Only the resulting edge timings that are within the current period [0-1) are of importance and the others are ignored. Using this boundary conditions and the circular nature of the control signal  $\Phi_x$  [0-2 $\pi$ ) the calculation for the reference phase  $\Phi_{\text{REF}} = 0$  simplifies to a prompting of  $\Phi_{\text{Cx,N}}$  for zero. To determine the falling edges the reference vector  $\Phi_{\text{REF}} = \{\pi, 3\pi\}$  is used.

Taking into account that  $\Phi_1$  determines the rising and  $\Phi_2$  the falling edges mitigates the demand for the calculation of the RF PWM signal, as depicted in Fig. 5.30, as the edge timing calculated by the CPE directly defines the PWM signal. Only for the general case when it is not explicitly known which signal transitions of the outphasing signals sig<sub>x</sub> dominate in the signal generation the RF PWM calculation block is required.

It shall be noted that instead of performing the cross point estimation it is possible to perform cross point detection based on the upsampled time domain signals. As this requires a high computational effort and complexity, as the signal has to be oversampled with at least the timing resolution of the phase modulators, this option is not further considered.

For the implementation of the CPE algorithm the required division in (5.12) imposes a challenge and should preferably be avoided. Therefore, the calculation of the edge timings was modified to an iterative approximation approach. The corresponding block diagram is given in Fig. 5.32. In the first step the phase difference  $d\Phi_x$  between the current  $\Phi_{x,N}$  and the upcoming sample  $\Phi_{x,N+1}$  is calculated and normalized to give the edge timing difference dt. The reference phase is considered by subtracting it from the current sample  $\Phi_{x,N}$  and the resulting signal is normalized to give the initial estimate for the edge timing t<sub>1</sub>. The initial edge timing defines the first normalized error weight e<sub>t1</sub>. Based on e<sub>t1</sub> and the total timing difference of the entire sample period dt the error signal e<sub>tN</sub> is calculated and used to correct the edge timing t<sub>1</sub>. The error e<sub>tN</sub> defines the timing correction of the edge timing and furthermore it is the normalized correction value for the next iteration. In this way the interpolated timing can be calculated to a given accuracy defined by the number of steps required.



Figure 5.32: CPE iterative interpolation.

By using the iterative interpolation approach the demand for the division involved in the cross point calculation can be mitigated. In Appendix 7.10 detailed block diagrams of a

possible implementation of the CPE are provided.

### 5.2.3.2 Minimum Error Signal Generation

The last step in the CPE signal calculation is the calculation of the output control signals, respectively the according IQ symbols. Based on the desired RF PWM signal calculated in the previous step the output signals are determined. For modulated signals with rapidly varying phase or signals with a frequency offset (carrier aggregation) up to two RF PWM pulses per carrier period can be required. As the digital RF PWM modulator presented in Fig. 5.21 is only capable of generating one pulse per period special care has to be taken.

To account for this property the signal is separated into a main  $IQ_{main}$  and an auxiliary path  $IQ_{aux}$ . The separation between main and auxiliary path is based on the decision of the minimum error pulse. If only a single pulse is present it is directed to the main path and the auxiliary path remains zero. In Fig. 5.33 an example of this separation is provided, like during the first two periods where only a single pulse is present.



Figure 5.33: Example timing diagram for pulse separation for main and auxiliary path.

In the last period two pulses are present. In such a case the main path holds the pulse which is longer and has hence the smaller error compared to the shorter pulse. The second (missing) pulse is assigned to the auxiliary path. In the following subsections different methods how to handle the auxiliary signal and their impact on signal quality will be discussed.

## 5.2.3.3 Single and Double Branch Modulator Performance

If no CPE is performed the signal quality is limited by the aliasing effects, as described in section 5.2.2. This can be seen in trace  $e_{NO CPE}$  in Fig. 5.34, which shows the spectral error signal (generated signal - reference signal) for a carrier frequency of 2.6 GHz and 40 MHz bandwidth when no cross point estimation is performed. The error signal representation is chosen to provide also insight into the inband signal quality. It can be seen that the error reduces the signal quality to slightly above 43 dBc at the signal borders. At the carrier frequency the spectral error gets very small due to the sinc nature of the ZOH.

When using a single modulator (Fig. 5.21) controlled by the main CPE signal (IQ<sub>main</sub>), as described in section 5.2.3.2, the signal quality ( $e_{CPE}$ ) can already be improved to ~65 dBc in the example given in Fig. 5.34. The remaining noise floor is due to the fact that the possible occurrence of a second pulse within one carrier period (IQ<sub>aux</sub>) is not treated.

To further improve signal quality a second modulator capable of generating the required second pulses might be used. Fig. 5.35 shows the block diagram of such a modulator structure. The IQ output of the CPE depicted in Fig. 5.30 is fed to two RF PWM modulators (Fig. 5.2(b)) and combined by a logical OR. It shall be noted that the second pulse could also be generated by a single modulator, if the control signal is oversampled compared to the carrier frequency.



Figure 5.34: Reference and error signal spectra with and without CPE for a carrier frequency of 2.6 GHz, a bandwidth of 40 MHz and a resolution of 11 bit.



Figure 5.35: Block diagram of main and auxiliary path modulator.

By using this structure it is possible to generate the missing pulses, further decreasing the error signal. This can be seen in trace  $e_{CPE 2-branch}$ , which depicts the error signal for the combined main and auxiliary signal. The noise floor with roughly -75 dBc is 10 dB lower than the single modulator option ( $e_{CPE}$ ). The remaining noise is dominated by the quantization noise caused by the 11 bit resolution of the phase modulator. A drawback of this operation is that the RF PWM modulator components depicted in Fig. 5.21 have to be duplicated, which leads to increased hardware complexity and power consumption.

#### 5.2.3.4 Delta Sigma Cross Point Estimation Noise Shaping

Instead of using a second modulator branch, the remaining error can be spectrally shaped by means of delta sigma noise shaping [248–252]. A delta sigma modulator to shape the quantization noise might already be present in the system to increase the signal quality if limited resolution is available. Fig. 5.36(a) provides a block diagram of such a combined CPE  $\Delta\Sigma$  system that allows to also shape the CPE error. The core of the circuit is the delta sigma modulator, which consists of an adder and the loop filter. The input signal of the  $\Delta\Sigma$  modulator is the IQ<sub>main</sub> signal of the CPE. Its output signal IQ<sub>DS</sub> is fed to the RF PWM modulator, which quantizes and generates the signal. The  $\Delta\Sigma$  uses the difference between the desired signal (IQ<sub>main</sub>) and the quantized signal to shape the quantization error.

In addition to the quantization error a possible error due to the missing second pulse might be present. This error is the signal of the auxiliary path  $IQ_{aux}$ , which forms together with the quantized signal  $IQ_Q$  the error signal for the noise shaping. In order to process the CPE signal, which is calculated at carrier rate, and to provide the highest oversampling ratio and thus best noise shaping the delta sigma runs also at carrier rate.



(a)  $\Delta\Sigma$  with auxiliary signal as error feedback



(b) Additive CPE noise shaping with optional  $\Delta\Sigma$ 

Figure 5.36: Block diagram for PWM modulator with CPE error noise shaping.

As the error due to the missing pulse is a priori known, no delay is required and it can be directly considered in the feedback path. This makes it possible to move the addition for the feedback signal, as given in Fig. 5.36(a), outside of the  $\Delta\Sigma$  loop, as depicted in Fig. 5.36(b). By doing so the  $\Delta\Sigma$  is only required to shape the quantization noise and therefore the  $\Delta\Sigma$  itself is optional in the CPE context. The spectral shaping of the error due to the second pulse results in a simple implementation, as only an addition of the two signal paths (main and aux) is required.

In Fig. 5.37 an example of the additive CPE operation is given. During the first two periods only one pulse occurs, consequently the output  $pwm_{ADD}$  is the same as the desired signal  $pwm_{CPE}$ . But in the third period two pulses occur. These two pulses are split into the modulator paths and the corresponding IQ representation is calculated and added (IQ<sub>ADD</sub>). This can be seen as a calculation of the corresponding IQ vector of the fundamental signal for the given signal period. The result is that the modulator selects the single pulse signal, whose IQ vector shows the smallest quantization error with respect to IQ<sub>ADD</sub>. Such an example is depicted in the third period in Fig. 5.37.



Figure 5.37: Example timing diagram for pulse generation for delta sigma CPE noise shaping.

The resulting spectral improvement can be seen in trace  $e_{CPE ADD}$  in Fig. 5.34. The inband signal quality reaches the level of the two branch system and converges towards the single branch noise floor when deviating from the carrier frequency. It can be concluded that by the simple addition of the two branch signals it is possible to shape the error and to mitigate the demand for a second modulator branch or the  $\Delta\Sigma$ . The  $\Delta\Sigma$  itself can be optionally included to shape the quantization noise but is not required for the CPE correction. Thus by using the additive CPE correction best inband performance without additional hardware and power consumption in the modulator can be achieved.

# 5.2.3.5 Measurement Results

To evaluate the proposed signal generation, measurements using an Arbitrary Waveform Generator (AWG) have been carried out. The digital output of an AWG with a maximum sample frequency of 1.2 GHz was used to generate the digital RF PWM signal. In order to emulate a high resolution of 10 bit with the available measurement equipment a carrier frequency of 1 MHz was selected to provide the required oversampling. A Vector Signal Analyzer (VSA) has been used to measure the signal.

The measured spectra for 20 kHz bandwidth, which corresponds roughly to the 40 MHz at 2.6 GHz carrier frequency, are presented in Fig. 5.38. When performing no CPE the signal quality is limited to roughly 40 dBc. The main signal error can be removed by using the single branch CPE calculation, as depicted in trace CPE in Fig. 5.38. By considering the 2 modulator branches (CPE<sub>2-branch</sub>) the signal quality can be further improved by 5 dB. The signal is now dominated by the quantization noise and the clock jitter.



Figure 5.38: Measured signal at  $1\,\mathrm{MHz}$  carrier frequency,  $20\,\mathrm{kHz}$  bandwidth and  $10\,\mathrm{bit}$  resolution.

Additionally the additive noise shaping for the single branch modulator has been measured and is given in trace  $CPE_{ADD}$ , which shows that the inband signal quality can be improved. The effects of the aliasing error are even more pronounced, if the signal has a frequency offset (carrier aggregation). This is due to the fact that the aliasing components do not fall into the zero of the ZOH sinc anymore and significantly impact the signal quality. Trace 'no CPE' in Fig. 5.39(a) shows these effects for a 20 KHz bandwidth with a 40 KHz frequency offset. By using the additive noise shaping ( $CPE_{ADD}$ ) the signal quality can already be improved. The out of band spectra is dominated by the noise shaping, as a second pulse per carrier period frequently occurs, resulting in a significant error that has to be corrected.

But when considering a second modulator branch to generate the missing pulses ( $CPE_{2-branch}$  in Fig. 5.39(a)) optimum signal quality can be achieved.

The resulting signal quality for the two branch option with and without frequency offset is the same, showing the capabilities of the CPE and indicating smooth operation. Thus to achieve best signal quality also for signals with frequency offset it is required to use a second modulator branch. When using a single modulator and the additive CPE option to get best signal quality (CPE<sub>ADD</sub>) the out of band spectra for a frequency offset is significantly increased compared to the quantization noise floor dominated signal (CPE<sub>2-branch</sub>).



Figure 5.39: Measured signal at 1 MHz carrier frequency, 20 kHz bandwidth and 40 kHz frequency offset (carrier aggregation).

Considering this limitation, it is not required to have a high resolution of 10 bit, as otherwise a similar spectrum can be achieved by noise shaping and a resolution of 7 bit. This is depicted in Fig. 5.39(b), which considers a frequency offset of 40 KHz, 7 bit resolution and a 2<sup>nd</sup> order delta sigma architecture as given in Fig. 5.36(b).

When comparing trace  $CPE_{ADD}$  for the 10 bit resolution in Fig. 5.39(a) and the 7 bit resolution and noise shaping in Fig. 5.39(b) only a minor increase of the out of band spectra can be achieved. The main contributor to the error spectrum is not the shaped quantization error, but the limited signal quality for single pulses as can be seen in trace CPE of Fig. 5.39(b). This means that for this case a resolution of 7 bit in combination with delta sigma is sufficient to achieve almost optimum signal quality. Further enhancements in resolution only marginally improve the out of band spectrum.

## 5.2.3.6 Summary CPE

Aliasing problems and related signal degradation in digital RF PWM modulators have been described. The root cause has been identified to be related to an error in the calculations of the cross points of the signal. The straight forward approach of calculating the cross points of the amplitude signal with a phase modulated reference function would result in a rather complicated calculation. The proposed method is reduced to two simple linear interpolations based on an outphasing method to generate the desired RF PWM signal. It shall be noted that the compensation is not limited to a specific architecture of the final modulator. To cover also the possible occurrence of two pulses per carrier period several methods have been investigated. With the proposed methods the aliasing problem can be significantly reduced and high signal quality can be ensured.

#### 5.2.4 Driver Properties

In this section the properties of the AND gate and the following driver stages are analyzed. In the case of ideal RF PWM generation infinite rise and fall times are required, which cannot be generated by real world driver circuits. In Fig. 5.40 a high level block diagram of a driver chain is depicted. Each of the driver blocks may be implemented as a complementary inverter circuit loaded with its own parasitic output capacitance and the parasitic input capacitance of the subsequent stages.

The rise and fall times of the circuit are determined by the parasitic capacitances and the maximum current that can be provided by the driver circuit. As a result of the



Figure 5.40: Driver block diagram.

finite rise and fall time the real driver response differs from the ideal one. Fig. 5.41(a) depicts the fundamental amplitudes at different stages of the driver chain. The curve can basically be divided into 3 regions. For very low duty cycle signals Pulse Swallowing (PS) can happen, which occurs when the output of one stage is too small to trigger the threshold of the subsequent stage [237]. For the intermediate duty cycle region the effects of pulse shortening/extension caused by different rise and fall times dominates the driver characteristic. The fundamental amplitude for high duty cycles around 50 % for the driver circuit is reduced compared to theory due the impact of limited rise and fall times.



Figure 5.41: Driver errors and AM/AM, AM/PM deviation.

The corresponding AM/AM and AM/PM characteristics for the output  $V_2$  of the second driver stage is plotted in Fig. 5.41(b). It can be seen that for the pulse swallowing region for low duty cycles the highest deviation for the AM/AM characteristic occurs. Also the phase response drifts away in the pulse swallowing region, as can be seen in the AM/PM curve. Above the pulse swallowing region the phase remains unaffected. In the intermediate and high duty cycle region the pulse shortening/extension effect, respectively the finite rise and fall time, dominate the behavior. It shall be noted that for illustrative purposes a worst case scenario is assumed to highlight the impact of the individual effects.

In Fig. 5.42(a) a time domain illustration of the pulse shortening effect is provided. In general the rising edge is slower than the falling edge, as the high side PMOS transistors have lower current density than the low side NMOS transistors. This leads to a reduction of pulse length, as the delay for triggering the next stage is higher for the rising edge than the (faster) falling edge. Ideally this effect is compensated in the next stage, whose input is the inverted signal which experiences effectively a pulse extension. This effect causes the increase of the fundamental voltage in the intermediate duty cycle region for all odd numbered signals in Fig 5.41(a).

Compared to the pulse shortening/extension effect the pulse swallowing effect has a more severe impact. Fig. 5.42(b) provides an illustration in the time domain of this effect. The pulse in the first period is sufficiently long, such that the threshold of the next stage is



Figure 5.42: Driver effects.

reached. But the effective pulse does not reach a steady state, which results in a reduced output pulse length and height. In the second period the input pulse is very short, such that the output pulse of the first stage is not capable of triggering the next stage. Thus the pulse disappears and is 'swallowed' in the driver chain. The higher the number of elements in the chain, the more pronounced is the effect, leading to a rapid transition as depicted in Fig. 5.41(a).

The PS effect is not limited to the static case but can also appear for dynamic signals with sufficiently long duty cycles. Due to the circular nature of the modulator a pulse can be split into two parts, as in the second period in the example in Fig. 5.43(a). As a result a part of the pulse is too short and swallowed. This is referred to as dynamic PS.

In Fig. 5.43(b) the pulse swallowing regions in the IQ plane of the modulator are depicted. For low duty cycles static PS occurs, independent of the phase. As a result the inner circle of the IQ plane (except zero) cannot be represented.



Figure 5.43: Static and dynamic pulse swallowing.

The region where dynamic PS can occur is also indicated. It is separated into two main areas, as dynamic PS can either happen at the beginning or the end of the period. For slowly varying signals dynamic PS is no problem as the pulses are extended by the adjacent samples. Thus dynamic PS can only occur for rapidly varying phase signals. In general the CPE will correct this signals accordingly, such that dynamic PS is avoided. But due to the additive error shaping of the CPE dynamic PS cannot be completely excluded.

### 5.2.5 Predistortion

In this section the required predistortion to compensate for the driver nonlinearity will be presented. The nonlinear effects of the driver lead to significant distortion that imposes a major limitation for the spectral quality [271]. In order to compensate these effects a predistortion scheme specifically targeted to the digital RF PWM modulator has to be applied. In Fig. 5.44 the corresponding block diagram is provided. In the first step the edge timing for the input IQ samples is calculated. This is followed by the signal prediction block, as it is required to know the upcoming sample. As for the CPE implementation this block can consist of a simple one tap delay. In the next step the desired pulses, based on the edge timing of the previous  $t_{N-1}$ , current  $t_N$  and upcoming sample  $t_{N+1}$ , are calculated. The desired pulses are then corrected individually by static predistortion using Look Up Tables (LUTs) for phase and amplitude correction. Based on the predistorted pulses, which fall into the current sample, the control signal is generated. Up to two pulses per sample can occur, which are separately converted into the IQ domain. Finally the IQ values are added to give the final predistortion signal.



Figure 5.44: Carry over based predistortion block diagram.

For very wideband signals one sample varies significantly from the other and a predistortion that works only on a per sample basis cannot fully correct the signal anymore. In Fig. 5.45 examples highlighting the difference between sample and pulse based predistortion are provided. By applying the predistortion on a per sample basis the relation of the pulses with the adjacent samples is ignored. For the first period in Fig. 5.45(a) the pulse occurs in the middle of the sampling period and thus has no influence on the other samples after predistortion. In this case the sample based predistortion delivers the correct results. But during the second period the predistorted signal results in a circular shifted pulse with only a very short pulse at the beginning of the sample period. After the driver this pulse disappears due to the PS effect. This leads to an error. Also in case of a pulse covering two adjacent sampling periods, such as in the third period the correction applied on a per sample basis delivers a wrong result, as the driver nonlinearity acts on a per pulse basis.



Figure 5.45: Sample and pulse based predistortion carry over/back effects.

In order to compensate for the behavior of the driver circuit the predistortion has to be applied on a per pulse basis, as depicted in Fig. 5.45(b). As a result, the nonlinearity of the driver can be perfectly compensated, with the drawback of introducing some special effects. After predistortion the pulse might be extended to reach into the next sample period (carry over) or it might be expanded to the previous sample period (carry back). In both cases interaction with the adjacent samples is given and therefore it is required to take the previous and the upcoming sample, Carry Over (CO) effect, into account for the calculation of the predistortion signal.

Due to CO effects it is possible that three signal edges within one sample period occur. Fig. 5.46(a) provides such an example. The pulse from the first period is extended to the second period, while the second pulse still completely remains within the second period after predistortion. As a result three signal transitions for the second carrier period are desired. But the modulator can only provide a single (circular shifted) pulse and thus cannot generated the desired signal. One option could be to ignore the CO contribution, as depicted in Fig. 5.46(a). As a result the pulses after the driver shows some error in the first period. By simply skipping the CO contribution a single sided error is generated, which is correlated to the signal and leads to raise of the inband noise floor.



Figure 5.46: Pulse predistortion time domain three signal transitions.

Instead of skipping the contribution of the CO pulse it can be considered as an additive manner. Fig. 5.46(b) provides a corresponding example. The IQ values of the predistorted pulses in the second period are calculated and added. The result is a pulse which represents the closest sample in the IQ domain. In this example the resulting pulse in the second period is slightly shifted compared to the CO skipping approach in Fig. 5.46(a) and exhibits circular properties such that it can be represented by the modulator. By using this additive approach the error is distributed among the two pulses involved and does not have a single side preference. The resulting error spectrum exhibits a  $\Delta\Sigma$  like behavior, as the error signal energy is shifted out of the band. This way high inband signal quality can be maintained and thus the additive approach is included in the proposed predistortion scheme in Fig. 5.44. By calculating the IQ values of the pulses individually the contribution of the adjacent pulses, as well as the additive handling of the three signal transitions, are implicitly considered.

Another problem that can occur for very rapidly varying signals is the overlap of signals after predistortion. In Fig. 5.47(a) such an example is depicted. After predistortion the signal transitions of the first and the second sample are in very close vicinity. As a result the negative pulse between the two signals cannot be generated due to PS, which leads to the fact that the generated pulse by the driver is too short. If the desired pulses are even closer, as for the transition of the second to the third sample, the required predistortion signal can overlap. This leads to an error during the signal calculation and results in a

pulse after the modulator, that is too long. In both cases involved the final signal after the modulator is erroneous, which leads to a degradation of the signal quality.



Figure 5.47: Pulse predistortion circular overlap problem.

The problem can be resolved by detecting and handling such a situation. Instead of treating the pulses individually they are 'added' in the IQ domain to give a single pulse that closely represents the signal content of the two pulses. Fig. 5.47(b) provides a corresponding example. The pulses for which the overlap problem would occur are combined in an additive way and predistorted. The result are well controlled pulses of the modulator, as well as after the driver. Still a residual error exists, which experiences  $\Delta\Sigma$  like noise shaping due to the addition of the pulses. Nevertheless this situation occurs rarely as it is only caused by very rapidly varying phase signals, which are corrected by the CPE anyways.

In Fig. 5.48(a) the spectra of the CO based predistortion for a modulator with 10 bit resolution are plotted. It depicts the reference signal (ref) and the error signal ( $e_{\rm NL}$ ) caused by the nonlinearity, if no predistortion is applied. By using the sample based approach for predistortion, the signal quality  $e_{\rm pre,SMPL}$  can be improved, but is still limited. This is caused by errors due to the CO effects, which are neglected. Applying the proposed CO based predistortion improves the signal quality  $e_{\rm pre,CO}$  significantly, coming close to the noise floor of an ideal driver  $e_{\rm ideal}$ . The difference between the ideal and the CO based predistortion spectrum is caused by two effects. Due to the nonlinearity the quantization steps are slightly shifted. This leads to an increase of the noise floor. This can be seen in trace  $e_{\rm pre,best}$ , which depicts the error signal of the predistortion applied on a per pulse basis, ignoring the limitations of the driver by applying the IQ addition of the single pulses within one sample period results in the small increase of the out of band spectrum  $e_{\rm pre,CO}$  for the CO based predistortion. Nevertheless high inband signal quality can be achieved.

For reduced modulator resolution, as depicted in Fig. 5.48(b), the spectrum after predistortion  $e_{pre,CO}$  shows still an increased noise floor around the signal band. This is caused by the modified quantization of the nonlinearity and is not a residual error of the predistortion. Fig. 5.49 provides an example for the influence of the driver nonlinearity on the amplitude quantization. For ideal driver characteristic, as plotted in Fig. 5.49(a), the resulting quantization steps are equally distributed.

The driver nonlinearity in the pulse swallowing region, such as in Fig. 5.49(b), leads to increased quantization steps due to the step amplitude response with respect to the duty cycle. In the most extreme case for hard pulse swallowing only a single quantization step in the pulse swallowing region is available. The resulting coarse quantizations lead to an increase of the effective amplitude resolution, as well as to a partial correlation with the signal. Therefore, the inband signal quality cannot completely be restored by the predistortion.



Figure 5.48: Spectra for a 40 MHz bandwidth signal at 2.6 GHz with driver nonlinearity corrected by different predistortion schemes.



Figure 5.49: Increased quantization step due to nonlinearity.

A method to decorrelate and shape the quantization error is the use of a  $\Delta\Sigma$  modulator. The predistorter can either be located before the  $\Delta\Sigma$ , as depicted in Fig. 5.50(a), or the predistorter can be integrated in the  $\Delta\Sigma$  loop (Fig. 5.50(b)).

Both predistorter locations have their advantages and drawbacks. When including the predistortion in the  $\Delta\Sigma$  loop, as depicted in Fig. 5.50(b), the correct calculation of the CO based predistortion is not possible. Due to the noise shaping of the  $\Delta\Sigma$  the next sample cannot be predicted and therefore the CO based predistortion cannot be calculated properly. Therefore, only the sample based predistortion approach can be applied within the  $\Delta\Sigma$  loop. As a result the overall signal quality is determined by the limitations of the sample based predistortion approach.

Locating the predistorter before the  $\Delta\Sigma$ , as done in Fig. 5.50(a), allows for the calculation of the CO based predistortion. A drawback of this combination is that only the inband signal is considered in the predistortion. As a result the out of band quantization noise shifted by the  $\Delta\Sigma$  folds back into the signal band due to the driver nonlinearity. This effect can be seen in Fig. 5.51, which depicts the predistortion results for a 7 bit modulator with and without  $\Delta\Sigma$  operation. For normal quantization the nonlinearity error  $e_{\rm NL}$  can be improved to  $e_{\rm pre}$  by applying the CO based predistortion. The inband spectral quality is still limited by the resolution and is further increased by the partial correlation of the quantization noise. By applying a 1<sup>st</sup> order  $\Delta\Sigma$  modulator after the predistortion the spectral quality can be improved to  $e_{\rm pre, DS}$ . The achievable resolution enhancement is



Figure 5.50: Predistorter location.

limited compared to the ideal case  $(e_{ideal,DS})$ , as a part of the out of band quantization noise is folded back into the signal band due to the nonlinear driver behavior.



Figure 5.51: Spectra for a 40 MHz bandwidth signal at 2.6 GHz with driver nonlinearity corrected by pre  $\Delta\Sigma$  predistortion using 1<sup>st</sup> order  $\Delta\Sigma$ .

These effects impose a limit for the signal quality of this combined predistortion/noise shaping approach. Although the quantization noise that is folded back into the signal band imposes a limitation, it was found that this approach delivers better results than using the sample based predistorter within the  $\Delta\Sigma$  loop.

#### 5.2.6 Predistortion Measurement Results

In order to verify the proposed predistortion scheme a driver model has been built. The same measurement equipment as in section 5.2.3.5 has been used, limiting the operating frequency to 1 MHz in order to provide sufficient resolution. The driver was built using several discrete inverters [272] soldered onto a prototyping board, which were loaded with a shunt capacitance to decrease their rise/fall times to emulate the expected behavior at 2.65 GHz. Fig. 5.52(a) shows a picture of the measurement setup including the driver model. The measured AM/AM and AM/PM distortion of the driver after two inverters is depicted in Fig. 5.52(b).



Figure 5.52: Measurement setup and driver distortion.

The statically measured distortion behavior is used for the predistortion of the modulated signals after their CPE correction. The signal processing chain provided in Fig. 5.50(a) was used for the measurements. A signal bandwidth of 20 kHz at the carrier frequency of 1 MHz was selected, which roughly corresponds to 40 MHz bandwidth at 2.65 GHz. The corresponding measurement results considering a resolution of 10 bit without noise shaping are provided in Fig. 5.53. It can be seen that without applying any predistortion (no pred) the signal is highly distorted and the spectral quality is rather poor. By applying the sample based predistortion (smpl based) the signal quality can already be improved, but still suffers from errors due to carry over effects. The best results can be achieved by using the proposed carry over based predistortion, as can be seen in trace 'CO based'.



Figure 5.53: Measured signal at the driver output at 1 MHz carrier frequency, 20 kHz bandwidth and 10 bit resolution.

In practice a resolution of around 7 bit is common for carrier frequencies around 2.65 GHz. Therefore, the same measurement using a resolution of 7 bit, while applying  $2^{nd}$  order  $\Delta\Sigma$  noise shaping, has been repeated. Fig. 5.54(a) provides the corresponding results. It can be seen that by using the carry over based predistortion approach the signal quality can be significantly increased, but the inband noise floor is limited by the out of band quantization noise that is folded back into the band by the nonlinearity.

In order to support multi band and multi standard operation the capability of generating signals with a frequency offset is a very important functionality. Therefore, a measurement with 40 kHz carrier offset, as depicted in Fig. 5.54(b), has been carried out. It can be seen that the signal quality almost reaches the values without a frequency offset, offering significant improvement by using the carry over based predistortion.



Figure 5.54: Measured signal at the driver output at 1 MHz carrier frequency, 20 kHz bandwidth and 40 kHz frequency offset (carrier aggregation) and 7 bit resolution.

#### 5.2.7 Digital Mismatch Compensation

Beside the driver nonlinearity the device mismatch of the phase modulators can have a significant impact on the overall performance. Due to accuracy limits within the manufacturing process the properties of the single devices deviate from each other. This leads to a deviation between the real phase response of the modulator and the theoretical linear one. The impact of the device mismatch can be reduced by adding hardware tuning elements or by digitally compensating the phase modulator behavior in the signal generation. Fig. 5.55 depicts a block diagram for digital mismatch compensation. In order to correct the mismatch it has to be identified by measuring the phase response of the modulator. This phase response delivers the PM/PM characteristic for each branch. Based on the inverse PM/PM characteristic the outphasing signals for each branch are corrected prior to quantization. The quantization step is followed by an optional Look Up Table (LUT) for scrambling to account for non monotonic phase responses of the modulator.



Figure 5.55:  $\Delta\Sigma$  and digital mismatch compensation.

The  $\Delta\Sigma$  modulator is optional and can be used in order to increase the inband signal quality. In order to provide the correct feedback signal it is required to consider the real modulator behavior by applying the identified PM/PM characteristic to each branch prior to the calculation of the quantized IQ<sub>Q</sub> value.

The required LUT size and hence measurement effort for the parallel architecture presented in section 5.2.1 is  $2 \times N$ , where N denotes the phase resolution. As the impact of the mismatch of the single branches is independent from each other the mismatch of each output combination  $(N \times N)$  can be calculated from the  $2 \times N$  data. For the series architecture the influence of the modulator's mismatch is coupled and therefore it is required to identify the full  $N \times N$  matrix for compensation. Aside from a significant increase in measurement effort, the implementation complexity increases. Thus the series architecture has significant drawbacks in terms of mismatch compensation.

In Fig. 5.56(a) an example for the digital mismatch compensation of a parallel modulator with 7 bit resolution combined with a 1<sup>st</sup> order  $\Delta\Sigma$  is provided. It can be seen that without compensation the noise floor around the inband signal is significantly increased. The mismatch leads to a non uniform quantization and therefore the quantization noise if partially correlated with the signal. This leads to the rise around the signal band. As in this region high signal quality is most important, mismatch compensation is required. By applying the digital mismatch compensation scheme proposed in Fig. 5.55, high signal quality can be recovered.



Figure 5.56: Mismatch compensation and measurement sensitivity for 1<sup>st</sup> order  $\Delta\Sigma$ .

A main factor for the mismatch compensation is the accuracy of the mismatch identification. Measuring the response of the phase modulator is only possible up to a certain accuracy when using measurement instruments. The resulting measurement error directly translates to reduced signal quality as the signal in the feedback path, used for noise shaping, and the actually generated signal deviate. This limits the effectiveness of the digital mismatch compensation. Fig. 5.56(b) depicts the sensitivity of the Error Vector Magnitude (EVM), which is defined here as the average inband noise floor, to the measurement accuracy. The EVM directly depends on the measurement accuracy given by the Root Mean Square (RMS) measurement error with respect to the full phase modulator resolution. Thus for highest signal quality it is required to measure the modulator mismatch with an accuracy in the sub picosecond range, which imposes a great challenge for the measurement instrumentation. Another important factor is that the mismatch remains constant over time, temperature or voltage variations. This could be ensured by using a locking mechanism, such as the one proposed in Fig. 5.23(a).

# 5.3 Summary RF PWM

In this chapter the general difference between BB PWM and RF PWM operation and the advantages of purely digital RF PWM signal generation have been highlighted. The responses of different amplifier classes to RF PWM operation have been investigated. It has been found that for all the amplifiers, except the inverter based voltage mode class D, no significant efficiency enhancement in back off is possible by pure RF PWM operation. Moreover, the relation of the output voltage to the duty cycle is highly nonlinear for most amplifier classes.

Nevertheless the concept of RF PWM is well suited for providing an efficient modulator implementation with a single bit digital output. In order to achieve highest signal quality, analog and digital effects have to be compensated. Therefore, the second part of this chapter was dedicated to the required signal processing for best spectral quality. Aliasing effects due to the discrete time nature of the digital modulator have been identified and an approach for their mitigation has been developed. The properties of the different variants of the developed compensation scheme based on Cross Point Estimation (CPE) have been evaluated in detail. In addition the properties of driver circuits for the RF PWM signal generations have been studied. The nonlinear behavior of real driver implementations results in reduced signal quality. In order to remove the driver's influence a special predistortion scheme for wideband signals has been developed. By performing the proposed predistortion on a per pulse and not on a per sample basis best performance can be achieved. Finally a concept for the digital compensation of the (device) mismatch of the phase modulators was derived and evaluated.

A big drawback of the signal processing schemes developed is their requirement for running at carrier rate. This imposes a big challenge for the implementation of the signal processing circuitry for transmitters operating in the GHz range. Nevertheless the impact of the specific errors has been highlighted and the demand for their compensation has been shown.

# 6. Conclusion

In the first chapter the general operational principle of PAs has been discussed and a new circuit variant for the class DE PA based on a parallel compensation inductance [S1] has been proposed. Additionally the requirements on the static efficiency curve for achieving good average efficiency have been derived. Based on a simplified circuit model the different loss mechanisms of RF power transistors and their influence on the efficiency of load modulation have been discussed [S2].

The second chapter deals with different efficiency enhancement concepts, where a main focus was set on BB PWM operation. The properties of BB PWM operation using an isolated load and the prospects of direct filter connection have been highlighted. Beside that different energy recovery methods have been reviewed and a concept for direct reuse of the out of band energy of BB PWM transmitters [S3] has been introduced. With that concept theoretical 100% efficiency over the whole output power range can be achieved.

A main chapter is dedicated to the design of PAs suitable for the use of BB PWM and direct filter connection. At first the concept of band limited BB PWM to avoid the aliasing problems of digital BB PWM is introduced. For the design of direct filter connection the resulting load modulation trajectories are very important. Therefore, generalized equations [S4] to determine the load modulation trajectories for BB PWM with arbitrary levels have been derived. For the efficient operation of the direct filter connection it is required to provide a high impedance over a wide frequency range for the modulation sidebands [S7]. Therefore, a study of the impact of the coupling on the reflections of cavity filters [S5] has been carried out. The results provide the starting point for the matching network design, which is done as a codesign of microstrips on PCB level and the cavity filter. For best performance it is required to have short connection between filter and PA within the matching network. Four different integrated circuits for direct filter connection have been designed. Two of them are based on a parallel approach, either on a common drain [S6] or a common source design [S8]. The other two PA designs follow a series approach by adding a switch to a cascode [S9, S10] and by source modulation [S11]. Amongst them the source modulation circuit shows the best efficiency performance, while requiring only one RF and one digital input signal.

The second main chapter is dedicated to RF PWM, in that context the PA properties under RF PWM excitation are discussed, but the main focus is set on the required signal processing for highest signal quality. In that context a method for the reduction of aliasing effects of digital RF PWM modulators based on Cross Point Estimation (CPE) has been developed [S12, S13]. Additionally a special predistortion scheme dedicated to the properties of RF PWM has been developed to account for the driver nonlinearity. Finally a digital mismatch compensation scheme that can optionally be combined with  $\Delta\Sigma$ modulation, has been introduced.

The concept of BB PWM could possibly be applied to smaller base stations by properly integrating existing BPFs into the matching network and using multilevel MMIC circuits. This includes the use of higher order filter networks and the optimization of the integration into the matching network, with the aim to provide a high impedance for the modulation sidebands over a very wide bandwidth. Further investigations into the digital predistortion of BB PWM signal based on the presented suggestions could result in improved spectral quality.

The presented RF PWM modulator concept could be used to replace conventional RF upconverters in small transceivers, where the power consumption of the DACs and mixers is critical for the system performance. Another benefit is the flexibility and reconfigura-

bility of such a modulator providing a step towards a fully digital transmitter. In order to achieve the best performance of the RF PWM modulator, accounting for the nonidealities of a real modulator, the proposed compensation schemes have to be implemented in hardware together with the RF PWM modulator core, providing a device that features direct base band to RF conversion.

# 7. Appendix

# 7.1 Generic Linear PA

In this section the efficiency of a general linear amplifier (class A to class C) as depicted in Fig. 7.1 is derived.



Figure 7.1: General class A, B and C schematic.

Its drain current is defined by

$$I_D(\theta) = \begin{cases} I_{D,MAX} \left( (1-k) \ m \sin(\theta) + k \right) & -\alpha \le \theta \le \pi + \alpha \\ 0 & \text{else} \end{cases} ,$$
(7.1)

where  $I_{D,MAX}$  denotes the maximum drain current of the transistor, k is the normalized bias current and m is the normalized current amplitude, respectively normalized gate voltage. In order to generate only a positive drain current the (integration) border  $\alpha$  is given by

$$\alpha = \min\left(\sin^{-1}\left(\frac{k}{(1-k)m}\right); \frac{\pi}{2}\right).$$
(7.2)

The restriction to a maximum of  $\pi/2$  is required to also cover the case that the load drain current is never zero (eg. class A operation).

The drain voltage is given by

$$V_D(\theta) = V_{DC} \left( 1 - V \sin\left(\theta\right) \cos\left(\phi\right) - V \cos\left(\theta\right) \sin\left(\phi\right) \right), \tag{7.3}$$

where  $V_{DC}$  denotes the drain bias voltage, V the normalized drain, voltage swing and  $\phi$  the phase angle of the load in case it is not purely resistive. As all higher order harmonics of the drain voltage are short circuited, only the fundamental of the drain current is of importance. It can be calculated by performing a Fourier analysis of the time domain signal (7.1) and yields

$$I_L = \frac{1}{\pi} \int_{-\alpha}^{\pi+\alpha} I_D(\theta) \sin(\theta) d\theta = \frac{I_{D,MAX}}{\pi} \left( (1-k) \frac{m}{2} \left( \pi + 2\alpha - \sin(2\alpha) \right) + 2k \cos(\alpha) \right).$$
(7.4)

The resulting power dissipation in the transistor depends on the average drain voltage and current overlap and is given as

$$P_T = \frac{1}{2\pi} \int_{-\alpha}^{\pi+\alpha} I_D(\theta) V_D(\theta) d\theta, \qquad (7.5)$$

which yields

$$P_{T} = \frac{I_{D,MAX}V_{DC}}{2\pi} \left[ \left( (1-k)m - kV\cos(\phi) \right) 2\cos(\alpha) + k(\pi + 2\alpha) - V(1-k)\frac{m}{2}\cos(\phi)(\pi + 2\alpha - \sin(2\alpha)) \right].$$
(7.6)

The output power delivered to the load can be calculated by

$$P_L = \frac{I_L V_L \cos(\phi)}{2},\tag{7.7}$$

where the load voltage  $V_L$  is given by

$$V_L = |I_L Z_L| = V_{DC} V. (7.8)$$

The drain efficiency is given by

$$\eta_D = \frac{P_L}{P_L + P_T} \tag{7.9}$$

and depends on the output power and the power dissipated in the transistor. Considering class A operation  $(k = 0.5, \alpha = \pi/2)$  the load current  $I_L$  can be simplified to

$$I_{L,A} = \frac{I_{D,MAX}m}{2}.$$
 (7.10)

Assuming a purely resistive load ( $\phi = 0$ ) allowing for maximum voltage swing for the maximum drain current (m = V) the power delivered to the load can be rewritten as

$$P_{L,A} = I_{D,MAX} V_{DC} \frac{V^2}{4}.$$
 (7.11)

The maximum output power (V = 1) with respect to the maximum drain voltage  $(V_{D,MAX} = 2V_{DC})$  is given by

$$P_{L,A} = \frac{I_{D,MAX} V_{D,MAX}}{8}.$$
(7.12)

The power dissipated in the transistor can be calculated by

$$P_{T,A} = I_{D,MAX} V_{DC} \left(\frac{1}{2} - \frac{V^2}{4}\right).$$
(7.13)

Furthermore the efficiency for class A operation can be calculated by (7.9) taking (7.11) and (7.13) into account. This delivers

$$\eta_A = \frac{I_{D,MAX} V_{DC} \frac{V^2}{4}}{I_{D,MAX} V_{DC} \frac{V^2}{4} + I_{D,MAX} V_{DC} \left(\frac{1}{2} - \frac{V^2}{4}\right)} = \frac{V^2}{2}$$
(7.14)

with respect to the normalized load voltage V.

Performing the same calculations for class B operation ( $k = 0, \alpha = 0$ ) and a purely resistive load ( $\phi = 0$ ), the load current  $I_L$  evaluates to

$$I_{L,B} = \frac{I_{D,MAX}m}{2},$$
(7.15)

showing linear behavior. The result is the same as for class A operation and thus the power delivered to the load is also the same as for class A operation (7.13).

The power dissipation in the transistor for class B operation is given by

$$P_{T,B} = I_{D,MAX} V_{DC} \left(\frac{V}{\pi} - \frac{V^2}{4}\right).$$
(7.16)

Thus the efficiency for class B operation with respect to normalized load voltage V is given by

$$\eta_B = \frac{I_{D,MAX} V_{DC} \frac{V^2}{4}}{I_{D,MAX} V_{DC} \frac{V^2}{4} + I_{D,MAX} V_{DC} \left(\frac{V}{\pi} - \frac{V^2}{4}\right)} = \frac{V\pi}{4}$$
(7.17)

# 7.2 Class D

## 7.2.1 Current Mode Class D RF PWM

In this section the control characteristic of the current mode class D PA with respect to RF PWM will be derived. The circuit for the current mode class D PA is given in Fig. 7.2.



Figure 7.2: Current mode class D.

The load voltage can be set to

$$V_L = N\cos(\theta) + V_{LDC} \tag{7.18}$$

to maintain symmetry and to allow for easier integration. The drain voltages during the first period  $-\theta_1/2 \le \theta \le \theta_1/2$  can be written as

$$V_{D1}^1 = 0; \ V_{D2}^1 = V_L \tag{7.19}$$

and during the second period  $\theta_1/2 \le \theta \le (2\pi - \theta_1/2)$  as

$$V_{D1}^2 = -V_L; \ V_{D2}^2 = 0. (7.20)$$

It shall be noted that the superscript  $x^1$  denotes the first period, where  $T_1$  is active, and the superscript  $x^2$  denotes the second period, where  $T_2$  is active. The DC content of the drain voltages has to correspond to the supply voltages and can be calculated by the integral over one cycle

$$V_{DC1} = \frac{1}{2\pi} \int_{0}^{2\pi} V_{D1} d\theta = \frac{1}{2\pi} \int_{\theta_1/2}^{2\pi - \theta_1/2} - (N\cos(\theta) + V_{LDC}) d\theta = \frac{N\sin(\frac{\theta_1}{2}) + (\frac{\theta_1}{2} - \pi)V_{LDC}}{\pi}.$$
(7.21)

$$N = \frac{\pi V_{DC1} - (\frac{\theta_1}{2} - \pi) V_{L DC}}{\sin(\frac{\theta_1}{2})}$$
(7.22)

Doing the same for the second drain voltage delivers

$$V_{DC2} = \frac{1}{2\pi} \int_{0}^{2\pi} V_{D2} d\theta = \frac{1}{2\pi} \int_{-\theta_1/2}^{\theta_1/2} \left( N \cos(\theta) + V_{LDC} \right) d\theta = \frac{N \sin(\frac{\theta_1}{2}) + \frac{\theta_1}{2} V_{LDC}}{\pi}.$$
 (7.23)

Resolving for N yields

$$N = \frac{\pi V_{DC2} - \frac{\theta_1}{2} V_{LDC}}{\sin(\frac{\theta_1}{2})}.$$
(7.24)

Setting (7.22) and (7.24) equal and resolving for  $V_{LDC}$  delivers

$$V_{L DC} = V_{DC2} - V_{DC1}.$$
 (7.25)

For equal supply voltages  $V_{DC} = V_{DC1} = V_{DC2}$  the control characteristic reduces to

$$N = \frac{\pi V_{DC}}{\sin(\frac{\theta_1}{2})}.\tag{7.26}$$

The output power for this case is given by

$$P_L = \frac{N^2}{2R_L} = \frac{\pi^2 V_{DC}^2}{2R_L \sin^2(\frac{\theta_1}{2})}.$$
(7.27)

The maximum negative voltage at the drain is given by  $V_{D1}(\theta_1)$  and can be calculated using

$$V_{D1N} = -N\cos(\frac{\theta_1}{2}) = -\frac{\pi V_{DC}}{\sin(\frac{\theta_1}{2})}\cos(\frac{\theta_1}{2}) = -\pi V_{DC}\cot(\frac{\theta_1}{2}),$$
(7.28)

which results in large negative voltages for lower duty cycles.

A modified version of the current mode class D PA with a series resonator load is depicted in Fig. 7.3.



Figure 7.3: Current mode class D with series resonator.

The load current can be set as

$$I_L = \frac{V_{L1}}{R_L} \cos(\theta) \tag{7.29}$$

to maintain symmetry and to allow for easier integration. The drain voltages during the first period  $-\theta_1 \leq \theta \leq \theta_1$  can be written as

$$V_{D1}^1 = 0; \ V_{D2}^1 = N_2 \left( \cos(\theta) - \cos(\frac{\theta_1}{2}) \right)$$
 (7.30)

and during the second period  $\theta_1 \leq \theta \leq (2\pi - \theta_1)$  as

$$V_{D1}^2 = N_1 \left( \cos(\frac{\theta_1}{2}) - \cos(\theta) \right); \ V_{D2}^2 = 0.$$
(7.31)

The DC content of the drain voltages has to correspond to the supply voltages and can be calculated by the integral over one cycle.

$$V_{DC1} = \frac{1}{2\pi} \int_{0}^{2\pi} V_{D1} d\theta = \frac{1}{2\pi} \int_{\theta_1/2}^{2\pi - \theta_1/2} N_1 \left( \cos(\frac{\theta_1}{2}) - \cos(\theta) \right) d\theta =$$

$$= N_1 \frac{\sin(\frac{\theta_1}{2}) + (\pi - \frac{\theta_1}{2})\cos(\frac{\theta_1}{2})}{\pi}$$
(7.32)

$$N_{1} = \frac{\pi V_{DC1}}{\sin(\frac{\theta_{1}}{2}) + (\pi - \frac{\theta_{1}}{2})\cos(\frac{\theta_{1}}{2})}$$
(7.33)

$$V_{DC2} = \frac{1}{2\pi} \int_{0}^{2\pi} V_{D2} d\theta = \frac{1}{2\pi} \int_{-\theta_1/2}^{\theta_1/2} N_2 \left( \cos(\theta) - \cos(\frac{\theta_1}{2}) \right) = N_2 \frac{\sin(\frac{\theta_1}{2}) - \frac{\theta_1}{2}\cos(\frac{\theta_1}{2})}{\pi} \quad (7.34)$$

$$N_{2} = \frac{\pi V_{DC2}}{\sin(\frac{\theta_{1}}{2}) - \frac{\theta_{1}}{2}\cos(\frac{\theta_{1}}{2})}$$
(7.35)

The load voltage can be defined as

$$V_L = V_{D2} - V_{D1}. (7.36)$$

The Fourier coefficients of the load voltage can be calculated by the Fourier series, which is given by

$$V_{L,FUND} = \frac{1}{\pi} \int_{0}^{2\pi} V_L \cos(\theta) d\theta = -\frac{N_1}{\pi} \int_{\theta_1/2}^{2\pi - \theta_1/2} \left( \cos(\frac{\theta_1}{2}) - \cos(\theta) \right) \cos(\theta) d\theta + \frac{N_2}{2\pi} \int_{-\theta_1/2}^{\theta_1/2} \left( \cos(\theta) - \cos(\frac{\theta_1}{2}) \right) \cos(\theta) d\theta$$
(7.37)  
$$= \frac{N_1}{\pi} \left( \sin(\theta_1) + 2\pi - \theta_1 \right) + \frac{N_2}{\pi} \left( \theta_1 - \sin(\theta_1) \right)$$
$$V_{L,FUND} = \frac{V_{DC1} \left( \sin(\theta_1) + 2\pi - \theta_1 \right)}{2\sin(\frac{\theta_1}{2}) + 2(\pi - \frac{\theta_1}{2}) \cos(\frac{\theta_1}{2})} + \frac{V_{DC2} \left( \theta_1 - \sin(\theta_1) \right)}{2\sin(\frac{\theta_1}{2}) - 2\frac{\theta_1}{2} \cos(\frac{\theta_1}{2})}$$
(7.38)

Considering equal supply voltage the load voltage can be approximated by

$$V_{L,FUND} \approx V_{DC}\pi. \tag{7.39}$$

The drain current at the fundamental component does not depend on the biasing current and is defined by

$$I_{D1,FUND} = I_L = \frac{V_{L,FUND}}{R_L} = \frac{1}{\pi} \int_0^{2\pi} I_{D1} \cos(\theta) d\theta = \frac{1}{\pi} \int_{-\theta_1/2}^{\theta_1/2} I_{D1,MAX} \cos(\theta) d\theta.$$
(7.40)

Solving the integral and rearranging delivers

$$I_{D1,MAX} = \frac{2\pi V_{L1}}{R_L \sin(\frac{\theta_1}{2})} \approx \frac{2\pi^2 V_{DC}}{R_L \sin(\frac{\theta_1}{2})}.$$
(7.41)

The maximum drain voltage can be calculated by

$$V_{D1,MAX} = \frac{\pi V_{DC1} \left( \cos(\frac{\theta_1}{2}) + 1 \right)}{\sin(\frac{\theta_1}{2}) + (\pi - \frac{\theta_1}{2}) \cos(\frac{\theta_1}{2})}.$$
(7.42)

#### 7.2.2 Voltage Mode Class D

The properties of the voltage mode class D will be derived in this section. The circuit of the voltage mode class D PA is given in Fig. 7.4.



Figure 7.4: Voltage mode class D.

The desired rectangular drain voltages can be written as

$$V_{D1} = K(1 - \text{pulse}(\theta)), \tag{7.43}$$

$$V_{D2} = K \operatorname{pulse}(\theta), \tag{7.44}$$

with  $pulse(\theta)$  being

$$pulse(\theta) = \begin{cases} 1 & 0 \le \theta \le \theta_1 \\ 0 & else \end{cases}$$
(7.45)

The DC content of the drain voltage of the first transistor can be calculated by

$$V_{DC1} = \frac{1}{2\pi} \int_{0}^{2\pi} V_{D1} d\theta = \frac{1}{2\pi} \int_{\theta_1}^{2\pi} K d\theta = \frac{K(2\pi - \theta_1)}{2\pi}.$$
 (7.46)

Doing the same for the second transistor results in

$$V_{DC2} = \frac{1}{2\pi} \int_{0}^{2\pi} V_{D2} d\theta = \frac{1}{2\pi} \int_{0}^{\theta_1} K \, d\theta = \frac{K\theta_1}{2\pi}.$$
 (7.47)

Considering the fact that the DC content of both drain signals has to be equal to the supply voltage  $V_{DC}$  gives

$$\frac{K(2\pi - \theta_1)}{2\pi} = \frac{K\theta_1}{2\pi}.$$
(7.48)

It can be seen that (7.48) can only be held for  $\theta_1 = \pi$ . Thus the circuit can only be operated in the desired way for equal duty cycles as a result of the fixed DC biasing. Due to that further derivations will be restricted to  $\theta_1 = \pi$ . Using (7.46) or (7.47) the scaling factor K can be resolved to  $K = 2V_{DC}$ . The fundamental load voltage can be determined by calculating the Fourier transformation of the drain voltage

$$V_L = \frac{1}{\pi} \int_0^{2\pi} V_{D1} \sin(\theta) d\theta = \frac{2V_{DC}}{\pi} \int_0^{\pi} \sin(\theta) d\theta = \frac{4V_{DC}}{\pi}.$$
 (7.49)

The corresponding load current is given by

$$I_L = \frac{V_L}{R_L} = \frac{4V_{DC}}{\pi R_L}.$$
(7.50)

The drain currents are defined as

$$I_{D1} = \begin{cases} I_L \sin(\theta) & 0 \le \theta \le \pi \\ 0 & \text{else} \end{cases}$$
(7.51)

and

$$I_{D2} = \begin{cases} 0 & 0 \le \theta \le \pi \\ -I_L \sin(\theta) & \text{else} \end{cases}$$
(7.52)

Considering the maximum drain current  $I_{DMAX}$  the load resistor can be resolved by

$$R_L = \frac{4V_{DC}}{\pi I_{DMAX}}.$$
(7.53)

The resulting output power with respect to maximum drain voltage  $V_{DMAX} = 2V_{DC}$  and maximum drain current is given by

$$P_L = \frac{V_L^2}{2R_L} = \frac{V_{DMAX}I_{DMAX}}{\pi}.$$
 (7.54)

Comparing the power per transistor  $(P_L/2)$  with the output power of a class A PA  $(V_{DMAX}I_{DMAX}/8)$  delivers a Power Utilization Factor (PUF) of

$$PUF = \frac{4}{\pi}.\tag{7.55}$$

## 7.2.3 Inverter Based Voltage Mode Class D

In this section the properties of the inverter based voltage mode class D with respect to RF PWM control will be derived. As efficient control of the output voltage with the push pull configuration is not possible an inverter structure as depicted in Fig. 7.5 can be used.



Figure 7.5: Voltage mode class D.

The drain voltage  $V_D$  is defined as

$$V_D = V_{DC} \operatorname{pulse}(\theta), \tag{7.56}$$

with

$$pulse(\theta) = \begin{cases} 1 & 0 \le \theta \le \theta_1 \\ 0 & else \end{cases}$$
 (7.57)

The Fourier coefficients of this waveform can be calculated by the Fourier series of the signal, which is given by

$$a_n = \frac{1}{\pi} \int_{0}^{2\pi} V_D \sin(n\theta) d\theta = \frac{V_{DC}}{\pi} \int_{0}^{\theta_1} \sin(n\theta) d\theta = V_{DC} \frac{1 - \cos(n\theta_1)}{\pi n},$$
(7.58)

$$b_n = \frac{1}{\pi} \int_0^{2\pi} V_D \cos(n\theta) d\theta = \frac{V_{DC}}{\pi} \int_0^{\theta_1} \cos(n\theta) d\theta = V_{DC} \frac{\sin(n\theta_1)}{\pi n}.$$
 (7.59)

The resulting amplitude can be calculated by

$$A_n = \sqrt{a_n^2 + b_n^2} = 2V_{DC} \frac{\sin(n\theta_1/2)}{n\pi}.$$
(7.60)

The corresponding phase is given by

$$\phi = \tan^{-1} \left( \frac{a_n}{b_n} \right) = \frac{n\theta_1}{2}.$$
(7.61)

The output control characteristic is determined by the amplitude characteristic and it is nonlinear. For predistortion the duty cycle signal has to be generated by  $\sin^{-1}$  to generate the desired amplitude signal and to preserve linearity.

A main problem of the voltage mode class D is its capacitive losses due to charging and discharging of the parasitic drain source capacitance  $C_{DS}$  of the transistors. To calculate their influence it is required to get the maximum output power of the circuit. The maximum drain amplitude according to (7.60) is

$$V_{MAX} = \frac{2}{\pi} V_{DC}, \qquad (7.62)$$

and the maximum load current is simply the maximum transistor current  $I_{MAX}$ . Thus the load resistance can be calculated by

$$R_L = \frac{V_{MAX}}{I_{MAX}} = \frac{2V_{DC}}{\pi I_{MAX}}$$
(7.63)

and the output power can be written as

$$P_L(V) = V^2 P_{MAX} = V^2 \frac{V_{DC} I_{MAX}}{\pi},$$
(7.64)

where V denotes the normalized load voltage.

For the efficiency of the circuit the (constant) capacitive losses have to be taken into account. The capacitive losses have to be considered for both signal transitions and for both transistors, which results in

$$P_{CAP} = 2V_{DC}^2 C_{DS} f_C. (7.65)$$

The resulting maximum efficiency is given by

$$\eta_{MAX} = \frac{P_L(1)}{P_L(1) + P_{CAP}} = \frac{1}{1 + \frac{2\pi V_{DC}C_{DS}f_C}{I_{MAX}}}.$$
(7.66)

Resolving this equation for  $P_{CAP}$  and inserting it into the efficiency calculation delivers

$$\eta(V) = \frac{V^2 \eta_{MAX}}{\eta_{MAX}(V^2 - 1) + 1}.$$
(7.67)
## 7.3 Class DE

#### 7.3.1 Class DE With Parallel Compensation Inductance

In this section the properties and design equations for the class DE PA with parallel compensation inductance, as given in Fig. 7.6, will be provided. For the derivation all



Figure 7.6: Voltage mode class DE with parallel compensation inductance.

voltages and currents will be normalized to  $V_{DC}$ . Setting up the current and voltage relations for the 1<sup>st</sup> period (switch S<sub>1</sub> closed) delivers the following relations. Due to the symmetry of the waveform the normalized DC voltage at the drain evaluates to 1/2, hence the inductor voltage during the first period results to

$$V_{L1}(\theta) = \frac{1}{2},\tag{7.68}$$

The current flow through the capacitor  $(I_{C1}(\theta) = 0)$  is zero, as the voltage remains constant, while the inductor current is given by

$$I_{L1}(\theta) = \frac{1}{\omega L_P} \int V_{L1}(\theta) d\theta = \frac{1}{2\omega L} \left(\theta + 2I_{L01}\right),$$
(7.69)

where  $I_{L01}$  denotes the initial inductor current. The load current is defined by

$$I_{R1}(\theta) = \frac{A}{R}\cos\left(\theta\right) + \frac{B}{R}\sin\left(\theta\right)$$
(7.70)

and the summation current taking the parallel inductor into account is given as

$$I_{S1}(\theta) = I_{L1} + I_{R1}. \tag{7.71}$$

The voltage at the inductance for the dead time (both switches are open) can be approximated by

$$V_{L2}(\theta) = (1 + \cos(\theta)) z - \frac{1}{2}.$$
(7.72)

The factor n can be resolved by inserting the boundary condition for the beginning of the dead time with  $V_{L2}(\theta_1) = 1/2$ . This delivers

$$z = \frac{1}{1 + \cos(\theta_1)}.$$
 (7.73)

The inductance current can be calculated by integrating over the voltage.

$$I_{L2}(\theta) = \frac{1}{\omega L} \int V_{L2}(\theta) d\theta = \frac{1}{\omega L} \left[ \left( z - \frac{1}{2} \right) \theta + z \sin(\theta) + I_{L02} \right]$$
(7.74)

The load current is the same as during the first period as the series resonant filter forces a sinusoidal current and is defined by

$$I_{R2}(\theta) = I_{R1}(\theta) = \frac{A}{R}\cos\left(\theta\right) + \frac{B}{R}\sin\left(\theta\right).$$
(7.75)

The current through the capacitance is the sum of the load and the inductor current and is given by

$$I_{C2}(\theta) = -I_{L2}(\theta) - I_{R2}(\theta).$$
(7.76)

Hence the voltage across the capacitor can be derived by integration.

$$U_{C2}(\theta) = \frac{1}{\omega C} \int I_{C2}(\theta) d\theta$$
(7.77)

$$U_{C2}(\theta) = -\frac{1}{\omega^2 LC} \left[ \left( z - \frac{1}{2} \right) \frac{\theta^2}{2} - z \cos(\theta) + I_{L02} \theta \right] - \frac{1}{\omega RC} \left( A \sin(\theta) - B \cos(\theta) \right) \quad (7.78)$$

By using the requirement that there is a continuous current through the inductance  $I_{L1}(\theta_1) = I_{L2}(\theta_1)$  and by taking into account the symmetry of the waveforms  $I_{L2}(\pi) = -I_{L1}(0)$  the initial inductor currents can be resolved as

$$I_{L02} = \frac{1}{2} \left[ \frac{1}{2} \theta_1 - \left( z - \frac{1}{2} \right) (\pi + \theta_1) - z \sin(\theta_1) \right]$$
(7.79)

and the initial current for period one can be calculated by

$$I_{L01} = \left(\frac{1}{2} - z\right)\pi - I_{L02}.$$
(7.80)

By performing a Fourier analysis of the voltage waveform the normalized fundamental voltage components can be determined.

$$A = \frac{2}{\pi} \int_{0}^{\pi} V_C(\theta) \cos(\theta) d\theta; \ B = \frac{2}{\pi} \int_{0}^{\pi} V_{C1}(\theta) \sin(\theta) d\theta$$
(7.81)

$$A = \frac{z}{\pi} \left( \pi - \theta_1 + \frac{\sin(2\theta_1)}{2} \right); \ B = \frac{z}{\pi} \left( 1 - \cos^2(\theta_1) \right)$$
(7.82)

To fulfill class E conditions it is required to have zero current switching.

$$I_{L2}(\pi) + I_{R2}(\pi) = 0 \tag{7.83}$$

When considering (7.74) and (7.75) the inductance can be resolved.

$$L = \frac{R}{\omega A} \left[ \left( z - \frac{1}{2} \right) \pi + \frac{\theta_1}{4} - \left( \frac{z}{2} - \frac{1}{4} \right) (\pi + \theta_1) - \frac{z}{2} \sin(\theta_1) \right]$$
(7.84)

The equation can be rearranged to calculate the inductance factor  $K_L$  as

$$\omega L = R K_L, \tag{7.85}$$

which yields

$$K_L = \frac{1}{A} \left[ \left( z - \frac{1}{2} \right) \pi + \frac{\theta_1}{4} - \left( \frac{z}{2} - \frac{1}{4} \right) (\pi + \theta_1) - \frac{z}{2} \sin(\theta_1) \right].$$
(7.86)

The voltage difference of the capacitor voltage (7.78) between the two states is exactly the supply voltage  $U_{C2}(\theta_1) - U_{C2}(\pi) = V_{DC}$ . Taking this into account, inserting (7.79), (7.86) and resolving delivers

$$\omega RC = \frac{1}{K_L} \left[ \left( z - \frac{1}{2} \right) \frac{(\pi^2 - \theta_1^2)}{2} + 1 + I_{L02} \left( \pi - \theta_1 \right) \right] + B \left( 1 + \cos(\theta_1) \right) - A \sin(\theta_1).$$
(7.87)

Following that the capacitor factor can be defined as

$$\omega C = \frac{K_C}{R},\tag{7.88}$$

with

$$K_C(\theta_1) = \frac{1}{K_L} \left[ \left( z - \frac{1}{2} \right) \frac{\left( \pi^2 - \theta_1^2 \right)}{2} + 1 + I_{L02} \left( \pi - \theta_1 \right) \right] + B \left( 1 + \cos(\theta_1) \right) - A \sin(\theta_1).$$
(7.89)

The maximum current through the switch can be determined by using (7.71) and setting its derivative to zero

$$I_{S1}(\theta)\frac{d}{d\theta} = 0. \tag{7.90}$$

Doing so the phase for the maximum current  $\theta'_{MAX}$  can be resolved.

$$\theta'_{MAX} = \sin^{-1}(arg) + \tan^{-1}\left(\frac{B}{A}\right); \ arg = \frac{1}{2K_L\sqrt{A^2 + B^2}}$$
 (7.91)

The maximum current through the switch occurs either during or at the end of the ON period. The corresponding phase is determined by

$$\theta_{MAX} = \begin{cases} \theta_1 & \theta'_{MAX} > \theta_1 \text{ or } ||arg|| > 1, \\ \theta'_{MAX} & \text{else} \end{cases}$$
(7.92)

The maximum switch current is the normalized switch current times the supply voltage at the maximum current phase  $I_{MAX} = V_{DC}I_{S1}(\theta_{MAX})$ . Using (7.71) and (7.86) yields

$$I_{MAX} = \frac{V_{DC}}{R} \left[ \frac{1}{K_L} \left( \frac{\theta_{MAX}}{2} + I_{L01} \right) + A\cos(\theta_{MAX}) + B\sin(\theta_{MAX}) \right].$$
(7.93)

The load resistance can be calculated by

$$R = \frac{V_{DC}}{I_{MAX}} K_R,\tag{7.94}$$

with  $K_R$  being the load resistance factor.

$$K_R(\theta_1) = \frac{1}{K_L} \left( \frac{\theta_{MAX}}{2} + I_{L01} \right) + A\cos(\theta_{MAX}) + B\sin(\theta_{MAX})$$
(7.95)

Taking into account that the shunt capacitance is twice the drain source capacitance  $C = 2 C_{DS}$ , inserting (7.94) into (7.88) and rearranging the equation results in

$$\omega_{MAX}(\theta_1) = \frac{I_{MAX} K_C(\theta_1)}{2C_{DS} V_{DC} K_R(\theta_1)}.$$
(7.96)

The normalized maximum frequency is defined as

$$\omega_{MAX,N} = \omega_{MAX} C_{DS} \frac{V_{DC}}{I_{MAX}} = \frac{K_C(\theta_1)}{2K_R(\theta_1)}.$$
(7.97)

The output power can be calculated using

$$P_L = \frac{V_{DC}^2 \left(A^2 + B^2\right)}{2R} = \frac{V_{DC} I_{MAX} \left(A^2 + B^2\right)}{2K_R(\theta_1)}$$
(7.98)

#### 7.3.2 Class DE Amplifier With Series Inductance

In this section the design parameters for the class DE PA in Fig. 7.7 will be provided. According to [55] the ratio of load resistance R to series inductance  $L_S$  is determined by



Figure 7.7: Voltage mode class DE with series compensation inductance.

$$\omega L_S = R K_{R,S}; \ K_{R,S} = \frac{A}{B}.$$
 (7.99)

The load current is given by

$$I_R(\theta) = V_{DC} \frac{B}{R} \sin(\theta) \,. \tag{7.100}$$

The voltage across the capacitance can be calculated by integrating the current. The voltage difference during the OFF period has to be the supply voltage and is given by the following relation

$$-V_{DC} = -\frac{1}{\omega C} \int_{\theta_1}^{\pi} I_R(\theta) d\theta.$$
(7.101)

Resolving the integral and rearranging the equation delivers

$$\omega RC = K_{C,S}; \ K_{C,S} = B\left(1 + \cos\left(\theta_1\right)\right).$$
(7.102)

The maximum switch current is given by

$$I_{MAX} = V_{DC} \frac{B}{R} \sin\left(\theta_{MAX}\right), \qquad (7.103)$$

with the maximum current phase  $\theta_{MAX}$  being

$$\theta_{MAX} = \begin{cases} \pi/2 & (\theta_1 > \pi/2), \\ \theta_1 & \text{else} \end{cases}$$
(7.104)

Thus the load resistance can be calculated by

$$R = \frac{V_{DC}}{I_{MAX}} K_{R,S}; \ K_{R,S} = B\sin(\theta_{MAX}).$$
(7.105)

The maximum frequency can be calculated the same way as for the parallel inductance (7.96) and results in

$$\omega_{MAX,S}(\theta_1) = \frac{I_{MAX} K_{C,S}(\theta_1)}{2C_{DS} V_{DC} K_{R,S}(\theta_1)} = \frac{I_{MAX} (1 + \cos(\theta_1))}{2C_{DS} V_{DC} \sin(\theta_{MAX})},$$
(7.106)

with the normalized maximum frequency

$$\omega_{MAX,S,N} = \omega_{MAX,S} C_{DS} \frac{V_{DC}}{I_{MAX}} = \frac{1 + \cos(\theta_1)}{2\sin(\theta_{MAX})}.$$
(7.107)

The output power for the series compensation inductance can be calculated using

$$P_{L,S} = \frac{V_{DC}^2 B^2}{2R} = \frac{V_{DC} I_{MAX} B}{2\sin(\theta_{MAX})}.$$
(7.108)

## 7.4 BB PWM

In this section the average drain efficiency of BB PWM operation between two operational points will be calculated. The load powers for each period with respect to the duty cycle d can be calculated by

$$P_{L1} = \frac{V_{L1}^2 d}{2R_L}, \ P_{L2} = \frac{V_{L2}^2 (1-d)}{2R_L}.$$
(7.109)

The corresponding DC power consumption is given by

$$P_{DC1} = \frac{V_{L1}^2 d}{2R_L \eta_{D1}}, \ P_{DC2} = \frac{V_{L2}^2 (1-d)}{2R_L \eta_{D2}}.$$
 (7.110)

The average drain efficiency is defined as follows

$$\eta_D = \frac{P_{L1} + P_{L2}}{P_{DC1} + P_{DC2}}.$$
(7.111)

Inserting (7.109) and (7.110) into (7.111) and deriving the dependency on the duty cycle d results in

$$\eta_D(d) = \frac{\left(V_{L1}^2 d + V_{L2}^2 (1-d)\right) \eta_{D1} \eta_{D2}}{V_{L1}^2 d \eta_{D2} + V_{L2}^2 (1-d) \eta_{D1}}.$$
(7.112)

Considering the relationship of duty cycle and load voltage

$$V = (V_{L1} - V_{L2})d + V_{L2}, (7.113)$$

inserting this into (7.112) and rearranging delivers

$$\eta_D(V) = \frac{\left(V_{L1}^2(V - V_{L2}) + V_{L2}^2(V_{L1} - V)\right)\eta_{D1}\eta_{D2}}{V_{L1}^2(V - V_{L2})\eta_{D2} + V_{L2}^2(V_{L1} - V)\eta_{D1}}.$$
(7.114)

For the system efficiency not only the drain, but also the coding efficiency is important. The coding efficiency is the ratio between the desired in band energy and the total energy generated by the PA.

$$\eta_C = \frac{P_L(V)}{P_{L1} + P_{L2}} \tag{7.115}$$

Taking into account that the inband energy can be calculated by

$$P_L(V) = \frac{V^2}{2R_L} = \frac{\left((V_{L1} - V_{L2})d + V_{L2}\right)^2}{2R_L}$$
(7.116)

and inserting into (7.115) and simplifying delivers

$$\eta_C(d) = \frac{\left((V_{L1} - V_{L2})d + V_{L2}\right)^2}{V_{L1}^2 d + V_{L2}^2 (1 - d)}.$$
(7.117)

Considering (7.113) and rearranging delivers

$$\eta_C(V) = \frac{V^2(V_{L1} - V_{L2})}{V_{L1}^2(V - V_{L2}) + V_{L2}^2(V_{L1} - V)}.$$
(7.118)

The system efficiency is the product of drain and coding efficiency and can be calculated by

$$\eta(V) = \eta_D(V)\eta_C(V) = \frac{V^2(V_{L1} - V_{L2})\eta_{D1}\eta_{D2}}{V_{L1}^2(V - V_{L2})\eta_{D2} + V_{L2}^2(V_{L1} - V)\eta_{D1}}$$
(7.119)

and is valid for  $V_{L2} \leq V \leq V_{L1}$ . To cover multilevel operation the calculation has to be done separately for the different levels.

## 7.5 Energy Recovery

In this section the properties of DC and RF energy recovery systems will be derived. The section deals with DC energy recovery for BB PWM and outphasing and also covers RF energy reuse. Finally the properties of DC BPSK energy recovery are analyzed.

#### 7.5.1 DC Energy Recovery BB PWM

The basic circuit diagram of BB PWM energy recovery is given in Fig. 7.8. Instead of dissipating the energy in the isolator it can be rectified and reused to achieve efficiency enhancement.



Figure 7.8: DC energy recovery BB PWM.

The power generated by the PA and fed to the circulator with respect to the drain voltage  $V_D$  and the duty cycle d is defined as

$$P_D = \frac{V_D^2 d}{2Z_L}.$$
(7.120)

The corresponding power at the load is given by

$$P_L = \frac{V_L^2}{2Z_L}.$$
 (7.121)

Considering  $V_L = V_D d$  results in

$$P_L = \frac{V_D^2 d^2}{2Z_L}.$$
(7.122)

According to the conservation of energy the sideband power at the rectifier is given by

$$P_{SB} = P_D - P_L = \frac{V_D^2}{2Z_L} (d - d^2).$$
(7.123)

The resulting coding efficiency is defined by the ratio of the output power at the load  $P_L$  to the total drain power  $P_D$ 

$$\eta_C(d) = \frac{P_L}{P_D} = \frac{\frac{V_D^2 d^2}{2Z_L}}{\frac{V_D^2 d}{2Z_L}} = d.$$
(7.124)

The overall efficiency is given by

$$\eta(d) = \frac{P_L}{P_{DC}} = \eta_C \eta_D = d\eta_D. \tag{7.125}$$

Considering now energy recovery with rectifier efficiency  $\eta_R$  and  $P_{DC} = P_D/\eta_d$  gives

$$\eta_{DC}(d) = \frac{P_L}{P_{DC} - P_{SB}\eta_r} = \frac{d^2}{\frac{d}{\eta_D} - (d - d^2)\eta_R} = \frac{d}{1 - (1 - d)\eta_R\eta_D}\eta_D.$$
(7.126)

Taking into account that the average normalized output voltage is equal to the duty cycle V = d delivers

$$\eta(V) = V\eta_d; \ \eta_{DC}(V) = \frac{V}{1 - (1 - V)\eta_R\eta_D}\eta_D$$
(7.127)

for the efficiency in dependence to the normalized load voltage.

#### 7.5.2 DC Energy Recovery Outphasing

In this section the theoretical performance of DC energy recovery for outphasing systems will be derived. Fig. 7.9 depicts the block diagram of an outphasing transmitter. Instead of dissipating the out of phase energy ( $\Delta$ ) it can be rectified and reused in order to enhance the efficiency.



Figure 7.9: DC energy recovery outphasing.

The drain voltages  $V_{D1}$  and  $V_D$  are defined as

$$V_{D1} = V_D e^{j\phi}; \ V_{D2} = V_D e^{-j\phi}. \tag{7.128}$$

The total output power is the sum of the absolute power of both PAs and is given by

$$P_D = \frac{V_{D1}^2 + V_{D2}^2}{2Z_L} = \frac{V_D^2}{Z_L}.$$
(7.129)

The voltage at the load resistor and at the out of phase reflection termination is given by

$$V_L = \frac{1}{\sqrt{2}}(V_{D1} + V_{D2}); \ V_R = \frac{1}{\sqrt{2}}(V_{D1} - V_{D2}).$$
 (7.130)

The output power at the load resistor is the inphase combined output power

$$P_L = \frac{V_L^2}{2Z_L} = \frac{V_{D1}^2 + 2V_{D1}V_{D2}\cos(2\phi) + V_{D2}^2}{4Z_L} = \frac{V_D^2(1 + \cos(2\phi))}{2Z_L}.$$
(7.131)

The power dissipated in the out of phase termination resistor is given by

$$P_R = \frac{V_R^2}{2Z_L} = \frac{V_{D1}^2 - 2V_{D1}V_{D2}\cos(2\phi) + V_{D2}^2}{4Z_L} = \frac{V_D^2(1 - \cos(2\phi))}{2Z_L}.$$
(7.132)

The efficiency can be calculated using

$$\eta(\phi) = \frac{P_L}{P_D - P_R \eta_D \eta_R} \eta_D = \frac{(1 + \cos(2\phi))}{2 - (1 - \cos(2\phi))\eta_D \eta_R} \eta_D = \frac{\cos^2(\phi)}{1 - (1 - \cos^2(\phi))\eta_D \eta_R} \eta_D,$$
(7.133)

where  $\eta_R$  denotes the rectifier efficiency defined in the previous section. Considering that the magnitude of both drain voltages are equal  $|V_D| = |V_{D1}| = |V_{D2}|$  the load voltage can be calculated by

$$V_L = \frac{1}{\sqrt{2}} V_D(e^{j\phi} + e^{-j\phi}) = \sqrt{2} V_D \cos\phi.$$
(7.134)

Using (7.134) and introducing the normalized voltage  $V = V_L/V_D/\sqrt{2}$  yields

$$\eta(V) = \frac{V^2}{1 - (1 - V^2) \eta_D \eta_R} \eta_D.$$
(7.135)

This equation simplifies to  $\eta(V) = V^2 \eta_D$  for the case that no energy recovery is done.

#### 7.5.3 RF Energy Recovery

In this section the properties of RF energy recovery will be derived. The concept is based on reusing the energy in the modulation sidebands for BB PWM operation by employing a variable reflection as depicted in Fig. 7.10. It will be assumed that the BPF is implemented as a parallel resonator. The resulting equations in terms of efficiency and output voltage when using a series resonator are the same and, therefore, only the derivation for the parallel resonator will be provided.



Figure 7.10: RF energy recovery.

Setting up the relationship between PA and load delivers

$$b_1 = a_2 S_R; \ b_2 = a_1, \tag{7.136}$$

with

$$b = \frac{V - Z_0 I}{2\sqrt{Z_0}}; \ a = \frac{V + Z_0 I}{2\sqrt{Z_0}}, \tag{7.137}$$

Combining both equations results in

$$V_1 + Z_0 I_1 = V_2 - Z_0 I_2, (7.138)$$

$$V_1 - Z_0 I_1 = (V_2 + Z_0 I_2) S_R. (7.139)$$

Taking this relation for both periods into account results in two equations for the ON period

$$V_1^{ON} + Z_0 I_1^{ON} = V_2^{ON} - Z_0 I_2^{ON}, (7.140)$$

$$V_1^{ON} - Z_0 I_1^{ON} = (V_2^{ON} + Z_0 I_2^{ON}) S_R^{ON}$$
(7.141)

and two equations for the OFF period

$$V_1^{OFF} + Z_0 I_1^{OFF} = V_2^{OFF} - Z_0 I_2^{OFF}, (7.142)$$

$$V_1^{OFF} - Z_0 I_1^{OFF} = (V_2^{OFF} + Z_0 I_2^{OFF}) S_R^{OFF}.$$
(7.143)

#### 7.5.3.1 Energy Recovery OFF Period

Now, energy recovery during the OFF period under the assumption that a parallel resonator is used will be considered. Setting up the equations for the ON period and assuming  $V_1^{ON} = 1$  and  $V_2^{ON} = V_2^{OFF} = V_L$  due to the parallel resonator and  $S_R^{ON} = 0$  results in the following equations

$$1 + Z_0 I_1^{ON} = V_L - Z_0 I_2^{ON}, (7.144)$$

$$1 - Z_0 I_1^{ON} = (V_L + Z_0 I_2^{ON})0. (7.145)$$

These can be resolved to deliver

$$I_1^{ON} = \frac{1}{Z_0},\tag{7.146}$$

$$I_2^{ON} = \frac{V_L - 2}{Z_0}.$$
(7.147)

Setting up the equations for the OFF period considering that the PA provides an open circuit and thus  $I_1^{OFF} = 0$  yields

$$V_1^{OFF} + Z_0 0 = V_L - Z_0 I_2^{OFF}, (7.148)$$

$$V_1^{OFF} - Z_0 0 = (V_2 L + Z_0 I_2^{OFF}) S_R^{OFF}.$$
(7.149)

As no energy is lost during the OFF period and thus  $I_2^{OFF} = 0$  (7.148) simplifies to

$$V_1^{OFF} = V_L. (7.150)$$

Inserting (7.150) into (7.149) delivers

$$V_L = (V_L + Z_0 0) S_R^{OFF}; \ S_R^{OFF} = 1.$$
(7.151)

Considering the following relation for the parallel resonator

$$\frac{-V_L}{Z_0} = I_2^{ON} d + I_2^{OFF} (1 - d)$$
(7.152)

and inserting (7.146) and (7.147) delivers

$$\frac{-V_L}{Z_0} = \frac{V_L - 2}{Z_0}d + 0(1 - d).$$
(7.153)

Resolving for  $V_L$  yields

$$V_L = \frac{2d}{1+d}.$$
 (7.154)

The input power can be calculated by

$$P_1 = \frac{V_1^{ON} I_1^{ON}}{2} \frac{d}{1} = \frac{1}{2Z_0} d \tag{7.155}$$

and the output power by using

$$P_L = \frac{V_L^2}{2Z_0} = \frac{4d^2}{2Z_0(1+d)^2}.$$
(7.156)

The corresponding coding efficiency is given by

$$\eta_C(d) = \frac{P_L}{P_1} = \frac{4d}{(1+d)^2}.$$
(7.157)

Resolving with respect to output voltage  $V_L$  delivers

$$\eta_C(V_L) = V_L(2 - V_L). \tag{7.158}$$

The resulting PA efficiency is determined by the coding efficiency  $\eta_C(V_L)$  and the drain efficiency  $\eta_D(V_L)$  and is given by

$$\eta(V_L) = \eta_C(V_L)\eta_D(V_L) = V_L(2 - V_L)\eta_D(V_L).$$
(7.159)

#### 7.5.3.2 Energy Recovery ON Period

Now the same calculation is done for energy recovery during the OFF period. Setting up the current and voltage relationship (7.138) and (7.139) for the ON period yields

$$1 + Z_0 I_1^{ON} = V_L - Z_0 I_2^{ON}, (7.160)$$

$$1 - Z_0 I_1^{ON} = (V_L + Z_0 I_2^{ON}) S_R^{ON}.$$
(7.161)

Resolving both equations for  $Z_0 I_1^{ON}$  delivers

$$Z_0 I_1^{ON} = V_L - Z_0 I_2^{ON} - 1, (7.162)$$

$$Z_0 I_1^{ON} = -V_L S_R^{ON} - Z_0 I_2^{ON} S_R^{ON} + 1. ag{7.163}$$

Setting them equal results in

$$V_L - Z_0 I_2^{ON} - 1 = -V_L S_R^{ON} - Z_0 I_2^{ON} S_R^{ON} + 1.$$
(7.164)

The equations for the OFF period considering  $V_1^{OFF} = 0$  and  $S_R^{OFF} = 0$  are defined as

$$0 + Z_0 I_1^{ON} = V_L - Z_0 I_2^{ON} (7.165)$$

$$0 - Z_0 I_1^{ON} = (V_L + Z_0 I_2^{ON})0. (7.166)$$

This results in

$$I_1^{OFF} = 0; \ I_2^{OFF} = \frac{V_L}{Z_0}.$$
 (7.167)

Using this and the condition for the parallel resonator current (7.152) delivers

$$I_2^{ON} = \frac{V_L}{Z_0} \left( 1 - \frac{2}{d} \right).$$
(7.168)

Inserting this into (7.164) and resolving for  $V_L$  yields

$$V_L = \frac{1}{S_R^{ON} + \frac{1 - S_R^{ON}}{d}}.$$
(7.169)

If  $S_R^{ON}=1$  the resulting output voltage does not depend on the duty cycle anymore and thus

$$S_R^{ON} = -1.$$
 (7.170)

Resolving (7.169) considering  $S_R^{ON} = -1$  results in

$$V_L(d) = \frac{d}{2-d}.$$
(7.171)

Taking (7.168) and (7.171) into account and inserting into (7.162) yields

$$I_1^{ON} = \frac{1}{Z_0} \frac{d}{2-d}.$$
(7.172)

The input power can can be calculated using

$$P_1 = \frac{V_1^{ON} I_1^{ON}}{2} \frac{d}{1} = \frac{d^2}{2Z_0(2-d)}.$$
(7.173)

The output power is given by

$$P_L = \frac{V_L^2}{2Z_0} = \frac{d^2}{2Z_0(2-d)^2}.$$
(7.174)

The coding efficiency results in

$$\eta_C(d) = \frac{P_L}{P_1} = \frac{1}{2-d} \tag{7.175}$$

and resolving in terms of output voltage  $V_L$  delivers

$$\eta_C(V_L) = \frac{1 + V_L}{2}.$$
(7.176)

Considering the whole PA efficiency results in

$$\eta(V_L) = \eta_C(V_L)\eta_D(V_L) = \frac{1+V_L}{2}\eta_d(V_L).$$
(7.177)

#### 7.5.3.3 Full Energy Recovery

Finally the properties of full energy recovery during both periods are considered. Setting up the equations for the ON period considering  $V_1^{ON} = 1$  and (7.170) results in

$$1 + Z_0 I_1^{ON} = V_L - Z_0 I_2^{ON}, (7.178)$$

$$1 - Z_0 I_1^{ON} = -V_L - Z_0 I_2^{ON}. (7.179)$$

Resolving both equations for  $Z_0 I_1^{ON}$  yields

$$Z_0 I_1^{ON} = V_L - Z_0 I_2^{ON} - 1, (7.180)$$

$$Z_0 I_1^{ON} = +V_L + Z_0 I_2^{ON} + 1. (7.181)$$

Setting (7.180) and (7.180) equal and resolving for  $I_2^{ON}$  delivers

$$I_2^{ON} = \frac{-1}{Z_0}.$$
(7.182)

Setting up the equations for the OFF period considering (7.151) and  $I_1^{OFF} = 0$  (open circuit) gives

$$V_1^{OFF} = V_L - Z_0 I_2^{OFF}, (7.183)$$

$$V_1^{OFF} = V_L + Z_0 I_2^{OFF}.$$
(7.184)

Resolving for  $I_2^{OFF}$  yields

$$I_2^{OFF} = 0 (7.185)$$

and thus

$$V_1^{OFF} = V_L. (7.186)$$

Considering (7.182), (7.185) and the parallel resonator (7.152) results in

$$V_L = d.$$
 (7.187)

Using this and inserting into (7.180) along with (7.182) and resolving for  $I_1^{ON}$  yields

$$I_1^{ON} = \frac{d}{Z0}.$$
 (7.188)

The input power can be calculated using

$$P_1 = \frac{V_1^{ON} I_1^{ON}}{2} \frac{d}{1} = \frac{d^2}{2Z_0}$$
(7.189)

and the output power by using

$$P_L = \frac{V_L^2}{2Z_0} = \frac{d^2}{2Z_0}.$$
(7.190)

The coding efficiency results in

$$\eta_C(d) = \frac{P_L}{P_1} = 1. \tag{7.191}$$

Thus the PA efficiency is determined only by the drain efficiency  $\eta_d$ .

$$\eta(V_L) = \eta_D(V_L) \tag{7.192}$$

#### 7.5.4 DC BPSK Energy Recovery

In this section the properties of BPSK operation with and without energy recovery will be derived. First BSPK operation with a broadband load (isolator + filter) will be derived. The corresponding block diagram is depicted in Fig. 7.11.



Figure 7.11: BPSK isolated operation.

The normalized drain voltages during the first period  $V_{D1}$  and during the second period  $V_{D2}$  due to the BPSK coding are given by

$$V_{D1} = 1; \ V_{D2} = -1, \tag{7.193}$$

which results in

$$V_L = V_{D1}d + V_{D2}(1-d) = 2d - 1. (7.194)$$

Resolving (7.194) for the duty cycle d delivers

$$d = \frac{1 + V_L}{2}.$$
 (7.195)

The output power at the load can be calculated by

$$P_L = \frac{V_L^2}{2R_L}.$$
 (7.196)

The input power is constant due to the constant envelope in combination with the broadband load and is given by

$$P_D = \frac{V_D^2}{2R_L} = \frac{1}{2R_L}.$$
(7.197)

The resulting efficiency is defined by

$$\eta = \frac{P_L}{P_D} \eta_D = V_L^2 \eta_D. \tag{7.198}$$

Due to the conservation of energy the power in the sideband termination resistor  $R_{SB}$  can be calculated by

$$P_{SB} = P_D - P_L = \frac{1 - V_L^2}{2R_L}.$$
(7.199)

In this section DC energy recovery within one BB BPSK period will be discussed. The block diagram is given in Fig. 7.12. A biphase coding is applied and the PA is operated



Figure 7.12: DC BPSK energy recovery.

regularly in one period and acts as a rectifier in the other one. The normalized drain voltage is the same as for the broadband load. Also the resulting load voltage is the same and defined by (7.194). The input current  $I_D$  differs from the broadband load operation and is defined by

$$I_D = \frac{V_L}{R_L} = \frac{2d - 1}{R_L}.$$
(7.200)

The input power for both periods is given by

$$P_{D1} = \frac{V_{D1}I_Dd}{2}; \ P_{D2} = \frac{V_{D2}I_D(1-d)}{2}.$$
 (7.201)

Considering (7.195) and (7.200) yields

$$P_{D1} = \frac{V_L + V_L^2}{4R_L}; \ P_{D2} = \frac{-V_L + V_L^2}{4R_L}.$$
(7.202)

The resulting DC power consumptions are defined as

$$P_{DC1} = \begin{cases} P_{D1}/\eta_D & P_{D1} > 0\\ P_{D1}\eta_R & \text{else} \end{cases}, \ P_{DC2} = \begin{cases} P_{D2}/\eta_D & P_{D2} > 0\\ P_{D2}\eta_R & \text{else} \end{cases}.$$
 (7.203)

Calculating the total DC power considering (7.203) and only positive load voltage  $V_L$  results in

$$P_{DC} = \frac{P_{D1}}{\eta_D} + P_{D2}\eta_r.$$
 (7.204)

The resulting efficiency considering (7.196) and (7.202) is given by

$$\eta = \frac{P_L}{P_{DC}} = \frac{P_L}{P_{D1} + P_{D2}\eta_R\eta_D}\eta_D = \frac{2V_L}{1 + V_L - (1 - V_L)\eta_R\eta_D}\eta_D.$$
(7.205)

## 7.6 Chireix

In this section the general properties of the Chireix combiner for arbitrary load voltage and phase relationships will be derived.

#### 7.6.1 Quarterwave Chireix Outphasing

In Fig. 7.13 a general outphasing combiner is depicted.



Figure 7.13: Chireix combiner with shunt reactance.

The relation for the currents at the summation point can be defined as

$$I_L = -I_{L1} - I_{L2}. (7.206)$$

Furthermore the load voltage  $V_L$  results in

$$V_L = I_L Z_L = -(I_{L1} + I_{L2}) Z_L. (7.207)$$

The relations for the transmission lines with their characteristic impedance  $Z_1, Z_2$  and electrical length  $\theta_1, \theta_2$  are defined by

$$V_L = V_1 \cos(\theta_1) + j Z_1 I_1 \sin(\theta_1)$$
(7.208)

$$I_{L1} = jV_1 \frac{1}{Z_1} \sin(\theta_1) + I_1 \cos(\theta_1)$$
(7.209)

$$V_L = V_2 \cos(\theta_2) + j Z_2 I_2 \sin(\theta_2)$$
(7.210)

$$I_{L2} = jV_2 \frac{1}{Z_2} \sin(\theta_2) + I_2 \cos(\theta_2).$$
(7.211)

Setting (7.208) and (7.210) equal and resolving for  $I_2$  delivers

$$I_2 = -j \frac{V_1 \cos(\theta_1) - V_2 \cos(\theta_2) + j Z_1 I_1 \sin(\theta_1)}{Z_2 \sin(\theta_2)}.$$
(7.212)

Inserting (7.209) and (7.211) into (7.207) yields

$$-V_L = jV_1 \frac{Z_L}{Z_1} \sin(\theta_1) + I_1 Z_L \cos(\theta_1) + jV_2 \frac{Z_L}{Z_2} \sin(\theta_1) + I_2 Z_L \cos(\theta_2).$$
(7.213)

Considering (7.212) results in

$$-V_L = I_1 Z_L \left( \cos(\theta_1) + \frac{Z_1 \cos(\theta_2)}{Z_2 \sin(\theta_2)} \sin(\theta_1) \right) + j V_2 \frac{Z_L}{Z_2 \sin(\theta_2)}$$

$$+ V_1 \left( j \frac{Z_L}{Z_1} \sin(\theta_1) - j \frac{Z_L \cos(\theta_2)}{Z_2 \sin(\theta_2)} \cos(\theta_1) \right).$$

$$(7.214)$$

Setting (7.208) and (7.214) equal delivers

$$-I_{1}\left(jZ_{1}\sin(\theta_{1})+Z_{L}\cos(\theta_{1})+Z_{L}\frac{Z_{1}\cos(\theta_{2})}{Z_{2}\sin(\theta_{2})}\sin(\theta_{1})\right)$$

$$=V_{1}\left(j\frac{Z_{L}}{Z_{1}}\sin(\theta_{1})-j\frac{Z_{L}\cos(\theta_{2})}{Z_{2}\sin(\theta_{2})}\cos(\theta_{1})+\cos(\theta_{1})\right)+jV_{2}\frac{Z_{L}}{Z_{2}\sin(\theta_{2})}$$
(7.215)

Resolving for  $-I_1$  and considering  $Z_1 = -V_1/I_1$  yields

$$Z_{1} = \frac{V_{1} \left( Z_{L} \cos(\theta_{1}) + \left( j Z_{1} + Z_{L} \frac{Z_{1}}{Z_{2}} \cot(\theta_{2}) \right) \sin(\theta_{1}) \right)}{V_{1} \left( j \frac{Z_{L}}{Z_{1}} \sin(\theta_{1}) + \left( 1 - j \frac{Z_{L}}{Z_{2}} \cot(\theta_{2}) \right) \cos(\theta_{1}) \right) + j V_{2} \frac{Z_{L}}{Z_{2} \sin(\theta_{2})}}.$$
 (7.216)

Doing the same for the second branch gives

$$Z_{2} = \frac{V_{2} \left( Z_{L} \cos(\theta_{2}) + \left( j Z_{2} + Z_{L} \frac{Z_{2}}{Z_{1}} \cot(\theta_{1}) \right) \sin(\theta_{2}) \right)}{V_{2} \left( j \frac{Z_{L}}{Z_{2}} \sin(\theta_{2}) + \left( 1 - j \frac{Z_{L}}{Z_{1}} \cot(\theta_{1}) \right) \cos(\theta_{2}) \right) + j V_{1} \frac{Z_{L}}{Z_{1} \sin(\theta_{1})}}.$$
 (7.217)

Considering equal line impedances  $Z_1 = Z_2 = F Z_L$  and implying quarterwave lines  $\theta_1 = \pi/2, \theta_2 = \pi/2$  and a real load  $Z_L = R_L$  results in

$$Z_1 = \frac{F^2 R_L}{1 + \frac{V_2}{V_1}}, \ Z_2 = \frac{F^2 R_L}{1 + \frac{V_1}{V_2}}.$$
(7.218)

When considering  $V_1 = e^{j\phi}$  and  $V_2 = e^{-j\phi}$ , (7.218) simplifies to

$$Z_{1,2} = R_L \frac{F^2}{2} \left(1 \pm j \tan(\phi)\right).$$
(7.219)

For F = 2 the equation further reduces to

$$Z_{1,2} = 2R_L \left(1 \pm j \tan(\phi)\right).$$
(7.220)

The parallel resistance is given by

$$R_1 = R_2 = \frac{F^2}{2} R_L \left( 1 + \tan^2(\phi) \right).$$
(7.221)

The output power can be calculated by

$$P = P_1 + P_2 = \frac{V_1^2}{2R_1} + \frac{V_2^2}{2R_2} = \frac{1}{F^2 R_L \left(1 + \tan^2(\phi)\right)}$$
(7.222)

and the power back off ratio p can be calculated by

$$p = \frac{P_{BO}}{P_0} = \frac{\frac{1}{F^2 R_L (1 + \tan^2(\phi_{BO}))}}{\frac{1}{F^2 R_L (1 + \tan^2(0))}} = \frac{1}{1 + \tan^2(\phi_{BO})} = \cos^2(\phi_{BO}).$$
(7.223)

Rearranging to get the back off outphasing angle  $tan(\phi_{BO})$  results in

$$\tan(\phi_{BO}) = \sqrt{\frac{1-p}{p}}.$$
(7.224)

The parallel reactance at the back off level can be calculated by

$$jX_1(\phi_{BO}) = jF^2 R_L \left( \tan(\phi_{BO}) + \frac{1}{\tan(\phi_{BO})} \right).$$
(7.225)

Inserting (7.224) results in

$$jX_1(p) = j\frac{F^2}{2}R_L\left(\sqrt{\frac{1-p}{p}} + \sqrt{\frac{p}{1-p}}\right) = j\frac{F^2R_L}{2\sqrt{p-p^2}}.$$
(7.226)

To compensate this reactance a parallel reactance of the opposite sign is added. These reactances are also known as the Chireix reactances, respectively Chireix combiner. The required parallel compensation reactances result in

$$jX_{P1} = -jX_1(\phi_{BO}); \ jX_{P2} = jX_1(\phi_{BO}).$$
 (7.227)

The final Chireix input impedances  $Z_1$  and  $Z_2$  simplify to a parallel circuit and are given by

$$Z_1 = \frac{F^2}{2} R_L \frac{(\tan(\phi) - j)}{\sqrt{p - p^2} (1 + j \tan(\phi)) - j},$$
(7.228)

$$Z_2 = \frac{F^2}{2} R_L \frac{(\tan(\phi) + j)}{\sqrt{p - p^2} (1 - j \tan(\phi)) + j}.$$
(7.229)

For zero outphasing angle  $\phi=0$  the input impedances  $Z_1^C$  and  $Z_2^C$  simplify to

$$Z_1^C = \frac{F^2 R_L}{2 + j2\sqrt{p - p^2}}; \ Z_2^C = \frac{F^2 R_L}{2 - j2\sqrt{p - p^2}}.$$
 (7.230)

#### 7.6.2 Asymmetric Chireix Outphasing

A Chireix combiner can also be built by using transmission lines with different lengths according to  $\theta_1 = \pi/2 + \theta_c$  and  $\theta_2 = \pi/2 - \theta_c$ , as depicted in Fig. 7.14.



Figure 7.14: Asymmetric transmission line Chireix combiner.

Considering transmission lines with the same characteristic impedance defined by  $Z_1 = Z_2 = F_c Z_L$  and a real load  $Z_L = R_L$  results in

$$Z_{1} = R_{L} \frac{\cos(\theta_{1}) + (jF_{c} + \cot(\theta_{2}))\sin(\theta_{1})}{\left(j\frac{1}{F_{c}}\sin(\theta_{1}) + \left(1 - j\frac{1}{F_{c}}\cot(\theta_{2})\right)\cos(\theta_{1})\right) + j\frac{V_{2}}{V_{1}}\frac{1}{F_{c}\sin(\theta_{2})}}.$$
 (7.231)

Simplification results in

$$Z_1 = R_L \frac{F_c^2 \cos^2(\theta_c)}{1 + j\frac{F_c}{2}\sin(2\theta_c) + \frac{V_2}{V_1}}; \ Z_2 = R_L \frac{F_c^2 \cos^2(\theta_c)}{1 - j\frac{F_c}{2}\sin(2\theta_c) + \frac{V_1}{V_2}}.$$
 (7.232)

Considering  $V_1 = e^{j\phi}, V_2 = e^{-j\phi}$  delivers

$$Z_{1} = R_{L} \frac{F_{c}^{2} \cos^{2}(\theta_{c}) \left(1 + \cos(2\phi) - j \left(\frac{F_{c}}{2} \sin(2\theta_{c}) - \sin(2\phi)\right)\right)}{2 + 2\cos(2\phi) + \frac{F_{c}^{2}}{4} \sin^{2}(2\theta_{c}) - F_{c} \sin(2\theta_{c}) \sin(2\phi)},$$
(7.233)

$$Z_{2} = R_{L} \frac{F_{c}^{2} \cos^{2}(\theta_{c}) \left(1 + \cos(2\phi) + j\left(\frac{F_{c}}{2}\sin(2\theta_{c}) - \sin(2\phi)\right)\right)}{2 + 2\cos(2\phi) + \frac{F_{c}^{2}}{4}\sin^{2}(2\theta_{c}) - F_{c}\sin(2\theta_{c})\sin(2\phi)}.$$
(7.234)

For zero outphasing angle  $\phi = 0$  the equations simplify to

$$Z_1 = R_L F_c^2 \frac{\cos^2(\theta_c)}{4 + \frac{F_c^2}{4} \sin^2(2\theta_c)} \left(2 - j\frac{F_c}{2}\sin(2\theta_c)\right).$$
(7.235)

To get the same impedances as for the conventional approach with parallel compensation reactances the line impedance  $F_c R_L$  and the Chireix compensation angle  $\theta_c$  can be calculated by setting (7.230) and (7.235) equal, which delivers the following two equations

$$\frac{F_c^2 \cos^2(\theta_c)}{4 + \frac{F_c^2}{4} \sin^2(2\theta_c)} = \frac{F^2}{4(p - p^2 + 1)},$$
(7.236)

$$\frac{F_c^2 \cos^2(\theta_c)}{4 + \frac{F_c^2}{4} \sin^2(2\theta_c)} \left(\frac{F_c}{2} \sin(2\theta_c)\right) = \frac{F^2}{2} \frac{\sqrt{p - p^2}}{p - p^2 + 1}.$$
(7.237)

Inserting (7.236) into (7.237) and resolving for  $\theta_c$  yields

$$\sin(2\theta_c) = \frac{4}{F_c}\sqrt{p - p^2}.$$
(7.238)

Considering (7.236) and (7.238) delivers

$$F_c = \frac{F}{\cos(\theta_c)}.\tag{7.239}$$

Inserting this into (7.238) results in

$$\sin(\theta_c) = \frac{2}{F}\sqrt{p-p^2}.$$
(7.240)

Inserting now (7.240) into (7.239) yields the design equation for  $F_c$ 

$$F_c = \frac{F^2}{\sqrt{F^2 - 4(p - p^2)}}.$$
(7.241)

For F = 2 the equation simplifies to

$$F_c = \frac{2}{\sqrt{1 - p + p^2}}.$$
(7.242)

#### 7.6.3 Error Analysis

Any imperfection in the supply voltages and/or matching network makes it impossible to achieve low output voltages. The load voltage is defined by (7.207) and thus for zero load voltage the following condition has to be satisfied

$$V_L = -(I_{L1} + I_{L2})Z_L = 0. (7.243)$$

Considering  $I_{L1} = I_{M1}e^{j\phi}$ ,  $I_{L2} = I_{M2}e^{-j\phi}$  and defining  $I_{M2} = I_{M1} + \Delta I$  results in

$$V_L = -(2I_{M1}\cos(\phi) + \Delta I e^{-j\phi})Z_L.$$
(7.244)

The first part of the equation  $I_{M1}\cos(\phi)$  will become zero for an angle of  $\phi = \pi/2$ , but the second part has a constant amplitude and only changes in phase. Therefore, the resulting minimum and maximum amplitudes are given by

$$V_{LMIN} = -j\Delta I Z_L; \ \phi = \frac{\pi}{2},$$
 (7.245)

$$V_{LMAX} = -(2I_{M1} + \Delta I) Z_L; \ \phi = 0.$$
(7.246)

Normalizing the minimum load voltage by the maximum load voltage delivers

$$V_{MIN} = \frac{V_{L\,MIN}}{V_{L\,MAX}} = j \frac{\Delta I}{2I_{M1} + \Delta I} = j \frac{I_{M2} - I_{M1}}{I_{M1} + I_{M2}}.$$
(7.247)

The remaining load voltage phasor is orthogonal to the phasor controlled by the outphasing and can therefore not easily be reduced. Considering an outphasing combiner using two equal quarter wave lines results in the following requirement

$$V_{MIN} = j \frac{\Delta V}{2V_{M1} + \Delta V} \approx j \frac{\Delta V}{2V}.$$
(7.248)

The resulting dynamic range for supply voltage mismatch can be approximated by

$$DR = 20\log_{10}\left(\frac{\Delta V}{2V}\right). \tag{7.249}$$

The current mismatch can also be caused by a mismatch of the transmission line impedance or by a phase difference, which will always be the case due to manufacturing tolerances. The impact of different transmission line impedances can be expressed by

$$V_{MIN} = j \frac{Z_{L1} - Z_{L2}}{Z_{L1} + Z_{L2}} = j \frac{\Delta Z}{2Z + \Delta Z} \approx j \frac{\Delta Z}{2Z},$$
(7.250)

$$DR = 20\log 10\left(\frac{\Delta Z}{2Z}\right). \tag{7.251}$$

Thus for a single frequency an output matching imperfection might be compensated by modifying the supply voltages accordingly, which is however not desirable in practice. Another possibility is to use a three (or more) way outphasing system. The only constraint for reaching zero output power is the requirement that the sum of the magnitudes of the smallest vectors is larger than the magnitude of the largest vector.



Figure 7.15: Voltage and current waveforms and equivalent circuit for (a) full and (b) partial PWM modulation.

### 7.7 Equivalent Load Impedance

#### 7.7.1 Full PWM Modulation

In this section the equivalent load impedance for PWM modulation with full modulation depth (between zero and one) will be derived. For the derivation a series resonator connected via a transmission line with a characteristic impedance  $Z_0$  and electrical length  $\theta$ , as depicted in Fig. 7.15, will be assumed. Later the calculation will be expanded to a general case, without any restriction on circuit topology.

For the derivations it is assumed that the filter has an infinite Q and the ratio  $f_{PWM}/f_c$ approaches zero. The ratio of average voltage and current over one cycle is equal to the impedance at carrier frequency

$$Z_c = Z_{IN}(\omega_c) = \frac{V(\omega_c)}{I(\omega_c)}.$$
(7.252)

The average voltage  $V(\omega_c)$  relates to the duty cycle d as  $V(\omega_c) = V_{ON}d$ , where the average current is defined by

$$I(\omega_c) = I_{ON}d + I_{OFF}(1-d).$$
(7.253)

This results in an impedance at carrier frequency given by

$$Z_c = \frac{V_{ON}d}{I_{ON}d + I_{OFF}(1-d)}.$$
(7.254)

The lossless transmission line TL with the properties  $Z_0$  and  $\theta$  delivers 2 equations for each condition (ON and OFF).

$$V_{ON} = V_{F,ON} \cos(\theta) + j I_{F,ON} Z_0 \sin(\theta)$$
(7.255)

$$I_{ON} = j \frac{1}{Z_0} V_{F,ON} \sin(\theta) + I_{F,ON} \cos(\theta)$$

$$(7.256)$$

$$V_{OFF} = V_{F,OFF}\cos(\theta) + jI_{F,OFF}Z_0\sin(\theta)$$
(7.257)

$$I_{OFF} = j \frac{1}{Z_0} V_{F,OFF} \sin(\theta) + I_{F,OFF} \cos(\theta)$$
(7.258)

Solving (7.255) for  $V_{F,ON}$  assuming the normalized voltage  $V_{ON}$  being 1 yields

$$V_{F,ON} = \frac{1 - jI_{F,ON}Z_0\sin(\theta)}{\cos(\theta)}.$$
 (7.259)

Combining this with (7.256) and solving for  $I_{F,ON}$  gives

$$I_{F,ON} = I_{ON}\cos(\theta) - j\frac{1}{Z_0}\sin(\theta).$$
(7.260)

Resolving (7.257), taking  $V_{OFF} = 0$  into account, results in

$$V_{F,OFF} = -jZ_0 \frac{\sin(\theta)}{\cos(\theta)} I_{F,OFF}.$$
(7.261)

Based on the same assumptions (7.258) is given by

$$I_{F,OFF} = I_{OFF} \cos(\theta). \tag{7.262}$$

The current through the series filter is continuous and thus  $I_{F,ON} = I_{F,OFF}$ . This property allows solving for  $I_{OFF}$ 

$$I_{OFF} = I_{ON} - j \frac{\sin(\theta)}{Z_0 \cos(\theta)}.$$
(7.263)

Now the impedance  $Z_S$  is introduced, which can be seen as the input impedance at the carrier frequency, if the series resonator provides an open circuit (low duty cycles). In other words the impedance  $Z_S$  also marks the intersection of the unit circle of the smith chart and the symmetry axis of the resonance locus. Its graphical definition is depicted in Fig. 7.16 and its value is given by

$$Z_S = -jZ_0\cot(\theta). \tag{7.264}$$



Figure 7.16: Definition of  $Z_C$  and  $Z_S$ .

The phase angle  $(\theta)$  relates to the detuning angle from the series resonator locus  $(\Phi)$  as  $\Phi = 180^{\circ} - \theta$ . Based on (7.264), (7.263) can be rearranged to

$$I_{OFF} = I_{ON} - \frac{1}{Z_S}.$$
 (7.265)

Extracting  $I_{ON}$  considering (7.254) and  $V_{ON} = 1$  yields

$$I_{ON} = \frac{d}{Z_C} + \frac{1-d}{Z_S}.$$
(7.266)

Introducing the weighted load impedance

$$Z_P = \frac{Z_C}{d} \tag{7.267}$$

and defining the weighted reactance

$$Z_{PR} = \frac{Z_S}{1-d} \tag{7.268}$$

delivers

$$I_{ON} = \frac{1}{Z_P} + \frac{1}{Z_{PR}}.$$
(7.269)

The equivalent load impedance  $Z_{ON}$  during the ON period can be calculated by

$$Z_{ON} = \frac{V_{ON}}{I_{ON}} = \frac{Z_P Z_{PR}}{Z_P + Z_{PR}},$$
(7.270)

which is a simple parallel circuit of the weighted load impedance  $Z_P$  and the weighted reactance  $Z_{PR}$ .

The current during the OFF period can be calculated considering (7.263) and (7.264) and results in

$$I_{OFF} = d\left(\frac{1}{Z_C} - \frac{1}{Z_S}\right). \tag{7.271}$$

The equivalent load impedance during the OFF period depends on the voltage as

$$Z_{OFF} = \frac{V_{OFF}}{I_{OFF}} = 0, \qquad (7.272)$$

as the input voltage during the OFF period is zero.

#### 7.7.2 Partial PWM Modulation

In the previous section the load impedances for a fully PWM modulated signal have been derived. In this section the equivalent load impedances of a constant signal  $(V_2)$  also during the OFF period imposed by a PWM modulated signal  $(V_1 - V_2)$  will be calculated. In Fig. 7.15(b) the definition of voltages and currents for this operation are depicted. To simplify the calculations the normalized reference voltage  $V_{REF} = V_2/V_1$  is introduced. The constant part corresponds to  $V_{REF}$  and the modulated part to  $1 - V_{REF}$ . It is important to note that  $V_1$  and  $V_2$  and also  $V_{REF}$  are complex numbers in general.

The load current during the first period depends on the constant part  $(V_{REF}/Z_C)$  and on a modulated part (7.269).

$$I_1 = \frac{V_{REF}}{Z_C} + \frac{1 - V_{REF}}{Z_{ON}}$$
(7.273)

This results in

$$Z_1 = \frac{V_1}{I_1} = \frac{Z_{ON} Z_C}{Z_{ON} V_{REF} + Z_C (1 - V_{REF})}$$
(7.274)

for the impedance during the first period. Similarly the current during the second period can be calculated by

$$I_2 = \frac{1 - V_{REF}}{Z_C} + I_{OFF}(1 - V_{REF}).$$
(7.275)

The resulting load impedance  $Z_2$  during the second period is given by

$$Z_2 = \frac{V_2}{I_2} = \frac{V_{REF}}{\frac{V_{REF}}{Z_C} + \left(\frac{1}{Z_C} - \frac{1}{Z_S}\right) d\left(1 - V_{REF}\right)}$$
(7.276)

## 7.8 Power Loss Band Pass Filters

In this section the insertion loss of band pass filters with respect to the loaded/unloaded Q factor will be derived. The basic circuit diagrams of a parallel and a series resonator with losses are provided in Fig. 7.17.



Figure 7.17: Losses in band pass filters.

The internal Q-factor of the parallel resonator depicted in Fig. 7.17(a) can be calculated using

$$Q_0 = R_P \sqrt{\frac{C}{L}},\tag{7.277}$$

while the external Q-factor  $Q_L$  also depends on the load resistor  $R_L$  and is given by

$$Q_L = \frac{R_P R_L}{R_P + R_L} \sqrt{\frac{C}{L}}.$$
(7.278)

For the calculation of the efficiency of the resonator the power lost in the parallel resistor  $R_P$  and the power fed to the load  $R_L$ 

$$P_P = \frac{V^2}{R_P}, \ P_L = \frac{V^2}{R_L}$$
(7.279)

are required. The efficiency can be calculated by

$$\eta_P = \frac{P_L}{P_P + P_L} = \frac{R_P}{R_L + R_P}.$$
(7.280)

Resolving (7.277) for the parallel resistor  $R_P$  delivers

$$R_P = \frac{Q_0}{\sqrt{\frac{C}{L}}}.\tag{7.281}$$

Inserting this into (7.278) and resolving for  $R_L$  results in

$$R_L = \frac{Q_0 Q_L}{\sqrt{\frac{C}{L}}(Q_0 - Q_L)}.$$
(7.282)

Using (7.281) and (7.282) to express (7.280) in terms of the loaded  $\rm Q_L$  and unloaded  $\rm Q_0$  Q-factor yields

$$\eta_P = 1 - \frac{Q_L}{Q_0}.$$
(7.283)

Considering the series circuit in Fig. 7.17(b) the loaded and unloaded Q factor can be expressed as

$$Q_0 = \frac{1}{R_S} \sqrt{\frac{L}{C}} \tag{7.284}$$

and

$$Q_L = \frac{1}{R_S + R_L} \sqrt{\frac{L}{C}}.$$
(7.285)

The power in the series resistor and the load can be calculated by

$$P_S = I^2 R_S, \ P_L = I^2 R_L. \tag{7.286}$$

The efficiency is given by

$$\eta_S = \frac{P_L}{P_S + P_L} = \frac{R_L}{R_L + R_S}.$$
(7.287)

Resolving (7.284) for  $R_S$  delivers

$$R_S = \frac{1}{Q_0} \sqrt{\frac{L}{C}}.$$
(7.288)

Taking this into consideration when resolving (7.285) results in

$$R_L = \left(\frac{1}{Q_L} - \frac{1}{Q_0}\right) \sqrt{\frac{L}{C}}.$$
(7.289)

Taking (7.288) and (7.289) into account when rearranging (7.280) yields

$$\eta_S = 1 - \frac{Q_L}{Q_0},\tag{7.290}$$

which is the same result as for the parallel resonator (7.283).

## 7.9 Transistor Parameters



Figure 7.18: GaN transistor and GaN diode IV curves.

## 7.10 CPE Implementation



Figure 7.19: CPE overview.



Figure 7.20: CPE iterative interpolation.

(a) Low edge timing

$l_{1L}$	$l_{0L}$	$t_{l1}$	$t_{l2}$	$p_{2l}$
0	0	0	0	0
0	1	0	$l_1$	0
1	0	$l_2$	0	0
1	1	$l_1$	$l_2$	0

(b)	High	edge	timing	g
(b)	High	edge	timing	

$h_{2L}$	$h_{1L}$	$h_{0L}$	t <sub>h1</sub>	$t_{h2}$	$p_{2h}$
0	0	0	0	$2\pi$	0
0	0	1	0	$2\pi$	0
0	1	0	h <sub>1</sub>	$2\pi$	0
0	1	1	0	h <sub>1</sub>	1
1	0	0	h <sub>2</sub>	$2\pi$	0
1	0	1	0	$2\pi$	0
1	1	0	h <sub>1</sub>	h <sub>2</sub>	1
1	1	1	0	h <sub>1</sub>	1

Table 7.1: Logic table edge timing selection.





(a) Low edge timing selection

(b) High edge timing selection



(c) Signal selection

Figure 7.21: CPE minimum error signal selection.

# List of Publications

- [S1] D. Seebacher, W. Bösch, M. Gadringer, P. Singerl, and C. Schuberth. High Frequency Class-DE Push/Pull Power Amplifier Utilizing a Parallel Compensation Inductance. In *Microelectronic Systems Symposium (MESS)*, May 2014.
- [S2] D. Seebacher, W. Bösch, P. Singerl, and C. Schuberth. The Impact of the Q-Factor of the Parasitic Capacitances of RF Transistors on their Load Modulation Capabilities. In Conf. on Ph.D. Research in Microelectronics and Electronics (PRIME), July 2014.
- [S3] D. Seebacher, W. Bösch, P. Singerl, and C. Schuberth. Efficiency Enhancement of Burst Mode Transmitters by RF Energy Recovery. In Conf. on Ph.D. Research in Microelectronics and Electronics (PRIME), pages 221–224, June 2013.
- [S4] D. Seebacher, P. Singerl, C. Schuberth, W. Bösch, and Gadringer M. Equivalent Load Impedance of Pulse Width Modulation Excited Band Pass Filters. accepted Int. Journal of RF and Microwave Computer Aided Engineering, July 2014.
- [S5] D. Seebacher, C. Schuberth, and W. Bösch. The Impact of Coupling on the Reflections of Cavity Resonators. In Int. Microwave and RF Conf. (IMARC), Dec 2013.
- [S6] D. Seebacher, W. Bösch, P. Singerl, and C. Schuberth. Highly Efficient Carrier Bursting RF Transmitter Employing Direct Band Pass Filter Connection. In *European Microwave Conf. (EuMC)*, pages 692–695, Oct 2013.
- [S7] D. Seebacher, P. Singerl, C. Schuberth, and M. Mataln. System and Method for a Power Amplifier. US Patent Application, 2012.
- [S8] D. Seebacher, C. Schuberth, P. Singerl, M. Gadringer, and W. Bösch. A Common Drain Parallel Amplifier in GaN Using Baseband PWM and Direct Filter Connection for Efficiency Enhancement. Int. Workshop on Integrated Nonlinear Microwave and Millimetre-wave Circuits (INMMIC), Apr 2014.
- [S9] D. Seebacher, W. Bösch, P. Singerl, and C. Schuberth. An Integrated Switchable Cascode RF Power Amplifier in GaN Technology. In Int. Microwave and RF Conf. (IMARC), Dec 2013.
- [S10] D. Seebacher, P. Singerl, C. Schuberth, and M. Mataln. System and Method for a Cascoded Amplifier. US Patent Application, 2012.
- [S11] D. Seebacher, C. Schuberth, P. Singerl, M. Gadringer, and W. Bösch. A Source Modulated Amplifier in GaN Designed for Baseband PWM and Direct Filter Connection. In Int. Black Sea Conference on Communications and Networking (BlackSeaCom), May 2014.
- [S12] D. Seebacher, P. Singerl, C. Schuberth, F. Dielacher, P. Reynaert, and W. Bösch. Reduction of Aliasing Effects of RF-PWM Modulated Signals by Cross Point Estimation. accepted IEEE Trans. on Circuits and Systems I: Regular Papers, June 2014.

[S13] D. Seebacher, P. Singerl, C. Schuberth, and M. Mataln. Methods and Devices for Providing Cross Point Information. US Patent Application, 2013.

# List of References

- ITU. Key 2006-2013 ict data for the world. http://www.itu.int/en/ITU-D/Statistics/Documents/statistics/2013/ITU\_Key\_2005-2013\_ICT\_data.xls, Feb. 2014.
- [2] Juha Rapeli. UMTS-a Path to 3rd generation Mobile Communications of the 21st Century. In *IEEE International Conference on Personal Wireless Communications*, pages 58–67, 1996.
- [3] P.C. Mason, J. M. Cullen, and N.C. Lobley. UMTS architectures. In *IEE Colloquium on Mobile Communications Towards the Next Millenium and Beyond*, pages 4/1–4/11, May 1996.
- [4] 3GPP. Technologies. http://www.3gpp.org/technologies, Feb. 2014.
- [5] C. Gessner, A. Roessler, and M. Kottkamp. UMTS Long Term Evolution (LTE) -Technology Introduction, Application Note. Technical Report 1MA111\_3E, Rohde and Schwarz, July 2012.
- [6] A. Roessler and M. Kottkamp. LTE- Advanced (3GPP Rel.11) Technology Introduction White Paper. Technical Report 7.2013 – 1MA232\_1E, Rohde and Schwarz, 2013.
- [7] Tomas Edler. Green Base Stations How to Minimize CO2 Emission in Operator Networks. In *Ericsson seminar, Bath Base Station Conf.*, 2008.
- [8] F. Ghannouchi. Toward sustainable green communication networks? In *GigaHertz Centre Annual Day 2013.* Chalmers University of Technology, Nov. 2013.
- [9] C Khirallah, JS Thompson, and H Rashvand. Energy and Cost Impacts of Relay and Femtocell Deployments in Long-Term-Evolution Advanced. *IET communications*, 5(18):2617–2628, 2011.
- [10] S.C. Cripps. RF Power Amplifiers for Wireless Communications. Artech House, 2006.
- [11] Steve C Cripps, Paul J Tasker, Alan L Clarke, Jonathan Lees, and Johannes Benedikt. On the Continuity of High Efficiency Modes in Linear RF Power Amplifiers. *IEEE Microwave and Wireless Components Letters*, 19(10):665–667, 2009.
- [12] Peter Wright, Jonathan Lees, Johannes Benedikt, Paul J Tasker, and Steve C Cripps. A Methodology for Realizing High Efficiency Class-J in a Linear and Broadband PA. *IEEE Transactions on Microwave Theory and Techniques*, 57(12):3196–3204, 2009.
- [13] Peter Wright, J Lees, PJ Tasker, J Benedikt, and Steve C Cripps. An Efficient, Linear, Broadband Class-J-Mode PA Realised Using RF Waveform Engineering. In *IEEE MTT-S International Microwave Symposium Digest (IMS)*, pages 653–656. IEEE, 2009.

- [14] Dristy Rahul Parveg, Peter Singerl, Andreas Wiesbauer, Hossein Mashad Nemati, and Christian Fager. A Broadband, Efficient, Overdriven Class-J RF Power Amplifier for Burst Mode Operation. In *European Microwave Conference (EuMC)*, pages 1666– 1669. IEEE, 2010.
- [15] Junghwan Moon, Jungjoon Kim, and Bumman Kim. Investigation of a Class-J Power Amplifier with a Nonlinear Cout for Optimized Operation. *IEEE Trans. Microw. Theory Tech*, 58(11):2800–2811, 2010.
- [16] Neal Tuffy, Anding Zhu, and Thomas J Brazil. Class-J RF Power Amplifier With Wideband Harmonic Suppression. In *IEEE MTT-S International Microwave Sympo*sium Digest (IMS), pages 1–4. IEEE, 2011.
- [17] Christer M Andersson, David Gustafsson, Koji Yamanaka, Eigo Kuwata, Hiroshi Otsuka, Masatoshi Nakayama, Yoshihito Hirano, Iltcho Angelov, Christian Fager, and Niklas Rorsman. Theory and Design of Class-J Power Amplifiers With Dynamic Load Modulation. *IEEE Transactions on Microwave Theory and Techniques*, 60(12):3778–3786, 2012.
- [18] F. Raab. Class-F Power Amplifiers With Maximally Flat Waveforms. IEEE Transactions on Microwave Theory and Techniques, 45(11):2007–2012, 1997.
- [19] F. Raab. Maximum Efficiency and Output of class-F Power Amplifiers. IEEE Transactions on Microwave Theory and Techniques, 49(6):1162–1166, 2001.
- [20] Y. Y. Woo, Y. Yang, and B. Kim. Analysis and Experiments for High-Efficiency Class-F and Inverse Class-F Power Amplifiers. *IEEE Transactions on Microwave Theory and Techniques*, 54(5):1969–1974, 2006.
- [21] Ji-Yeon Kim, Duk-Soo Oh, and Jong-Heon Kim. Design of a Harmonically Tuned class-F Power Amplifier. In Asia-Pacific Microwave Conference (APMC), pages 1–4, 2007.
- [22] D. Schmelzer and S.I. Long. A GaN HEMT Class F Amplifier at 2 GHz With >80% PAE. *IEEE Journal of Solid-State Circuits*, 42(10):2130–2136, 2007.
- [23] Y. Y. Woo, Y. Yang, I. Kim, and B. Kim. Efficiency Comparison Between Highly Efficient Class-F and Inverse Class-F Power Amplifiers. *IEEE Microwave Magazine*, 8(3):100–110, 2007.
- [24] Andrei Grebennikov. Load Network Design Technique for Class F and Inverse Class F PAs. *High Frequency Electronics*, pages 58–76, 2011.
- [25] V. Carrubba, J. Lees, J. Benedikt, P.J. Tasker, and S.C. Cripps. A Novel Highly Efficient Broadband Continuous Class-F RFPA Delivering 74% Average Efficiency for an Octave Bandwidth. In *IEEE MTT-S International Microwave Symposium Digest (IMS)*, pages 1–1, 2011.
- [26] Neal Tuffy, Lei Guan, Anding Zhu, and T. J. Brazil. A Simplified Broadband Design Methodology for Linearized High-Efficiency Continuous Class-F Power Amplifiers. *IEEE Transactions on Microwave Theory and Techniques*, 60(6):1952–1963, 2012.
- [27] Bernhard Ingruber, Josef Baumgartner, Dieter Snely, Martin Wachutka, Gottfried Magerl, and Felix A. Petz. Rectangularly Driven class-A Harmonic-Control Amplifier. *IEEE Transactions on Microwave Theory and Techniques*, 46(11):1667–1672, 1998.

- [28] P. Saad, H.M. Nemati, M. Thorsell, K. Andersson, and C. Fager. An Inverse Class-F GaN HEMT Power Amplifier With 78% PAE at 3.5 GHz. In *European Microwave Conference (EuMC)*, pages 496–499, 2009.
- [29] M. Roberg, J. Hoversten, and Z. Popovic. GaN HEMT PA With Over 84% Power Added Efficiency. *Electronics Letters*, 46(23):1553–1554, 2010.
- [30] Peter Singerl, Christian Fager, Ziming Wang, Christian Schuberth, and Franz Dielacher. A Highly Efficient 1-GHz, 15-W Power Amplifier Design based on a 50-V LDMOS Transistor. In *IEEE MTT-S Int. Microwave Symp. Digest (IMS)*, pages 1102–1105. IEEE, 2010.
- [31] Christian Friesicke and Arne F. Jacob. Mode Continua for Inverse Class-F RF Power Amplifiers. In German Microwave Conference (GeMIC), pages 1–4, 2011.
- [32] J. Moon, S. Jee, J. Kim, J. Son, S. Kim, J. Lee, S. Kim, and B. Kim. Investigation of a Class-F<sup>-1</sup> Power Amplifier With a Nonlinear Output Capacitor. In *European Microwave Integrated Circuits Conference (EuMIC)*, pages 124–127, 2011.
- [33] Hidenori Kobayashi, Jeffrey M Hinrichs, and Peter M Asbeck. Current-Mode Class-D Power Amplifiers for High-Efficiency RF Applications. *IEEE Transactions on Microwave Theory and Techniques*, 49(12):2480–2485, 2001.
- [34] Ulf Gustavsson. Design of an Inverse Class D Amplifier Using GaN-HEMT Technology. MSc, Orebro University, 2006.
- [35] Anthony Long, Jingshi Yao, and Stephen I Long. A 13 W Current Mode Class D High Efficiency 1 GHz Power Amplifier. In *Midwest Symposium on Circuits and Systems*, volume 1, pages I–33. IEEE, 2002.
- [36] Tsai-Pi Hung, Andre G Metzger, Peter J Zampardi, Masaya Iwamoto, and Peter M Asbeck. Design of High-Efficiency Current-Mode Class-D Amplifiers for Wireless Handsets. *IEEE Transactions on Microwave Theory and Techniques*, 53(1):144–151, 2005.
- [37] Christian Schuberth, Peter Singerl, Holger Arthaber, Michael Gadringer, and Gottfried Magerl. Design of a Current Mode Class-D RF Amplifier Using Load Pull Techniques. In *IEEE MTT-S International Microwave Symposium Digest (IMS)*, pages 1521–1524. IEEE, 2009.
- [38] C. Schuberth, P. Singerl, M.E. Gadringer, H. Arthaber, A. Wiesbauer, and G. Magerl. Highly Efficient Switched-Mode Transmitter Using a Current Mode Class-D RF Amplifier. Int. Journal of RF and Microwave Computer-Aided Engineering, 20(4):446– 457, 2010.
- [39] Jonas Fritzin, Christer Svensson, and Atila Alvandpour. A Class-D Outphasing RF Amplifier With Harmonic Suppression in 90nm CMOS. In *Proceedings of the ESS-CIRC*, pages 310–313. IEEE, 2010.
- [40] Tsai-Pi Hung, David K Choi, Lawrence E Larson, and Peter M Asbeck. CMOS Outphasing Class-D Amplifier with Chireix Combiner. *IEEE Microwave and Wireless Components Letters*, 17(8):619–621, 2007.
- [41] Cijvat Pieternella. Two 130nm CMOS Class-D RF Power Amplifiers Suitable for Polar Transmitter Architectures. International Conference on Solid-State and Integrated-Circuit Technology, 2008.

- [42] Song Lin and Aly E Fathy. Development of a Wideband Highly Efficient GaN Vmcd VHF/UHF Power Amplifier. Progress In Electromagnetics Research C, 19:135–147, 2011.
- [43] S. Maroldt, R. Kiefer, S. Mueller, W. Benz, R. Quay, and O. Ambacher. High Efficiency Digital GaN MMIC Power Amplifiers for Future Switch-Mode Based Mobile Communication Systems. In *Compound Semiconductor Integrated Circuit Symposium* (CISC), pages 1–4, 2009.
- [44] A. Wentzel, Chafik Meliani, and W. Heinrich. RF Class-S Power Amplifiers: State-ofthe-Art Results and Potential. In *International Microwave Symposium Digest (IMS)*, pages 812–815, 2010.
- [45] W. Heinrich, A. Wentzel, and Chafik Meliani. Advanced Switch-Mode Concepts Using GaN: The Class-S Amplifier. In International Conference on Microwave Radar and Wireless Communications (MIKON), pages 1–6, 2010.
- [46] Dhamia Al-Mozani, Andreas Wentzel, Chafik Meliani, and Wolfgang Heinrich. A 900 MHz Voltage-Mode Class-S Power Amplifier. In *German Microwave Conference* (*GeMiC*), pages 1–4, 2012.
- [47] Shinichi Hori, A. Wentzel, Makoto Hayakawa, W. Heinrich, and Kazuaki Kunihiro. A Watt-Class Digital Transmitter with a Voltage-Mode Class-S Power Amplifier and an Envelope Delta-Sigma Modulator for 450 MHz Band. In *Compound Semiconductor Integrated Circuit Symposium (CSICS)*, pages 1–4, 2012.
- [48] A. Wentzel, Chafik Meliani, Georg Fischer, and W. Heinrich. An 8W GaN-Based H-Bridge Class-D PA for the 900 MHz Band Enabling Ternary Coding. In *IEEE MTT-S International Microwave Symposium Digest (IMS)*, pages 1–3, 2012.
- [49] A. Wentzel, Sergui Chevtchenko, Paul Kurpas, and W. Heinrich. A Dual-Band Voltage-Mode Class-D PA for 0.8/1.8 GHz Applications. In *International Microwave Symposium Digest (IMS)*, pages 1–4, 2013.
- [50] A. Wentzel and W. Heinrich. A GaN Voltage-Mode class-D MMIC With Improved Overall Efficiency for Future RRH Applications. In *European Microwave Conference* (*EuMC*), pages 549–552, 2013.
- [51] DC Hamill. Impedance Plane Analysis of Class DE Amplifier. *Electronics Letters*, 30(23):1905–1906, 1994.
- [52] S.A. El-Hamamsy. Design of High-Efficiency RF Class-D Power Amplifier. IEEE Trans. on Power Electronics, 9(3):297–308, 1994.
- [53] Marian K Kazimierczuk and Wojciech A Tabisz. Class CE High-Efficiency Tuned Power Amplifier. *IEEE Transactions on Circuits and Systems*, 36(3):421–428, 1989.
- [54] MK Kazimierczuk and W. Szaraniev. Class-D Zero-Voltage-Switching Inverter With Only One Shunt Capacitor. *IEE Proceedings B: Electric Power Applications*, 139(5):449–456, 1992.
- [55] H. Koizumi, T. Suetsugu, M. Fujii, K. Shinoda, S. Mori, and K. Iked. Class DE High-Efficiency Tuned Power Amplifier. *IEEE Trans. Circuits Syst. I*", 43(1):51–60, 1996.

- [56] D. Czarkowski and MK Kazmierczuk. ZVS Class D Series Resonant Inverter Discrete-Time State-Space Simulation and Experimental Results. *IEEE Trans. on Circuits and Systems I: Fundamental Theory and Applications*, 45(11):1141–1147, 1998.
- [57] M. Albulet. An Exact Analysis of Class-DE Amplifier at Any Output Q. IEEE Trans. on Circuits and Systems I: Fundamental Theory and Applications, 46(10):1228–1239, 1999.
- [58] Andrei Grebennikov. High-Efficiency Class-FE Tuned Power Amplifiers. IEEE Transactions on Circuits and Systems I: Regular Papers, 55(10):3284–3292, 2008.
- [59] H. Sekiya, M. Shinsaku, and I. Sasase. Exact Analysis of Class DE Amplifier with FM and PWM Control. *IEICE Transactions on Communications*, 86(10):3082–3093, 2003.
- [60] Ian Douglas De Vries. *High Power and High Frequency Class-DE Inverters*. PhD thesis, University of Cape Town, 1999.
- [61] H. Seykia, S. Oshikawa, and T. Yahagi. Novel Design Procedure for Class DE Amplifier. Journal of Circuits, Systems, and Computers, 17(02):191–209, 2008.
- [62] H. Sekiya, T. Watanabe, T. Suetsugu, and M.K. Kazimierczuk. Analysis and Design of Class DE Amplifier With Nonlinear Shunt Capacitances. *IEEE Trans. on Circuits* and Systems I: Regular Papers, 56(10):2362–2371, 2009.
- [63] H. Sekiya, N. Sagawa, and M.K. Kazimierczuk. Analysis of Class DE Amplifier With Nonlinear Shunt Capacitances at Any Grading Coefficient for High Q and 25% Duty Ratio. *IEEE Transactions on Power Electronics*, 25(4):924–932, 2010.
- [64] Mury Thian and Vincent F Fusco. Analysis and Design of Class-E3F and Transmission-Line Class-E3F2 Power Amplifiers. *IEEE Transactions on Circuits and* Systems I: Regular Papers, 58(5):902, 2011.
- [65] Gerald Dean Ewing. High-Efficiency Radio-Frequency Power Amplifiers. PhD thesis, Oregon State Unicersity, Corvallis, Oregon, June 1964.
- [66] Nathan O Sokal and Alan D Sokal. Class E a New Class of High-Efficiency Tuned Single-Ended Switching Power Amplifiers. *IEEE Journal of Solid-State Circuits*, 10(3):168–176, 1975.
- [67] Frederick Raab. Idealized Operation of the Class E Tuned Power Amplifier. IEEE Transactions on Circuits and Systems, 24(12):725–735, 1977.
- [68] M Kazimierczuk and Krzysztof Puczko. Exact Analysis of Class E Tuned Power Amplifier at Any Q and Switch Duty Cycle. *IEEE Transactions on Circuits and Systems*, 34(2):149–159, 1987.
- [69] John F Davis and David B Rutledge. A Low-Cost Class-E Power Amplifier With Sine-Wave Drive. In *IEEE MTT-S International Microwave Symposium Digest (IMS)*, volume 2, pages 1113–1116. IEEE, 1998.
- [70] Andrei V Grebennikov and Herbert Jaeger. Class E With Parallel Circuit-A New Challenge for High-Efficiency RF and Microwave Power Amplifiers. In *IEEE MTT-S International Microwave Symposium Digest*, volume 3, pages 1627–1630. IEEE, 2002.

- [71] Nathan O Sokal. Class-E High-Efficiency RF/Microwave Power Amplifiers: Principles of Operation, Design Procedures, and Experimental Verification. In Analog Circuit Design, pages 269–301. Springer, 2003.
- [72] M Mikolajewski and Juliusz Modzelewski. Effects of the Sine-Wave Drive in Class-E Tuned Power Amplifiers. In International Conference on Microwaves, Radar & Wireless Communications, pages 565–568. IEEE, 2006.
- [73] Andreas Adahl and Herbert Zirath. An 1 GHz Class E LDMOS Power Amplifier. In European Microwave Conference, pages 285–288. IEEE, 2003.
- [74] Mury Thian and Vincent Fusco. Amplitude-Modulation Characteristics of Power-Combining Class-E Amplifier With Finite Choke. *IET microwaves, antennas & prop*agation, 5(10):1195–1202, 2011.
- [75] FM Al-Raie. Design of Input Matching Networks for Class-E RF Power Amplifiers. *High Frequency Electronics*, pages 40–48, 2011.
- [76] David A Calvillo-Cortes, Leo CN de Vreede, and Michel de Langen. A Compact and Power-Scalable 70W GaN Class-E Power Amplifier Operating from 1.7 to 2.6 GHz. In Asia-Pacific Microwave Conference Proceedings (APMC), pages 1546–1549. IEEE, 2011.
- [77] M.P. van der Heijden, M. Acar, J.S. Vromans, and D.A. Calvillo-Cortes. A 19W High-Efficiency Wide-Band CMOS-GaN Class-E Chireix RF Outphasing Power Amplifier. In *IEEE MTT-S Int. Microwave Symp. Digest (IMS)*, pages 1–4, 2011.
- [78] Kanjun Shi, David A Calvillo-Cortes, Leo CN de Vreede, and Fred van Rijs. A Compact 65W 1.7–2.3 GHz Class-E GaN Power Amplifier for Base Stations. In European Microwave Integrated Circuits Conference (EuMIC), pages 542–545. IEEE, 2011.
- [79] David A Calvillo-Cortes, Mark P van der Heijden, Mustafa Acar, Michel de Langen, Robin Wesson, Fred van Rijs, and Leo CN de Vreede. A Package-Integrated Chireix Outphasing RF Switch-Mode High-Power Amplifier. *IEEE Transactions on Microwave Theory and Techniques*, 61:3721–3732, 2013.
- [80] Mustafa Acar, Anne Johan Annema, and Bram Nauta. Generalized Analytical Design Equations for Variable Slope Class-E Power Amplifiers. In *IEEE International Conference on Electronics, Circuits and Systems*, pages 431–434. IEEE, 2006.
- [81] M. Acar, A.J. Annema, and B. Nauta. Analytical Design Equations for Class-E Power Amplifiers. *IEEE Trans. on Circuits and Systems I: Fundamental Theory and Applications I: Regular Papers*, 54(12):2706–2717, 2007.
- [82] Mustafa Acar, Anne Johan Annema, and Bram Nauta. Variable-Voltage Class-E Power Amplifiers. In *IEEE/MTT-S International Microwave Symposium*, pages 1095– 1098. IEEE, 2007.
- [83] W Brown, J Mims, and N Heenan. An Experimental Microwave-Powered Helicopter. In IRE International Convention Record, volume 13, pages 225–235. IEEE, 1965.
- [84] WC Brown. The Microwave Powered Helicopter. J. Microwave Power, 1:1–20, 1966.
- [85] William C Brown. Experiments Involving a Microwave Beam to Power and Position a Helicopter. *IEEE Transactions on Aerospace and Electronic Systems*, 5(5):692–702, 1969.

- [86] William C Brown and E Eugene Eves. Beamed Microwave Power Transmission and its Application to Space. *IEEE Transactions on Microwave Theory and Techniques*, 40(6):1239–1250, 1992.
- [87] Zhang Xuejun, L.E. Larson, P.M. Asbeck, and R.A. Langridge. Analysis of Power Recycling Techniques for RF and Microwave Outphasing Power Amplifiers. *IEEE Trans.on Circuits and Systems II: Analog and Digital Signal Processing*, 49(5):312 – 320, May. 2002.
- [88] Michael Roberg, Erez Falkenstein, and Zoya Popovic. High-Efficiency Harmonically-Terminated Rectifier for Wireless Powering Applications. In *IEEE MTT-S Interna*tional Microwave Symposium Digest (IMS), pages 1–3. IEEE, 2012.
- [89] Michael Roberg, Tibault Reveyrand, Ignacio Ramos, Erez Avigdor Falkenstein, and Zoya Popovic. High-Efficiency Harmonically Terminated Diode and Transistor Rectifiers. *IEEE Transactions on Microwave Theory and Techniques*, 60:4043–4052, 2012.
- [90] David C Hamill. Class DE Inverters and Rectifiers for DC-DC conversion. In IEEE Power Electronics Specialists Conference, volume 1, pages 854–860. IEEE, 1996.
- [91] K Shinoda, M Matsuo, T Suetsugu, and S Mori. PWM Control Scheme of Resonant DC/DC Converter With Class DE Inverter and Class E Rectifier. In International Telecommunications Energy Conference, pages 829–832. IEEE, 1996.
- [92] H Koizumi, H Sekiya, M Matsuo, T Suetsugu, S Mori, and I Sasase. Resonant DC/DC Converter With Class DE Inverter and Class E Rectifier Using Thinned-Out Method. In International Telecommunications Energy Conference, pages 109–114. IEEE, 1998.
- [93] T Reveyrand, I Ramos, and Z Popović. Time-Reversal Duality of High-Efficiency RF Power Amplifiers. *Electronics Letters*, 48(25):1607–1608, 2012.
- [94] David C Hamill. Time Reversal Duality and the Synthesis of a Double Class E DC-DC Converter. In *IEEE Power Electronics Specialists Conference*, pages 512–521. IEEE, 1990.
- [95] M Nieves Ruiz, Reinel Marante, and Jose A Garcia. A Class E Synchronous Rectifier Based on an e-pHEMT Device for Wireless Powering Applications. In *IEEE MTT-S International Microwave Symposium Digest (IMS)*, pages 1–3. IEEE, 2012.
- [96] MARIAN K Kazimierczuk and JJ Jozwik. Analysis and Design of Class E Zero-Current-Switching Rectifier. *IEEE Transactions on Circuits and Systems*, 37(8):1000– 1009, 1990.
- [97] S Birca-Galateanu. Class E Inverters and Rectifiers. In European Conference on Circuits and Systems for Communications, pages 50–58. IEEE, 2008.
- [98] Marian K Kazimierczuk and Jacek Jozwik. Resonant DC/DC Converter With Class-E Inverter and Class-E Rectifier. *IEEE Transactions on Industrial Electronics*, 36(4):468–478, 1989.
- [99] F Van Rijs and SJCH Theeuwen. Efficiency Improvement of LDMOS Transistors for Base Stations: Towards the Theoretical Limit. In *International Electron Devices Meeting*, pages 1–4. IEEE, 2006.
- [100] Johan Ankarcrona, Klas-Håkan Eklund, Lars Vestling, and Jörgen Olsson. Simulation and Modeling of the Substrate Influence on the High Frequency Performance for RF LDMOS. In *GHz Symposium*, 2003.

- [101] Frederick H Raab. Average Efficiency of Class-G Power Amplifiers. IEEE Transactions on Consumer Electronics, 32(2):145–150, 1986.
- [102] William H Doherty. A New High Efficiency Power Amplifier for Modulated Waves. Proceedings of the Institute of Radio Engineers, 24(9):1163–1182, 1936.
- [103] Craig Steinbeiser, Thomas Landon, and Charles Suckling. 250W HVHBT Doherty with 57% WCDMA Efficiency Linearized to -55dBc for 2c11 6.5dB PAR. In *IEEE Compound Semiconductor Integrated Circuit Symposium*, pages 1–4, 2007.
- [104] Asdesach Zena Markos. Efficiency Enhancement of Linear GaN RF Power Amplifiers Using the Doherty Technique. Kassel University Press GmbH, 2008.
- [105] Jing Xia, Xiaowei Zhu, Lei Zhang, Jianfeng Zhai, and Yinjin Sun. High-Efficiency GaN Doherty Power Amplifier for 100-MHz LTE-Advanced Application Based on Modified Load Modulation Network. *IEEE Transactions on Microwave Theory and Techniques*, 61(8):2911–2921, 2013.
- [106] Andrei Grebennikov and Senad Bulja. High-Efficiency Doherty Power Amplifiers: Historical Aspect and Modern Trends. Proceedings of the IEEE, 100:3191–3219, 2012.
- [107] Rocco Giofrè, Paolo Colantonio, Franco Giannini, and Luca Piazzon. New Output Combiner for Doherty Amplifiers. *IEEE Microwave and Wireless Components Letters*, 23:31–33, 2013.
- [108] Charles F Campbell. A Fully Integrated Ku-Band Doherty Amplifier MMIC. IEEE Microwave and Guided Wave Letters, 9(3):114–116, 1999.
- [109] Debopriyo Chowdhury, Christopher D Hull, Ofir B Degani, Yanjie Wang, and Ali M Niknejad. A Fully Integrated Dual-Mode Highly Linear 2.4 GHz CMOS Power Amplifier for 4G WiMax Applications. *IEEE Journal of Solid-State Circuits*, 44(12):3393– 3402, 2009.
- [110] Ercan Kaymaksut, Brecht François, and Patrick Reynaert. Analysis and Design of Series Combining Transformers for Integrated Doherty Power Amplifiers. In Asia-Pacific Microwave Conference Proceedings (APMC), pages 1621–1624. IEEE, 2010.
- [111] Ercan Kaymaksut and Patrick Reynaert. A 2.4 GHz Fully Integrated Doherty Power Amplifier Using Series Combining Transformer. In *Proceedings of the ESSCIRC*, pages 302–305. IEEE, 2010.
- [112] Ercan Kaymaksut and Patrick Reynaert. Transformer-Based Uneven Doherty Power Amplifier in 90 nm CMOS for WLAN Applications. *IEEE Journal of Solid-State Circuits*, 47(7):1659–1671, 2012.
- [113] Jinshu Zhao, Robert Wolf, and Frank Ellinger. Fully Integrated LTE Doherty Power Amplifier. In International Semiconductor Conference Dresden-Grenoble (ISCDG), pages 1–3. IEEE, 2013.
- [114] Gil Wong Choi, Hyoung Jong Kim, Woong Jae Hwang, Suk Woo Shin, Jin Joo Choi, and Sung Jae Ha. High Efficiency Class-E Tuned Doherty Amplifier Using GaN HEMT. In *IEEE MTT-S International Microwave Symposium Digest (IMS)*, pages 925–928. IEEE, 2009.
- [115] Yong-Sub Lee, Mun-Woo Lee, and Yoon-Ha Jeong. Highly Efficient Doherty Amplifier Based on Class-E Topology for WCDMA Applications. *IEEE Microwave and Wireless Components Letters*, 18(9):608–610, 2008.

- [116] KW Eccleston, KJI Smith, PT Gough, and SI Mann. Harmonic Load Modulation in Doherty Amplifiers. *Electronics Letters*, 44(2):128–129, 2008.
- [117] Yong-Sub Lee, Yoon-Ha Jeong, et al. High-Efficiency Doherty Amplifier Using GaN HEMT class-F Cells for WCDMA Applications. In International Conference on Microwave and Millimeter Wave Technology, volume 1, pages 270–273. IEEE, 2008.
- [118] Paolo Colantonio, Franco Giannini, Rocco Giofre, and Luca Piazzon. Theory and Experimental Results of a Class F AB-C Doherty Power Amplifier. *IEEE Transactions* on Microwave Theory and Techniques, 57(8):1936–1947, 2009.
- [119] Ramzi Darraji and FadhelM Ghannouchi. Digital Doherty Amplifier With Enhanced Efficiency and Extended Range. *IEEE Transactions on Microwave Theory and Techniques*, 59(11):2898–2909, 2011.
- [120] Ramzi Darraji, Fadhel M Ghannouchi, and Oualid Hammi. A Dual-Input Digitally Driven Doherty Amplifier Architecture for Performance Enhancement of Doherty Transmitters. *IEEE Transactions on Microwave Theory and Techniques*, 59(5):1284– 1293, 2011.
- [121] Ildu Kim and Bumman Kim. A 2.655 GHz 3-Stage Doherty Power Amplifier Using Envelope Tracking Technique. In *IEEE MTT-S International Microwave Symposium Digest (IMS)*, pages 1496–1499. IEEE, 2010.
- [122] Mun-Woo Lee, Sang-Ho Kam, and Yoon-Ha Jeong. Highly Efficient Three-Stage Doherty Power Amplifier With Adaptive Driving Amplifier for 3.5 GHz WiMAX Applications. In *IEEE MTT-S International Microwave Symposium Digest (IMS)*, pages 1–4. IEEE, 2011.
- [123] Marco J Pelk, WC Neo, John R Gajadharsing, Raymond S Pengelly, and Leo CN de Vreede. A High-Efficiency 100-W GaN Three-Way Doherty Amplifier for Base-Station Applications. *IEEE Transactions on Microwave Theory and Techniques*, 56(7):1582–1591, 2008.
- [124] Andrei Grebennikov. A High-Efficiency 100-W Four-Stage Doherty GaN HEMT Power Amplifier Module for WCDMA Systems. In *IEEE MTT-S International Microwave Symposium Digest (IMS)*, pages 1–4, 2011.
- [125] D Cox. Linear Amplification With Nonlinear Components. IEEE Transactions on Communications, 22(12):1942–1945, 1974.
- [126] D Cox and R Leck. Component Signal Separation and Recombination for Linear Amplification with Nonlinear Components. *IEEE Transactions on Communications*, 23(11):1281–1287, 1975.
- [127] Kai-Yuan Jheng, Yuan-Jyue Chen, and An-Yeu Wu. Multilevel LINC System Design for Power Efficiency Enhancement. In *IEEE Workshop on Signal Processing Systems*, pages 31–34. IEEE, 2007.
- [128] Junqing Guan, Ahmed Farouk Aref, and Renato Negra. System-Level Performance Study of a Multistandard Outphasing Transmitter Using Optimised Multilevels. In *IEEE MTT-S International Microwave Symposium Digest (IMS)*, pages 1–4. IEEE, 2011.
- [129] SungWon Chung, Philip A Godoy, Taylor W Barton, Everest W Huang, David J Perreault, and Joel L Dawson. Asymmetric Multilevel Outphasing Architecture for
Multi-Standard Transmitters. In *IEEE Radio Frequency Integrated Circuits Symposium*, pages 237–240. IEEE, 2009.

- [130] SungWon Chung, Philip A Godoy, Taylor W Barton, David J Perreault, and Joel L Dawson. Asymmetric Multilevel Outphasing Transmitter Using Class-E PAs With Discrete Pulse Width Modulation. In *IEEE MTT-S International Microwave Sympo*sium Digest (IMS), pages 264–267. IEEE, 2010.
- [131] Philip A Godoy, S Chung, Taylor W Barton, David J Perreault, and Joel L Dawson. A Highly Efficient 1.95-GHz, 18-W Asymmetric Multilevel Outphasing Transmitter for Wideband Applications. In *IEEE MTT-S International Microwave Symposium Digest (IMS)*, pages 1–4. IEEE, 2011.
- [132] Philip A Godoy, SungWon Chung, Taylor W Barton, David J Perreault, and Joel L Dawson. A 2.4-GHz, 27-dBm Asymmetric Multilevel Outphasing Power Amplifier in 65-nm CMOS. *IEEE Journal of Solid-State Circuits*, 47:2372–2384, 2012.
- [133] Henry Chireix. High Power Outphasing Modulation. Proceedings of the Institute of Radio Engineers, 23(11):1370–1392, 1935.
- [134] Anu Huttunen and Risto Kaunisto. A 20-W Chireix Outphasing Transmitter for WCDMA Base Stations. *IEEE Transactions on Microwave Theory and Techniques*, 55(12):2709–2718, 2007.
- [135] Ilkka Hakala, David K Choi, Leila Gharavi, Nikolai Kajakine, Julius Koskela, and Risto Kaunisto. A 2.14-GHz Chireix Outphasing Transmitter. *IEEE Transactions* on Microwave Theory and Techniques, 53(6):2129–2138, 2005.
- [136] Ahmed Birafane and Ammar B Kouki. On the Linearity and Efficiency of Outphasing Microwave Amplifiers. *IEEE Transactions on Microwave Theory and Techniques*, 52(7):1702–1708, 2004.
- [137] M El-Asmar, A Birafane, and AB Kouki. Impact of PA Mismatching in Chireix-Outphasing System Without Stubs. In *International Symposium on Signals, Systems* and Electronics, pages 201–204. IEEE, 2007.
- [138] W Gerhard and R Knoechel. Improvement of Power Amplifier Efficiency by Reactive Chireix Combining, Power Back-Off and Differential Phase Adjustment. In *IEEE MTT-S International Microwave Symposium Digest*, pages 1887–1890. IEEE, 2006.
- [139] Ahmed Birafane and Ammar B Kouki. Phase-Only Predistortion for LINC Amplifiers with Chireix-Outphasing Combiners. *IEEE Transactions on Microwave Theory* and Techniques, 53(6):2240–2250, 2005.
- [140] Rui Liu, Dominique Schreurs, Walter De Raedt, and Robert Mertens. A Compact Tuneable Output Network for High Efficient Chireix Outphasing Power Amplifier Design. In Workshop on Integrated Nonlinear Microwave and Millimeter-Wave Circuits (INMMIC), pages 67–70. IEEE, 2010.
- [141] F Raab. Efficiency of Outphasing RF Power-Amplifier Systems. IEEE Transactions on Communications, 33(10):1094–1099, 1985.
- [142] Walter Gerhard and Reinhard Knoechel. Novel Transmission Line Combiner for Highly Efficient Outphasing RF Power Amplifiers. In *European Microwave Confer*ence, pages 1433–1436. IEEE, 2007.

- [143] Jawad H Qureshi, Marco J Pelk, Mauro Marchetti, WC Edmund Neo, John R Gajadharsing, Mark P van der Heijden, and LCN de Vreede. A 90-W Peak Power GaN Outphasing Amplifier With Optimum Input Signal Conditioning. *IEEE Transactions* on Microwave Theory and Techniques, 57(8):1925–1935, 2009.
- [144] Sungho Lee and Sangwook Nam. A CMOS Outphasing Power Amplifier With Integrated Single-Ended Chireix Combiner. *IEEE Transactions on Circuits and Systems II: Express Briefs*, 57(6):411–415, 2010.
- [145] Ramon Beltran, Frederick H Raab, and Arturo Velazquez. HF Outphasing Transmitter Using Class-E Power Amplifiers. In *IEEE MTT-S International Microwave Symposium Digest*, pages 757–760. IEEE, 2009.
- [146] Ramon Beltran, Frederick H Raab, and Green Mountain. An Outphasing Transmitter Using Class-E PAs and Asymmetric Combining: Part 1. *High Frequency Electronics*, page 18, 2011.
- [147] David J Perreault. A New Power Combining and Outphasing Modulation System for High-Efficiency Power Amplification. *IEEE Transactions on Circuits and Systems I: Regular Papers*, 58(8):1713–1726, 2011.
- [148] Taylor Wallis Barton. Phase Manipulation for Efficient Radio Frequency Transmission. PhD thesis, Massachusetts Institute of Technology, 2012.
- [149] T. W. Barton, J. Dawson, and D. Perreault. Four-Way Lossless Outphasing and Power Combining with Hybrid Microstrip/Discrete Combiner for Microwave Power Amplification. In Int. Microwave Symp. Digest (IMS), 2013.
- [150] Donald F Kimball, Jinho Jeong, Chin Hsia, Paul Draxler, Sandro Lanfranco, Walter Nagy, Kevin Linthicum, Lawrence E Larson, and Peter M Asbeck. High-Efficiency Envelope-Tracking W-CDMA Base-Station Amplifier Using GaN HFETs. *IEEE Transactions on Microwave Theory and Techniques*, 54(11):3848–3856, 2006.
- [151] Rameswor Shrestha, Ronan van der Zee, Anton de Graauw, and Bram Nauta. A Wideband Supply Modulator for 20MHz RF Bandwidth Polar PAs in 65nm CMOS. *IEEE Journal of Solid-State Circuits*, 44(4):1272–1280, 2009.
- [152] Junghwan Moon, Junghwan Son, Juyeon Lee, and Bumman Kim. A Multimode/Multiband Envelope Tracking Transmitter With Broadband Saturated Amplifier. *IEEE Transactions on Microwave Theory and Techniques*, 59(12):3463–3473, 2011.
- [153] Myoungbo Kwak, Donald F Kimball, Calogero D Presti, Antonino Scuderi, Carmelo Santagati, Jonmei J Yan, Peter M Asbeck, and Lawrence E Larson. Design of a Wideband High-Voltage High-Efficiency BiCMOS Envelope Amplifier for Micro-Base-Station RF Power Amplifiers. *IEEE Transactions on Microwave Theory and Techniques*, 60(6):1850–1861, 2012.
- [154] Feipeng Wang, Adedayo Ojo, Donald Kimball, Peter Asbeck, and Lawrence Larson. Envelope Tracking Power Amplifier With Pre-Distortion Linearization for WLAN 802.11g. In *IEEE MTT-S International Microwave Symposium Digest (IMS)*, volume 3, pages 1543–1546. IEEE, 2004.
- [155] Z Yusoff, J Lees, J Benedikt, PJ Tasker, and SC Cripps. Linearity Improvement in RF power Amplifier System Using Integrated Auxiliary Envelope Tracking System. In *IEEE MTT-S International Microwave Symposium Digest (IMS)*, pages 1–4. IEEE, 2011.

- [156] Frederick H Raab and DJ Rupp. High-Efficiency Single-Sideband HF/VHF Transmitter Based Upon Envelope Elimination and Restoration. In International Conference on HF Radio Systems and Techniques, pages 21–25. IET, 1994.
- [157] John Hoversten, Scott Schafer, Michael Roberg, Mark Norris, Dragan Maksimovic, and Zoya Popovic. Codesign of PA, Supply, and Signal Processing for Linear Supply-Modulated RF Transmitters. *IEEE Transactions on Microwave Theory and Techniques*, 60(6):2010–2020, 2012.
- [158] Chi-Tsan Chen, Yi-Chieh Lin, Tzyy-Sheng Horng, Kang-Chun Peng, and Chien-Jung Li. Kahn Envelope Elimination and Restoration Technique Using Injection-Locked Oscillators. In *IEEE MTT-S International Microwave Symposium Digest* (*IMS*), pages 1–3. IEEE, 2012.
- [159] Leonard R Kahn. Single-Sideband Transmission by Envelope Elimination and Restoration. Proceedings of the IRE, 40(7):803–806, 1952.
- [160] Leonard R Kahn. Comparison of Linear Single-Sideband Transmitters With Envelope Elimination and Restoration Single-Sideband Transmitters. *Proceedings of the IRE*, 44(12):1706–1712, 1956.
- [161] Frederick H Raab. High-Efficiency Linear Amplification by Dynamic Load Modulation. In *IEEE MTT-S International Microwave Symposium Digest (IMS)*, volume 3, pages 1717–1720. IEEE, 2003.
- [162] Jia-Shiang Fu and Amir Mortazawi. Improving Power Amplifier Efficiency and Linearity Using a Dynamically Controlled Tunable Matching Network. *IEEE Trans*actions on Microwave Theory and Techniques, 56(12):3239–3244, 2008.
- [163] Hossein Mashad Nemati, Christian Fager, Ulf Gustavsson, Rik Jos, and Herbert Zirath. Design of Varactor-Based Tunable Matching Networks for Dynamic Load Modulation of High Power Amplifiers. *IEEE Transactions on Microwave Theory and Techniques*, 57(5):1110–1118, 2009.
- [164] Ali Soltani Tehrani, Hossein Mashad Nemati, Haiying Cao, Thomas Eriksson, and Christian Fager. Dynamic Load Modulation of High Power Amplifiers With Varactor-Based Matching Networks. In *IEEE MTT-S International Microwave Symposium Digest (IMS)*, pages 1537–1540. IEEE, 2009.
- [165] Mattias Thorsell, Kristoffer Andersson, and Christian Fager. Characterization Setup for Device Level Dynamic Load Modulation Measurements. In *IEEE MTT-S International Microwave Symposium Digest (IMS)*, pages 1197–1200. IEEE, 2009.
- [166] Christian Fager, Haiying Cao, Thomas Eriksson, Rik Jos, H Nemati, A Soltani, and Herbert Zirath. High Efficiency Transmitter Using Varactor Based Dynamic Load Modulation. In *IEEE International Microwave Workshop Series on RF Front-ends* for Software Defined and Cognitive Radio Solutions (IMWS), pages 1–4. IEEE, 2010.
- [167] Hossein Mashad Nemati, Alan L Clarke, Steve C Cripps, Johannes Benedikt, Paul J Tasker, Christian Fager, Jan Grahn, and Herbert Zirath. Evaluation of a GaN HEMT Transistor for Load-and Supply-Modulation Applications Using Intrinsic Waveform Measurements. In *IEEE MTT-S International Microwave Symposium Digest (IMS)*, pages 509–512. IEEE, 2010.
- [168] Kenle Chen and Dimitrios Peroulis. Design of Adaptive Highly Efficient GaN Power Amplifier for Octave-Bandwidth Application and Dynamic Load Modulation. *IEEE Transactions on Microwave Theory and Techniques*, 60(6):1829–1839, 2012.

- [169] M Ozen, CM Andersson, Mattias Thorsell, Kristoffer Andersson, Niklas Rorsman, Christian Fager, M Acar, MP Van Der Heijden, and R Jos. High Efficiency RF Pulse Width Modulation With Tunable Load Network Class-E PA. In *IEEE 12th Annual Wireless and Microwave Technology Conference (WAMICON)*, pages 1–6. IEEE, 2011.
- [170] Mustafa Ozen, Christer M Andersson, Thomas Eriksson, Mustafa Acar, Rik Jos, and Christian Fager. Linearization Study of a Highly Efficient CMOS-GaN RF Pulse Width Modulation Based Transmitter. In *European Microwave Conference (EuMC)*, pages 136–139. IEEE, 2012.
- [171] Frederick H Raab, Peter Asbeck, Steve Cripps, Peter B Kenington, Zoya B Popovic, Nick Pothecary, John F Sevic, and Nathan O Sokal. RF and Microwave Power Amplifier and Transmitter Technologies-Part 1. *High Frequency Electronics*, 2(3):22– 36, 2003.
- [172] T. Blocher and P. Singerl. Coding Efficiency for Different Switched-Mode RF Transmitter Architectures. In *IEEE Int. Midwest Symp. Circuits and Systems (MWSCAS)*, pages 276–279, 2009.
- [173] Jeon. Young-Sang and Nam Sangwook. A Discrete-Amplitude Pulse Width Modulation for a High-Efficiency Linear Power Amplifier. *ETRI Journal*, 33(5):679–688, Oct. 2011.
- [174] U. Gustavsson, T. Eriksson, H. Mashad Nemati, P. Saad, P. Singerl, and C. Fager. An RF Carrier Bursting System Using Partial Quantization Noise Cancellation. *IEEE Trans. on Circuits and Systems I: Regular Papers*, 59(3):515–528, 2012.
- [175] S. Kim and Y. E. Wang. Theory of Switched RF Resonators. IEEE Trans. Circuits Systems I, 53(12):2521–2528, 2006.
- [176] Shuhsien Liao and Y. E. Wang. A Pulsed Load Modulation (PLM) Power Amplifier with 0.35µm pHEMT Technology. In *IEEE Compound Semiconductor Integrated Circuit Symp. CISC 2009*, pages 1–4, 2009.
- [177] Shu-Hsien Liao and Y. E. Wang. High Efficiency WCDMA Power Amplifier With Pulsed Load Modulation (PLM). *IEEE J. Solid-State Circuits*, 45(10):2030–2037, 2010.
- [178] B. François, P. Reynaert, A. Wiesbauer, and P. Singerl. Analysis of Burst-Mode RF PA with Direct Filter Connection. In Proc. European Microwave Conf. (EuMC), pages 974–977, 2010.
- [179] S. C. Pires, P. M. Cabral, and J. C. Pedro. A Carrier-Burst Transmitter Implementation: Design of Bandpass Filter and Amplifier-BPF Connection. In Proc. Workshop Integrated Nonlinear Microwave and Millimetre-Wave Circuits (INMMIC), pages 1–3, 2012.
- [180] Patrick Reynaert and Michel Steyaert. A 2.45-GHz 0.13-mu m CMOS PA With Parallel Amplification. *IEEE Journal of Solid-State Circuits*, 42(3):551–562, 2007.
- [181] Patrick Reynaert and Michiel Steyaert. A Fully Integrated CMOS RF Power Amplifier With Parallel Power Combining and Power Control. In Asian Solid-State Circuits Conference, pages 137–140. IEEE, 2005.

- [182] Alireza Shirvani, David K Su, and Bruce A Wooley. A CMOS RF Power Amplifier With Parallel Amplification for Efficient Power Control. *IEEE Journal of Solid-State Circuits*, 37(6):684–693, 2002.
- [183] Ichiro Aoki, Scott D Kee, David B Rutledge, and Ali Hajimiri. Distributed Active Transformer-a New Power-Combining and Impedance-Transformation Technique. *IEEE Transactions on Microwave Theory and Techniques*, 50(1):316–331, 2002.
- [184] Ichiro Aoki. Distributed Active Transformer for Integrated Power Amplification. PhD thesis, California Institute of Technology, 2002.
- [185] Seungwoo Kim, Kyungho Lee, Jongwoo Lee, Bumman Kim, Scott D Kee, Ichiro Aoki, and David B Rutledge. An Optimized Design of Distributed Active Transformer. *IEEE Transactions on Microwave Theory and Techniques*, 53(1):380–388, 2005.
- [186] Gang Liu, Tsu-Jae King Liu, and Ali M Niknejad. A 1.2 V, 2.4 GHz Fully Integrated Linear CMOS Power Amplifier With Efficiency Enhancement. In *IEEE Custom Inte*grated Circuits Conference, pages 141–144. IEEE, 2006.
- [187] Andrew D Pye and Mona M Hella. Analysis and Optimization of Transformer-Based Series Power Combining for Reconfigurable Power Amplifiers. *IEEE Transactions on Circuits and Systems I: Regular Papers*, 58(1):37–50, 2011.
- [188] Gang Liu, Peter Haldi, Tsu-Jae King Liu, and Ali M Niknejad. Fully Integrated CMOS Power Amplifier With Efficiency Enhancement at Power Back-Off. *IEEE Journal of Solid-State Circuits*, 43(3):600–609, 2008.
- [189] Amirpouya Kavousian, David K Su, and Bruce A Wooley. A Digitally Modulated Polar CMOS PA With 20MHz Signal BW. In *IEEE International Solid-State Circuits Conference*, pages 78–588. IEEE, 2007.
- [190] Amirpouya Kavousian, David K Su, Mohammad Hekmat, Alireza Shirvani, and Bruce A Wooley. A Digitally Modulated Polar CMOS Power Amplifier With a 20-MHz Channel Bandwidth. *IEEE Journal of Solid-State Circuits*, 43(10):2251–2258, 2008.
- [191] Shuli Chi, Peter Singerl, and Christian Vogel. Coding Efficiency Optimization for Multilevel PWM Based Switched-Mode RF Transmitters. In *IEEE Int. Midwest Symp. on Circuits and Systems (MWSCAS)*, pages 1–4. IEEE, 2011.
- [192] Shuli Chi, Peter Singerl, and Christian Vogel. Efficiency Optimization for Burst-Mode Multilevel Radio Frequency Transmitters. *IEEE Trans. on Circuits and Sys*tems I: Regular Papers, 60:1901–1914, 2013.
- [193] Jinseong Jeong and Y. E. Wang. Pulsed Load Modulation (PLM) Technique for Efficient Power Amplification. *IEEE Transactions on Circuits and Systems II: Express Briefs*, 55(10):1011–1015, 2008.
- [194] Jau-Horng Chen, Hao-Shun Yang, Hou-Chung Lin, and Y-JE Chen. A polartransmitter architecture using multiphase pulsewidth modulation. *IEEE Transactions* on Circuits and Systems I: Regular Papers, 58(2):244–252, 2011.
- [195] Katharina Hausmair, Shuli Chi, and Christian Vogel. How to Reach 100% Coding Efficiency in Multilevel Burst-Mode RF Transmitters. In Proc. IEEE Int. Symp. Circuits Syst, 2013.

- [196] R. Langridge, T. Thornton, P.M. Asbeck, and L.E. Larson. A Power Re-Use Technique for Improved Efficiency of Outphasing Microwave Power Amplifiers. *IEEE Trans. on Microwave Theory and Techniques*, 47(8):1467–1470, Aug. 1999.
- [197] Philip A Godoy, David J Perreault, and Joel L Dawson. Outphasing Energy Recovery Amplifier With Resistance Compression for Improved Efficiency. *IEEE Transactions* on Microwave Theory and Techniques, 57(12):2895–2906, 2009.
- [198] Young-Sang Jeon, Hoe-Sung Yang, and Sangwook Nam. A Power Re-Use Technique for Improved Efficiency of Pulsed Oscillating Amplifiers. *IEEE Microw. Wireless Compon. Lett.*, 16(10):567–569, 2006.
- [199] Sheikh Nijam Ali and Thomas E. Johnson. RF Switch-Mode Power Amplifier with an Integrated Diplexer for Signal Reconstruction and Energy Recovery. In *IEEE MTT-S Int. Microwave Symp. Digest (MTT)*, June 2012.
- [200] S.N. Ali and T. Johnson. A New High Efficiency RF Switch-Mode Power Amplifier Architecture for Pulse Encoded Signals. In Wireless and Microwave Technology Conf. (WAMICON), April 2012.
- [201] Shuli Chi, Christian Vogel, and Peter Singerl. The Frequency Spectrum of Polar Modulated PWM Signals and the Image Problem. In 17th IEEE Int. Conf. on Electronics, Circuits, and Systems (ICECS), pages 679–682, Dec. 2010.
- [202] Sergio C. Pires, Pedro M. Cabral, and Jose C. Pedro. RF Carrier Phase-Burst Transmitter. In Int. Microwave Symp. (IMS), 2013.
- [203] Kevin Tom, Mike Faulkner, and Thomas Lejon. Performance Analysis of Pulse Width Modulated RF Class-E Power Amplifier. In *IEEE Vehicular Technology Conference*, volume 4, pages 1807–1811. IEEE, 2006.
- [204] Michael Nielsen and Torben Larsen. An RF Pulse Width Modulator for Switch-Mode Power Amplification of Varying Envelope Signals. In *Topical Meeting on Silicon Monolithic Integrated Circuits in RF Systems*, pages 277–280. IEEE, 2007.
- [205] Jeffrey S. Walling, Hasnain Lakdawala, Yorgos Palaskas, Ashoke Ravi, Ofir Degani, Krishnamurthy Soumyanath, and David J. Allstot. A Class-E PA With Pulse-Width and Pulse-Position Modulation in 65 nm CMOS. *IEEE Journal of Solid-State Circuits*, 44(6):1668–1678, 2009.
- [206] Frederick H Raab. Class-D Power Amplifier With RF Pulse-Width Modulation. In *IEEE MTT-S International Microwave Symposium Digest (IMS)*, pages 924–927. IEEE, 2010.
- [207] Bonghyuk Park and Jaeho Jung. A Fully Integrated Pulsewidth Modulator for Class-S System. In International Conference on Advanced Communication Technology (ICACT), pages 274–277. IEEE, 2012.
- [208] K. Hausmair, Shuli Chi, Peter Singerl, and Christian Vogel. Aliasing-Free Digital Pulse-Width Modulation for Burst-Mode RF Transmitters. *IEEE Trans. on Circuits* and Systems I: Regular Papers, 60(2):415–427, 2013.
- [209] Katharina Hausmair, Peter Singerl, and Christian Vogel. Multiplierless Implementation of an Aliasing-Free Digital Pulse-Width Modulator. *IEEE Trans. on Circuits* and Systems II, 60:592–596, Sept 2013.

- [210] Darko Kajfez and Eugene J Hwan. Q-factor Measurement with Network Analyzer. IEEE Trans. Microwave Theory and Techniques, 32(7):666–670, 1984.
- [211] En-Yuan Sun and Shuh-Han Chao. Unloaded Q Measurement The Critical Points Method. IEEE Trans. Microwave Theory and Techniques, 43(8):1983–1986, 1995.
- [212] Lye Heng Chua and Dariush Mirshekar-Syahkal. Accurate and Direct Characterization of High-Q Microwave Resonators Using One-Port Measurement. *IEEE Trans. Microwave Theory and Techniques*, 51(3):978–985, 2003.
- [213] Changying Wu, Kai Zhang, Gao Wei, and Jia-Dong Xu. Applying Critical-Points Method in the Presence of Phase Shift Due to Feed Line. *Progress In Electromagnetics Research*, 5:15–23, 2008.
- [214] DT Lopes, CC Motta, et al. Unloaded Q-Factor Measurements in Klystron Cavities. In SBMO/IEEE MTT-S Int. Microwave & Optoelectronics Conf. (IMOC), pages 946–949, 2011.
- [215] M Sudow, Martin Fagerlind, Mattias Thorsell, Kristoffer Andersson, Niklas Billstrom, P-A Nilsson, and Niklas Rorsman. An AlGaN/GaN HEMT-Based Microstrip MMIC Process for Advanced Transceiver Design. *IEEE Transactions on Microwave Theory and Techniques*, 56(8):1827–1833, 2008.
- [216] Yinsidi Jiao, Wei-Han Yu, Pui-In Mak, and Rui P Martins. A Dynamic-Range-Improved 2.4 GHz WLAN Class-E PA Combining PWPM and Cascode Modulation. In *IEEE Asia Pacific Conf. on Circuits and Systems (APCCAS)*, pages 148–151. IEEE, 2012.
- [217] Li-Yuan Yang, Hsin-Shu Chen, and Y-JE Chen. A 2.4 GHz Fully Integrated Cascode-Cascade CMOS Doherty Power Amplifier. *IEEE Microwave and Wireless Components Letters*, 18(3):197–199, 2008.
- [218] Hyoung-Seok Oh, Cheon-Soo Kim, Hyun-Kyu Yu, and Choong-Ki Kim. A Fully-Integrated +23-dBm CMOS Triple Cascode Linear Power Amplifier with Inner-Parallel Power Control Scheme. In *IEEE Radio Frequency Integrated Circuits (RFIC)* Symp., 2006.
- [219] Kevin W Kobayashi. An 8-Watt 250-3000MHz Low Noise GaN MMIC Feedback Amplifier With +50dBm OIP3. In *IEEE Compound Semiconductor Integrated Circuit Symp. (CSICS)*, pages 1–4. IEEE, 2011.
- [220] Audrey Martin, Tibault Reveyrand, Michel Campovecchio, Raphael Aubry, Stéphane Piotrowicz, Didier Floriot, Raymond Quéré, et al. Design Method of Balanced AlGaN/GaN HEMT Cascode Cells for Wideband Distributed Power Amplifiers. *Proc. European Microwave Association*, 261:267, 2008.
- [221] Adeline Dechansiaud, Raphaël Sommet, Tibault Reveyrand, Raymond Quéré, C Chang, D Bouw, M Camiade, and F Deborgies. New Compact Power Cells for Ku Band Applications. In *IEEE Workshop on Integrated Nonlinear Microwave and Millimetre-Wave Circuits (INMMIC)*, pages 1–4, 2011.
- [222] Bruce M Green, Kenneth K Chu, Joseph A Smart, Vinayak Tilak, Hyungtak Kim, James R Shealy, and Lester F Eastman. Cascode Connected AlGaN/GaN HEMTs on SiC Substrates. *IEEE Microwave and Guided Wave Letters*, 10(8):316–318, 2000.

- [223] Joon Hyung Kim, Sung Jun Lee, Jae Ho Jung, and Chul Soon Park. 60% High-Efficiency 3G LTE Power Amplifier with Three-Level Delta Sigma Modulation Assisted by Dual Supply Injection. In *IEEE MTT-S Int. Microwave Symp. Digest* (*IMS*), pages 1–4. IEEE, 2011.
- [224] Balaji Lakshminarayanan, Horace C Gordon, and Thomas M Weller. A Substrate-Dependent CAD Model for Ceramic Multilayer Capacitors. *IEEE Trans. Microwave Theory and Techniques*, 48(10):1687–1693, 2000.
- [225] David A Calvillo-Cortes, Mustafa Acar, Mark P van der Heijden, Melina Apostolidou, Leo CN de Vreede, Domine Leenaerts, and Jan Sonsky. A 65nm CMOS Pulse-Width-Controlled Driver With 8Vpp Output Voltage for Switch-mode RF PAs up to 3.6 GHz. In *IEEE Int. Solid-State Circuits Conf. Digest of Technical Papers* (ISSCC), pages 58–60, 2011.
- [226] P. Singerl, M. Mataln, C. Schuberth, F. Dielacher, D. Seebacher, C. Vogel, and K. Hausmair. Implementation of a 2.6 GHz, 5-W Carrier Bursting Switched Mode Transmitter. In Workshop Proceedings European Microwave Week (EuMW), 2013.
- [227] Thomas Ahlström and Claes Bengtsson. Digital Interference Cancellation in a Full Duplex System, 2008.
- [228] Jung Il Choi, Mayank Jain, Kannan Srinivasan, Phil Levis, and Sachin Katti. Achieving Single Channel, Full Duplex Wireless Communication. In *Proceedings International Conference on Mobile Computing and Networking*, pages 1–12. ACM, 2010.
- [229] Na Li, Weihong Zhu, and Haihua Han. Digital Interference Cancellation in Single Channel, Full Duplex Wireless Communication. In International Conference on Wireless Communications, Networking and Mobile Computing (WiCOM), pages 1–4, 2012.
- [230] Taneli Riihonen and Risto Wichman. Analog and Digital Self-Interference Cancellation in Full-Duplex MIMO-OFDM Transceivers with Limited Resolution in A/D Conversion. In Asilomar Conference on Signals, Systems and Computers (ASILO-MAR), pages 45–49, 2012.
- [231] N. Phungamngern, P. Uthansakul, and M. Uthansakul. Digital and RF Interference Cancellation for Single-Channel Full-Duplex Transceiver Using a Single Antenna. In International Conference on Electrical Engineering/Electronics, Computer, Telecommunications and Information Technology (ECTI-CON), pages 1–5, 2013.
- [232] Yang-Seok Choi and Hooman Shirani-Mehr. Simultaneous Transmission and Reception: Algorithm, Design and System Level Performance. *IEEE Transactions on Wireless Communications*, 12(12):5992–6010, 2013.
- [233] Pieter AJ Nuyts, Peter Singerl, Franz Dielacher, Patrick Reynaert, and Wim Dehaene. A Fully Digital GHz-Range Multimode Transmitter Front-End Based on Locked Delay Lines. In *European Reconfigurable Radio Technologies Workshop*, pages 126–128, 2010.
- [234] Pieter AJ Nuyts, Peter Singerl, Franz Dielacher, Patrick Reynaert, and Wim Dehaene. A Fully Digital Delay Line Based GHz Range Multimode Transmitter Front-End in 65-nm CMOS. *IEEE Journal of Solid-State Circuits*, 47(7):1681–1692, 2012.
- [235] Pieter AJ Nuyts, Brecht François, Wim Dehaene, and Patrick Reynaert. A CMOS Burst-Mode Transmitter With Watt-Level RF PA and Flexible Fully Digital Front-End. *IEEE Transactions on Circuits and Systems II: Express Briefs*, 2012.

- [236] Pieter AJ Nuyts, Patrick Reynaert, and Wim Dehaene. Frequency-Domain Analysis of Digital PWM-Based RF Modulators for Flexible Wireless Transmitters. *IEEE Transactions on Circuits and Systems I: Regular Papers*, 2013.
- [237] Pieter AJ Nuyts, Patrick Reynaert, and Wim Dehaene. Continuous-Time Digital Front-Ends for Multistandard Wireless Transmission. Springer, 2014.
- [238] Mustafa Ozen, Rik Jos, Christer M Andersson, Mustafa Acar, and Christian Fager. High-Efficiency RF Pulsewidth Modulation of Class-E Power Amplifiers. *IEEE Trans.* on Microwave Theory and Techniques, 59(11):2931–2942, 2011.
- [239] Michael Nielsen and Torben Larsen. A 2-GHz GaAs HBT RF Pulsewidth Modulator. IEEE Trans. on Microwave Theory and Techniques, 56(2):300–304, 2008.
- [240] Ashoke Ravi, Paolo Madoglio, Hongtao Xu, Kailash Chandrashekar, Marian Verhelst, Stefano Pellerano, Luis Cuellar, Mariano Aguirre-Hernandez, Masoud Sajadieh, Jorge E Zarate-Roldan, et al. A 2.4-GHz 20–40-MHz Channel WLAN Digital Outphasing Transmitter Utilizing a Delay-Based Wideband Phase Modulator in 32-nm CMOS. *IEEE Journal of Solid-State Circuits*, 2012.
- [241] M.S. Alavi, R.B. Staszewski, L.C.N. de Vreede, A. Visweswaran, and J.R. Long. All-Digital RF I/Q Modulator. *IEEE Trans. on Microwave Theory and Techniques*, 60(11):3513–3526, 2012.
- [242] M.S. Alavi, G. Voicu, R.B. Staszewski, L.C.N. de Vreede, and J.R Long. A 2x13-bit all-digital I/Q RF-DAC in 65-nm CMOS. In *Radio Frequency Integrated Circuits* Symposium (RFIC), pages 167–170, 2013.
- [243] Robert Bogdan Staszewski, Khurram Muhammad, Dirk Leipold, Chih-Ming Hung, Yo-Chuol Ho, John L Wallberg, Chan Fernando, Ken Maggio, Roman Staszewski, Tom Jung, et al. All-Digital TX Frequency Synthesizer and Discrete-Time Receiver for Bluetooth Radio in 130-nm CMOS. *IEEE Journal of Solid-State Circuits*, 39(12):2278–2291, 2004.
- [244] Robert Bogdan Staszewski, John L Wallberg, Sameh Rezeq, Chih-Ming Hung, Oren E Eliezer, Sudheer K Vemulapalli, Chan Fernando, Ken Maggio, Roman Staszewski, Nathen Barton, et al. All-Digital PLL and Transmitter for Mobile Phones. *IEEE Journal of Solid-State Circuits*, 40(12):2469–2482, 2005.
- [245] Petri Eloranta, Pauli Seppinen, Sami Kallioinen, Tuomas Saarela, and Aarno Parssinen. A Multimode Transmitter in 0.13 μm CMOS Using Direct-Digital RF Modulator. *IEEE Journal of Solid-State Circuits*, 42(12):2774–2784, 2007.
- [246] Zdravko Boos, Andreas Menkhoff, Franz Kuttner, Markus Schimper, Jose Moreira, Hans Geltinger, Timo Gossmann, Peter Pfann, Alexander Belitzer, and Thomas Bauernfeind. A Fully Digital Multimode Polar Transmitter Employing 17b RF DAC in 3G Mode. In *IEEE Int. Solid-State Circuits Conference Digest of Technical Papers* (*ISSCC*), pages 376–378. IEEE, 2011.
- [247] Min Park, Michael H Perrott, and Robert Bogdan Staszewski. An Amplitude Resolution Improvement of an RF-DAC Employing Pulsewidth Modulation. *IEEE Trans.* on Circuits and Systems I: Regular Papers, 58(11):2590–2603, 2011.
- [248] Quentin Beraud-Sudreau, André Mariano, Dominique Dallet, Yann Deval, and J Begueret. A Fully Integrated 4GHz Continuous-Time Bandpass Delta-Sigma Converter. In *IEEE Int. Conf. on Electronics, Circuits, and Systems (ICECS), 2010*, pages 778– 781, 2010.

- [249] Albert Jerng and Charles G Sodini. A Wideband  $\Delta\Sigma$  Digital-RF Modulator for High Data Rate Transmitters. *IEEE Journal of Solid-State Circuits*, 42(8):1710–1722, 2007.
- [250] Mohamed Helaoui, Safar Hatami, Renato Negra, and Fadhel M Ghannouchi. A Novel Architecture of Delta-Sigma Modulator Enabling All-Digital Multiband Multistandard RF Transmitters Design. *IEEE Trans. on Circuits and Systems II: Express Briefs*, 55(11):1129–1133, 2008.
- [251] Pin-En Su and Sudhakar Pamarti. A 2.4 GHz Wideband Open-Loop GFSK Transmitter with Phase Quantization Noise Cancellation. *IEEE Journal of Solid-State Circuits*, 46(3):615–626, 2011.
- [252] Martin Schmidt, Sebastian Haug, Markus Grozing, and Manfred Berroth. A Pipelined 3-level Bandpass Delta-Sigma Modulator for Class-S Power Amplifiers. In *IEEE Int. Symp.on Circuits and Systems (ISCAS)*, pages 2757–2760, 2011.
- [253] P.A.J. Nuyts, P. Reynaert, and W. Dehaene. A Fully Digital PWM-Based 1 to 3 GHz Multistandard Transmitter in 40-nm CMOS. In *IEEE Radio Frequency Integrated Circuits Symposium (RFIC)*, pages 419–422, June 2013.
- [254] Jack E Volder. The CORDIC Trigonometric Computing Technique. IRE Trans. on Electronic Computers, EC-8(3):330–334, 1959.
- [255] Piotr Dudek, Stanislaw Szczepanski, and John V Hatfield. A High-Resolution CMOS Time-to-Digital Converter Utilizing a Vernier Delay Line. *IEEE Journal of Solid-State Circuits*, 35(2):240–247, 2000.
- [256] Chorng-Sii Hwang, Poki Chen, and Hen-Wai Tsao. A High-Precision Time-to-Digital Converter Using a Two-Level Conversion Scheme. *IEEE Trans. on Nuclear Science*, 51(4):1349–1352, 2004.
- [257] S. Henzler, S. Koeppe, D. Lorenz, W. Kamp, R. Kuenemund, and D. Schmitt-Landsiedel. A Local Passive Time Interpolation Concept for Variation-Tolerant High-Resolution Time-to-Digital Conversion. *IEEE Journal of Solid-State Circuits*, 43(7):1666–1676, July 2008.
- [258] Richard Schreier, Gabor C Temes, and John Wiley. Understanding Delta-Sigma Data Converters, volume 74. IEEE Press Piscataway, NJ, 2005.
- [259] Vandana Bassoo, Lance Linton, and Michael Faulkner. Analysis of Distortion in Pulse Modulation Converters for Switching Radio Frequency Power Amplifiers. *IET Microwaves, Antennas & Propagation*, 4(12):2088–2096, 2010.
- [260] Stefano Santi, Riccardo Rovatti, and Gianluca Setti. Spectral Aliasing Effects of PWM Signals with Time-Quantized Switching Instants. In Int. Symp. on Circuits and Systems, volume 4, 2004.
- [261] Bah-Hwee Gwee, Joseph S. Chang, and Victor Adrian. A Micropower Low-Distortion Digital Class-D Amplifier Based on an Algorithmic Pulsewidth Modulator. *IEEE Trans. on Circuits and Systems I: Regular Papers*, 52(10):2007–2022, 2005.
- [262] A. M. A. Amin and S. A. El-Korfolly, M. I.and Mohammed. Exploring Aliasing Distortion Effects on Regularly-Sampled PWM Signals. In *IEEE Conf. Industrial Electronics and Applications*, pages 2036–2041, 2008.

- [263] JM Goldberg and MB Sandler. Pseudo-Natural Pulse Width Modulation for High Accuracy Digital-to-Analogue Conversion. *Electronics Letters*, 27(16):1491–1492, 1991.
- [264] MB Sandler. Digital-to-Analogue Conversion Using Pulse Width Modulation. Electronics & communication engineering journal, 5(6):339–348, 1993.
- [265] JM Goldberg and MB Sandler. New High Accuracy Pulse Width Modulation Based Digital-to-Analogue Convertor/Power Amplifier. *IEE Proceedings-Circuits, Devices* and Systems, 141(4):315–324, 1994.
- [266] Olivier JJ Michel, Alfred O Hero III, and Anne Emmanuelle Badel. Tree-Structured Nonlinear Signal Modeling and Prediction. *IEEE Trans. on Signal Processing*, 47(11):3027–3041, 1999.
- [267] Y Ferdi, JP Herbeuval, and A Charef. Variance Reduction of Prediction Error Using Fractional Digital Differentiation: Application to ECG Signal Processing. In Int. Conf. of the IEEE Engineering in Medicine and Biology Society, volume 3, pages 2170–2173. IEEE, 2001.
- [268] E Colinet, J Juillard, and R Kielbasa. A Novel Architecture of a Bandpass Sigma-Delta Converter Based on Signal Prediction. In *Midwest Symp. on Circuits and Systems (MWSCAS)*, volume 2, pages II–189. IEEE, 2004.
- [269] Wenlian Zhang, Chunming Zhang, Jinglei Han, and Zhibiao Shao. A Prony Model Based Signal Prediction Algorithm for Folding and Interpolating ADC. In Int. Conf. on Solid-State and Integrated Circuit Technology (ICSICT), pages 1–3. IEEE, 2012.
- [270] Jingcheng Zhuang and R.B. Staszewski. A Low-Power All-Digital PLL Architecture Based on Phase Prediction. In *IEEE Int. Conf. on Electronics, Circuits and Systems* (*ICECS*), pages 797–800, 2012.
- [271] D. Markert, C. Haslach, G. Luz, G. Fischer, and A. Pascht. Wideband measurements and linearization of a simplified architecture for analog RF-PWM. In *European Mi*crowave Conference (EuMC), pages 716–719, 2013.
- [272] Texas Instruments. Datasheet SN74LVC2GU04 DUAL INVERTER GATE, 2007.

## List of Figures

1.1	Number of mobile subscriptions using broadband services and total number of subscriptions, data ratriaved from [1]	1
19	Power consumption radio base station data retrieved from [8]	2
1.2	Tower consumption radio base station, data retrieved from [0].	2
2.1	Class A schematic and efficiency.	4
2.2	Drain voltage and current time domain waveforms for class A, B and C. $$ .	5
2.3	General harmonic contribution and power, gain and efficiency versus con-	
	duction angle	5
2.4	Class J schematic and time domain waveforms. $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	6
2.5	$ClassF \ schematic \ and \ time \ domain \ waveforms. \ \ \ldots $	7
2.6	Inverse class $F$ schematic and time domain waveforms. 	7
2.7	Current mode class D schematic and time domain waveforms	8
2.8	Voltage mode class D schematic and time domain waveforms. $\ldots$ .	9
2.9	Inverter based voltage mode class D schematic and time domain waveforms.	9
2.10	Conventional class DE schematic and time domain waveforms	10
2.11	Voltage mode class DE using parallel compensation inductance	10
2.12	Voltage mode class E schematic and time domain waveforms. $\ldots$ .	11
2.13	Class B rectifier.	12
2.14	$Class C \ rectifier \ (self \ aligned \ diode). \ \ \ldots $	12
2.15	Class D rectifier.	13
2.16	Class DE rectifier	13
2.17	$Class E \ rectifier. \ldots \ldots$	14
2.18	Class F rectifier	14
2.19	$\label{eq:inverse} Inverse \ class F \ rectifier. \ . \ . \ . \ . \ . \ . \ . \ . \ . \$	15
2.20	Load pull setup and transistor model	16
2.21	Theoretical load pull contours and impact of loss mechanisms	17
2.22	Theoretical efficiency versus sweep parameters	18
2.23	Probability density and resulting weighting functions.	19
2.24	Example efficiency functions and resulting average efficiency	19
3.1	Doherty block diagram and efficiency.	20
3.2	Doherty control, output power and partial efficiency.	21
3.3	LINC	22
3.4	Chireix block diagram	22
3.5	Chireix load modulation trajectories and efficiency	23
3.6	Block diagrams of envelope tracking and envelope elimination and restoration.	23
3.7	Load modulation block diagram.	24
3.8	Block diagram of broadband load operation	25
3.9	Drain voltage and current for PWM and $\Delta\Sigma$ operation for an isolated load	
	and efficiency.	25
3.10	Block diagram of direct filter operation.	26

3.11	Drain voltage and current for PWM and $\Delta\Sigma$ operation for a parallel filter and resulting officiency.	26
9 1 9	Die die die meine af dim at filten an anti-m	20
3.12	Block diagram of direct niter operation.	21
3.13	Drain voltage and current for PWM and $\Delta \Sigma$ operation for the series filter and efficiency	27
3.14	Direct filter connection of class B PA with switch	28
3.15	Drain voltage and current for PWM and series filter with switch and efficiency.	28
3.16	Block diagrams for multilevel combining.	29
3.17	Multi level efficiency for static, BB PWM and direct filter operation	29
3.18	Drain voltage and current for 2 level BB PWM encoding with $75\%$ output voltage.	30
3.19	Multiphase block diagram and signal generation for 2 levels	31
3.20	Multilevel and Multiphase spectra.	31
3.21	Drain voltage and current for 2 phase BB PWM encoding with 75% output	01
0.21	voltage	31
3.22	DC energy recovery outphasing	32
3.23	DC energy recovery BB PWM.	33
3.24	Block diagrams of BB PWM transmitters with/without energy reuse.	34
3.25	Duty cycle of 40% and conventional burst mode PA.	34
3.26	RF energy recovery block diagram OFF period for a duty cycle of 40 %.	35
3.27	Simulation results energy recovery during OFF period	35
3.28	RF energy recovery block diagram ON period for a duty cycle of 40%	36
3.29	Simulation results energy recovery during ON period.	37
3.30	RF energy recovery block diagram full recovery for a duty cycle of 40%.	38
3.31	Comparison of partial and full energy recovery.	38
3.32	General RF energy recovery block diagram.	38
3.33	BPSK PWM with isolator.	39
3.34	BPSK PWM with DC energy recovery.	39
3.35	Inverter based voltage mode class D.	40
4.1	Block diagram of BB PWM modulators.	42
4.2	Aliasing problem digital BB PWM	43
4.3	Aliasing free band limited BB PWM	43
4.4	Equivalent load impedances for full/partial load modulation.	44
4.5	Example resonance locus with impedance definitions of $Z_{\rm C}$ and $Z_{\rm S}$ and	
	resulting load impedances for full and partial modulation.	45
4.6	Different matching networks including high-Q BPF for direct filter connection.	46
4.7	Impedance behavior of the circuits in Fig. 4.6.	46
4.8	Unloaded/loaded Q-factor requirements of band pass filter.	47
4.9	Mechanical dimensions of the cavity and coupling equivalent circuit elements.	48
4.10	Equivalent circuit for capacitive coupling.	48
4.11	Insertion loss $(S_{21})$ and minimum insertion loss $(max(S_{21}))$ for inductive and	
	capacitive coupling for coupling length, respectively diameter and different	40
4 10	Q-ractors versus frequency.	49
4.12	Reflection $S_{11}$ versus frequency and minimum reflection $\min(S_{11})$ for differ-	FO
1 10	ent coupling lengths, respectively diameter and Q-factors for $Z_0 = 50\Omega$ .	00 E 1
4.13	Equivalent circuit for inductive coupling	51
4.14	reconcisional diameter	E 1
/ 15	respectively utalleter. $\dots$ and coupling length and dispector versus coupling hand	91
4.10	width	50
	YY 1(1 U11)	- 04

4.16	Equivalent circuit model for resonators considering the coupling and higher order modes with different coupling polarity.	52
4 17	Manufactured churcher and commence accelter	. 00 E9
4.17	Manufactured aluminum and copper cavity.	. 55
4.18	Measurement and model fit data (markers) of reflection coefficient $S_{11}$ and	F 4
4.10	Insertion loss $S_{21}$ for the first 4 1E-mode resonances	. 54
4.19	Gate signal conditioning direct filter connection.	. 54
4.20	Example impedance of direct filter connection and associated signal gener-	
		. 55
4.21	Schematic of the common source PA with direct filter connection.	. 57
4.22	Drain voltages and currents for the Class-B parallel PA circuit for a PWM frequency of 132.5 MHz, a carrier frequency of 2.65 GHz and a duty cycle	
4.00	of 50 %.	. 58
4.23	Load pull efficiency contours, matching network and load modulation tra-	-
	jectories for a carrier frequency of 2.6515 GHz.	. 58
4.24	Drain current of both PAs for low power mode	. 59
4.25	Drain current of both PAs for high power mode.	. 59
4.26	Efficiency, gain and output power versus frequency.	. 60
4.27	Efficiencies versus load voltage for a carrier frequency of 2.65 GHz and	
	20.7 MHz modulation frequency.	. 60
4.28	Schematic of the common drain PA.	61
4.29	Exemplary drain voltages and currents for the common drain PA for a	
	PWM frequency of 132.5 MHz, a carrier frequency of 2.65 GHz and a duty	
	cycle of $50\%$ .	62
4.30	Load pull efficiency contours, matching network and load modulation tra-	
	jectories for a carrier frequency of 2.6495 GHz.	62
4.31	Drain current of both transistors for low power mode.	63
4.32	Drain current of both transistors for high power mode	63
4.33	Efficiency, gain and output power versus frequency.	63
4.34	Efficiencies versus load voltage for a carrier frequency of 2.65 GHz and	
1.01	20.7 MHz modulation frequency	64
4.35	Switchable cascode general circuit	65
4.36	Simulated drain voltages and currents for operation at 2.65 GHz with switch	. 00
<b>1.00</b>	open/closed	65
1 37	Switchable essence implementation with breadband load	66
4.90	Simulated efficiency and gain versus lead veltage for both operating condi	. 00
4.00	tions at 2.65 CHz for the breadband load aircuit in Fig. 4.37	67
4.20	Effective resident and extension experience for the base dhead head head	. 07
4.39	Enciency, gain and output power versus frequency for the broadband load	60
4 40	Curcuit III F 1g. 4.57.	. 00
4.40	Switchable cascode implementation with direct filter connection.	. 69
4.41	Load pull efficiency contours, matching network and load modulation tra-	
	jectories for a carrier frequency of 2.048 GHz for the direct filter connection	co
4 40	circuit in Fig. 4.40.	. 69
4.42	Efficiency, gain and output power versus frequency for the direct filter con-	-
	nection circuit in Fig. 4.40.	. 70
4.43	Simulated efficiency enhancement by BB PWM with and without direct	
	filter connection.	. 70
4.44	Schematic of the source modulated PA.	. 71
4.45	Exemplary drain voltage and current for the source modulated PA for a	
	PWM frequency of $132.5\mathrm{MHz}$ , a carrier frequency of $2.65\mathrm{GHz}$ and a duty	
	cycle of $50\%$ .	72
4.46	Drain current and voltage for both operational modes	. 72

4.47	Load pull efficiency contours, matching network and load modulation tra-	79
4 40	Jectories for a carrier frequency of 2.048 GHz.	10
4.48	Efficiency, gain and output power versus frequency.	13
4.49	Efficiencies versus load voltage for a carrier frequency of 2.048 GHz and	74
4 50	20.7 MHz modulation frequency.	14
4.50	Block diagram biphase energy recovery.	75
4.51	Schematic of GaN class E biphase energy recovery transmitter using trans-	
4 50	mission line matching and a cavity resonator.	75
4.52	Load Pull contours at the intrinsic transistor plane for drain efficiency $\eta_D$	=0
	and output power $P_L$ at a carrier frequency of 2.648 GHz	76
4.53	Load pull contours at the intrinsic transistor plane for drain efficiency $\eta_D$	-
	and load modulation trajectories at a carrier frequency of 2.648 GHz	76
4.54	Drain current and voltage for PA and rectifier operation	77
4.55	Efficiencies vs. load voltage $V_L$	77
4.56	BPSK spectrum.	78
4.57	Multi level BB PWM implementation with FPGA and Doherty PA <sup>1</sup>	78
4.58	Single level FPGA output spectrum for a carrier frequency of 2.1 GHz and	
	a modulation frequency $f_{PWM}$ of 25 MHz with a signal bandwidth of 8 MHz.	79
4.59	Static total $PAE_D$ and inband $PAE_{IN}$ for the Doherty PA and BB PWM	
	operation at 2.75 GHz with a modulation frequency $f_{PWM}$ of 70 MHz	79
4.60	Doherty PA output spectrum for a carrier frequency of 2.75 GHz with a	
	modulation frequency $f_{PWM}$ of 70 MHz and a signal bandwidth of 14 MHz.	80
4.61	Receive band issue BB PWM	81
4.62	Multilevel BB PWM decomposition and potential error source	81
4.63	Location of the predistorter.	82
4.64	Ideal BL and saturated PWM signal for a duty cycle of 40 %	82
4.65	BB PWM harmonic decomposition.	-83
		00
5.1	Comparison of BB PWM and RF PWM spectrum	85
$5.1 \\ 5.2$	Comparison of BB PWM and RF PWM spectrum	85 85
$5.1 \\ 5.2 \\ 5.3$	Comparison of BB PWM and RF PWM spectrum	85 85 87
$5.1 \\ 5.2 \\ 5.3 \\ 5.4$	Comparison of BB PWM and RF PWM spectrum	85 85 87 87
$5.1 \\ 5.2 \\ 5.3 \\ 5.4 \\ 5.5$	Comparison of BB PWM and RF PWM spectrum	85 85 87 87 87
5.1 5.2 5.3 5.4 5.5 5.6	Comparison of BB PWM and RF PWM spectrum	85 85 87 87 87 88
5.1 5.2 5.3 5.4 5.5 5.6 5.7	Comparison of BB PWM and RF PWM spectrum	85 85 87 87 87 88 88 88
5.1 5.2 5.3 5.4 5.5 5.6 5.7 5.8	Comparison of BB PWM and RF PWM spectrum	85 85 87 87 87 87 88 88 88 88
5.1 5.2 5.3 5.4 5.5 5.6 5.7 5.8 5.9	Comparison of BB PWM and RF PWM spectrum	85 85 87 87 87 88 88 88 88 89
5.1 5.2 5.3 5.4 5.5 5.6 5.7 5.8 5.9 5.10	Comparison of BB PWM and RF PWM spectrum	85 85 87 87 87 87 87 88 88 88 89 89
5.1 5.2 5.3 5.4 5.5 5.6 5.7 5.8 5.9 5.10 5.11	Comparison of BB PWM and RF PWM spectrum	85 85 87 87 87 87 88 88 88 89 90 90
5.1 5.2 5.3 5.4 5.5 5.6 5.7 5.8 5.9 5.10 5.11 5.12	Comparison of BB PWM and RF PWM spectrum	85 85 87 87 87 87 88 88 89 90 90 90 91
$5.1 \\ 5.2 \\ 5.3 \\ 5.4 \\ 5.5 \\ 5.6 \\ 5.7 \\ 5.8 \\ 5.9 \\ 5.10 \\ 5.11 \\ 5.12 \\ 5.13 $	Comparison of BB PWM and RF PWM spectrum Block diagram of analog/digital transmitter	85 85 87 87 87 87 87 88 88 89 90 90 90 91 91
$5.1 \\ 5.2 \\ 5.3 \\ 5.4 \\ 5.5 \\ 5.6 \\ 5.7 \\ 5.8 \\ 5.9 \\ 5.10 \\ 5.11 \\ 5.12 \\ 5.13 \\ 5.14 $	Comparison of BB PWM and RF PWM spectrum Block diagram of analog/digital transmitter	85 85 87 87 87 88 88 89 90 90 90 91 91
5.1 5.2 5.3 5.4 5.5 5.6 5.7 5.8 5.9 5.10 5.11 5.12 5.13 5.14	Comparison of BB PWM and RF PWM spectrum Block diagram of analog/digital transmitter	85 85 87 87 87 87 87 87 87 88 88 89 90 90 90 91 91 91
$5.1 \\ 5.2 \\ 5.3 \\ 5.4 \\ 5.5 \\ 5.6 \\ 5.7 \\ 5.8 \\ 5.9 \\ 5.10 \\ 5.11 \\ 5.12 \\ 5.13 \\ 5.14 \\ 5.15 $	Comparison of BB PWM and RF PWM spectrum Block diagram of analog/digital transmitter	85 85 87 87 87 87 87 87 87 87 87 87 87 87 87
$5.1 \\ 5.2 \\ 5.3 \\ 5.4 \\ 5.5 \\ 5.6 \\ 5.7 \\ 5.8 \\ 5.9 \\ 5.10 \\ 5.11 \\ 5.12 \\ 5.13 \\ 5.14 \\ 5.15 \\ 5.16 \\$	Comparison of BB PWM and RF PWM spectrum	85 85 87 87 87 87 87 87 87 87 87 87 87 87 87
$5.1 \\ 5.2 \\ 5.3 \\ 5.4 \\ 5.5 \\ 5.6 \\ 5.7 \\ 5.8 \\ 5.9 \\ 5.10 \\ 5.11 \\ 5.12 \\ 5.13 \\ 5.14 \\ 5.15 \\ 5.16 $	Comparison of BB PWM and RF PWM spectrum	85 85 87 87 87 87 87 87 87 87 87 87 87 87 87
$\begin{array}{c} 5.1 \\ 5.2 \\ 5.3 \\ 5.4 \\ 5.5 \\ 5.6 \\ 5.7 \\ 5.8 \\ 5.9 \\ 5.10 \\ 5.11 \\ 5.12 \\ 5.13 \\ 5.14 \\ 5.15 \\ 5.16 \\ 5.17 \end{array}$	Comparison of BB PWM and RF PWM spectrum	85 85 87 87 87 87 87 87 87 87 87 87 87 87 87
$5.1 \\ 5.2 \\ 5.3 \\ 5.4 \\ 5.5 \\ 5.6 \\ 5.7 \\ 5.8 \\ 5.9 \\ 5.10 \\ 5.11 \\ 5.12 \\ 5.13 \\ 5.14 \\ 5.15 \\ 5.16 \\ 5.17 \\$	Comparison of BB PWM and RF PWM spectrum	85 85 87 87 87 87 87 87 87 87 87 87 87 87 87
$5.1 \\ 5.2 \\ 5.3 \\ 5.4 \\ 5.5 \\ 5.6 \\ 5.7 \\ 5.8 \\ 5.9 \\ 5.10 \\ 5.11 \\ 5.12 \\ 5.13 \\ 5.14 \\ 5.15 \\ 5.16 \\ 5.17 \\ 5.18 $	Comparison of BB PWM and RF PWM spectrum	85 85 87 87 87 87 87 87 87 87 87 87 87 87 87
$5.1 \\ 5.2 \\ 5.3 \\ 5.4 \\ 5.5 \\ 5.6 \\ 5.7 \\ 5.8 \\ 5.9 \\ 5.10 \\ 5.11 \\ 5.12 \\ 5.13 \\ 5.14 \\ 5.15 \\ 5.16 \\ 5.17 \\ 5.18 $	Comparison of BB PWM and RF PWM spectrum	85 85 87 87 87 87 87 87 87 87 87 87 87 87 87
$5.1 \\ 5.2 \\ 5.3 \\ 5.4 \\ 5.5 \\ 5.6 \\ 5.7 \\ 5.8 \\ 5.9 \\ 5.10 \\ 5.11 \\ 5.12 \\ 5.13 \\ 5.14 \\ 5.15 \\ 5.16 \\ 5.17 \\ 5.18 \\ 5.19 \\ 5.19 \\$	Comparison of BB PWM and RF PWM spectrum	85 85 87 87 87 87 87 87 87 87 87 87 87 87 87

5.20	RF PWM for direct digital to RF conversion.	. 95
5.21	RF PWM modulator block diagram.	. 96
5.22	RF PWM signal generation.	. 97
5.23	Digital phase modulator implementation.	. 98
5.24	Combination and quantization block.	. 98
5.25	Delta Sigma operation for resolution enhancement.	. 99
5.26	Spectrum of RF PWM signal at 2.6 GHz with 40 MHz bandwidth for limited	
	resolution without and with $\Delta \Sigma$ .	. 99
5.27	Illustration of aliasing of the higher order harmonics due to ZOH	. 100
5.28	ZOH problem in the time domain.	. 101
5.29	CPE using (phase modulated) triangular reference function	. 102
5.30	CPE block diagram.	. 103
5.31	CPE calculation for a single branch signal	. 103
5.32	CPE iterative interpolation.	. 104
5.33	Example timing diagram for pulse separation for main and auxiliary path.	. 105
5.34	Reference and error signal spectra with and without CPE for a carrier	
	frequency of 2.6 GHz, a bandwidth of 40 MHz and a resolution of 11 bit.	. 106
5.35	Block diagram of main and auxiliary path modulator.	. 106
5.36	Block diagram for PWM modulator with CPE error noise shaping	. 107
5.37	Example timing diagram for pulse generation for delta sigma CPE noise	
	shaping	. 107
5.38	Measured signal at 1 MHz carrier frequency, 20 kHz bandwidth and 10 bit	
	resolution.	. 108
5.39	Measured signal at 1 MHz carrier frequency, 20 kHz bandwidth and 40 kHz	
	frequency offset (carrier aggregation).	. 109
5.40	Driver block diagram.	. 110
5.41	Driver errors and AM/AM, AM/PM deviation.	. 110
5.42	Driver effects.	. 111
5.43	Static and dynamic pulse swallowing.	. 111
5.44	Carry over based predistortion block diagram.	. 112
5.45	Sample and pulse based predistortion carry over/back effects.	. 112
5.46	Pulse predistortion time domain three signal transitions.	. 113
5.47	Pulse predistortion circular overlap problem.	. 114
5.48	Spectra for a 40 MHz bandwidth signal at 2.6 GHz with driver nonlinearity	
	corrected by different predistortion schemes.	. 115
5.49	Increased quantization step due to nonlinearity.	. 115
5.50	Predistorter location.	. 116
5.51	Spectra for a 40 MHz bandwidth signal at 2.6 GHz with driver nonlinearity	
	corrected by pre $\Delta\Sigma$ predistortion using 1 <sup>st</sup> order $\Delta\Sigma$ .	. 116
5.52	Measurement setup and driver distortion.	. 117
5.53	Measured signal at the driver output at 1 MHz carrier frequency, 20 kHz	
	bandwidth and 10 bit resolution.	. 117
5.54	Measured signal at the driver output at 1 MHz carrier frequency, 20 kHz	
	bandwidth and 40 kHz frequency offset (carrier aggregation) and 7 bit reso-	
		. 118
5.55	$\Delta\Sigma$ and digital mismatch compensation.	. 118
5.56	Mismatch compensation and measurement sensitivity for $1^{st}$ order $\Delta \Sigma$ .	. 119
7.1	General class A, B and C schematic.	. 123
7.2	$Current \ mode \ class D. \ \ \ldots \$	. 125
7.3	Current mode class D with series resonator. $\ldots \ldots \ldots \ldots \ldots \ldots$	. 126
7.4	$Voltage \ mode \ class D. \ \ \ldots $	. 128

7.5	Voltage mode class D
7.6	Voltage mode class DE with parallel compensation inductance. $\ldots$ 131
7.7	Voltage mode class DE with series compensation inductance
7.8	DC energy recovery BB PWM
7.9	DC energy recovery outphasing
7.10	RF energy recovery
7.11	BPSK isolated operation
7.12	DC BPSK energy recovery
7.13	Chireix combiner with shunt reactance
7.14	Asymmetric transmission line Chireix combiner
7.15	Voltage and current waveforms and equivalent circuit for (a) full and (b)
	partial PWM modulation
7.16	Definition of $Z_C$ and $Z_S$
7.17	Losses in band pass filters
7.18	GaN transistor and GaN diode IV curves
7.19	CPE overview
7.20	CPE iterative interpolation
7.21	CPE minimum error signal selection

## List of Tables

2.1	Overview amplifier classes	15
4.1	Equivalent circuit parameters	53
7.1	Logic table edge timing selection.	154

## List of Abbreviations

ACLR	Adjacent Channel Leakage Ratio
AM	Amplitude Modulator
AWG	Arbitrary Waveform Generator
BALUN	Balanced Unbalanced
	Pass Pand Pulsa Width Modulation
	Dase Dand Fulse width Modulation
BT BT	
BPF	Band Pass Filter
BPSK	Binary Phase Shift Keying
CO	Carry Over
CORDIC	COordinate Rotation DIgital Computer
CPE	Cross Point Estimation
DAC	Digital to Analog Converter
DE	Drain Efficiency
DPD	Digital Predistortion
ED	Envelope Detector
EER	Envelope Elimination and Restoration
ET	Envelope Tracking
EVM	Error Vector Magnitude
FPGA	Field Programmable Gate Array
IMN	Input Matching Network
LINC	Linear Amplification Using Nonlinear Components
LO	Local Oscillator
LP	Load Pull
LUT	Look Up Table
MMIC	Monolithic Microwave Integrated Circuit
MOD	

OMN	Output Matching Network
PA	Power Amplifier
PAE	Power Added Efficiency
PAPR	Peak to Average Power Ratio
PCB	Printed Circuit Board
PDF	Probability Density Function
PLL	Phase Locked Loop
PM	Phase Modulator
PS	Pulse Swallowing
PSD	Power Spectral Density
PUF	Power Utilization Factor
PWM	Pulse Width Modulation
RFC	RF Choke
RF PWM	Radio Frequency Pulse Width Modulation
RMS	Root Mean Square
SDR	Software Defined Radio
SMPA	Switched Mode Power Amplifier
TL	Transmission Line
VSA	Vector Signal Analyzer
ZCS	Zero Current Switching
ZOH	Zero Order Hold
ZVS	Zero Voltage Switching